UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended March 26, 2017

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES AND EXCHANGE ACT OF 1934

Commission file number 001-34460

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

(Exact name of Registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

13-3818604

(I.R.S. Employer Identification No.)

4820 Eastgate Mall, Suite 200 San Diego, CA 92121 (858) 812-7300

(Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (Section 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \boxtimes No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act:

Large accelerated filer o

Non-accelerated filer o

(Do not check if a smaller reporting company)

Smaller reporting company o Emerging growth company o

Accelerated filer ⊠

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. Yes o No o

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🖂

As of April 28, 2017, 86,446,084 shares of the registrant's common stock were outstanding.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

FORM 10-Q

FOR THE QUARTERLY PERIOD ENDED MARCH 26, 2017

INDEX

		Page
	PART I. FINANCIAL INFORMATION	
<u>Item 1.</u>	Financial Statements	<u>3</u>
	Condensed Consolidated Balance Sheets at March 26, 2017 and December 25, 2016 (unaudited)	<u>3</u>
	Condensed Consolidated Statements of Operations and Comprehensive Loss for the Three Months Ended March 26, 2017 and March 27, 2016 (unaudited)	<u>4</u>
	Condensed Consolidated Statements of Cash Flows for the Three Months Ended March 26, 2017 and March 27, 2016 (unaudited)	<u>5</u>
	Notes to Condensed Consolidated Financial Statements (unaudited)	<u>6</u>
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>24</u>
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	<u>32</u>
<u>Item 4.</u>	Controls and Procedures	<u>33</u>
	PART II. OTHER INFORMATION	
<u>Item 1.</u>	Legal Proceedings	<u>34</u>
Item 1A.	<u>Risk Factors</u>	<u>34</u>
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	<u>34</u>
<u>Item 3.</u>	Defaults Upon Senior Securities	<u>34</u>
<u>Item 4.</u>	Mine Safety Disclosures	<u>34</u>
<u>Item 5.</u>	Other Information	<u>34</u>
<u>Item 6.</u>	Exhibits	<u>35</u>

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (in millions, except par value and number of shares) (Unaudited)

	Mar	ch 26, 2017	December 25, 2016		
Assets					
Current assets:					
Cash and cash equivalents	\$	73.4	\$ 69.1		
Restricted cash		0.5	0.5		
Accounts receivable, net		227.9	229.4		
Inventoried costs		64.8	55.4		
Prepaid expenses		8.9	8.9		
Other current assets		11.8	9.8		
Total current assets		387.3	373.1		
Property, plant and equipment, net		50.8	49.8		
Goodwill		485.4	485.4		
Intangible assets, net		29.9	32.6		
Other assets		8.2	7.7		
Total assets	\$	961.6	\$ 948.6		
Liabilities and Stockholders' Equity					
Current liabilities:					
Accounts payable	\$	51.8	\$ 52.7		
Accrued expenses		49.8	50.0		
Accrued compensation		35.2	39.1		
Accrued interest		9.6	3.6		
Billings in excess of costs and earnings on uncompleted contracts		41.8	41.8		
Other current liabilities		6.4	7.7		
Current liabilities of discontinued operations		1.4	1.6		
Total current liabilities		196.0	196.5		
Long-term debt principal, net of current portion		369.3	431.0		
Other long-term liabilities		41.2	41.0		
Non-current liabilities of discontinued operations		3.7	3.7		
Total liabilities		610.2	672.2		
Commitments and contingencies					
Stockholders' equity:					
Preferred stock, \$0.001 par value, 5,000,000 shares authorized, 0 shares outstanding at March 26, 2017 December 25, 2016	7 and	_	_		
Common stock, \$0.001 par value, 195,000,000 shares authorized; 86,439,842 and 73,945,533 shares is and outstanding at March 26, 2017 and December 25, 2016, respectively	ssued	_	_		
Additional paid-in capital		1,041.0	956.2		
Accumulated other comprehensive loss		(1.6)	(1.7)		
Accumulated deficit		(688.0)	(678.1)		
Total stockholders' equity		351.4	276.4		
Total liabilities and stockholders' equity		961.6	\$ 948.6		

The accompanying notes are an integral part of these condensed consolidated financial statements.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (in millions, except per share amounts) (Unaudited)

	Three M	onths Ended
	March 26, 2017	March 27, 2016
Service revenues	\$ 85.0	\$ 82.6
Product sales	82.8	70.4
Total revenues	167.8	153.0
Cost of service revenues	61.8	60.3
Cost of product sales	60.9	56.8
Total costs	122.7	117.1
Gross profit	45.1	35.9
Selling, general and administrative expenses	38.7	37.7
Research and development expenses	4.4	2.9
Unused office space, restructuring expenses, and other	0.3	5.5
Operating income (loss) from continuing operations	1.7	(10.2)
Other income (expense):		_
Interest expense, net	(8.2)	(8.7)
Loss on extinguishment of debt	(2.1)	—
Other income, net	0.2	0.3
Total other expense, net	(10.1)	(8.4)
Loss from continuing operations before income taxes	(8.4)	(18.6)
Provision for income taxes from continuing operations	1.5	3.6
Loss from continuing operations	(9.9)	(22.2)
Discontinued operations		
Loss from operations of discontinued component	(0.1)	_
Income tax expense	_	_
Loss from discontinued operations	(0.1)	
Net loss	\$ (10.0)	\$ (22.2)
Basic and diluted loss per common share:		
Net loss from continuing operations	\$ (0.13)	\$ (0.37)
Net loss from discontinued operations	_	_
Net loss per common share	\$ (0.13)	\$ (0.37)
Basic and diluted weighted average shares outstanding	77.3	59.6
Comprehensive Loss		
Net loss (from above)	\$ (10.0)	\$ (22.2)
Change in cumulative translation adjustment	0.1	—
Comprehensive loss	\$ (9.9)	\$ (22.2)

The accompanying notes are an integral part of these condensed consolidated financial statements.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (in millions) (Unaudited)

	Three Mo	nths Ended
	March 26, 2017	March 27, 2016
Operating activities:		
Net loss	\$ (10.0)	\$ (22.2)
Less: Loss from discontinued operations	(0.1)	
Loss from continuing operations	(9.9)	(22.2)
Adjustments to reconcile loss from continuing operations to net cash used in operating activities from continuing operations:		
Depreciation and amortization	5.6	6.1
Stock-based compensation	2.1	1.5
Deferred income taxes	0.8	1.2
Amortization of deferred financing costs	0.4	0.4
Amortization of discount on Senior Secured Notes	0.2	0.2
Loss on extinguishment of debt	2.1	—
Provision for doubtful accounts	—	0.3
Litigation related charges	_	1.7
Provision for non-cash restructuring charges	_	3.0
Changes in assets and liabilities, net of acquisitions:		
Accounts receivable	1.5	(4.0)
Inventoried costs	(10.1)	(1.6)
Prepaid expenses and other assets	(3.8)	0.1
Accounts payable	0.4	(8.8)
Accrued compensation	(4.0)	(4.2)
Accrued expenses	(0.3)	(3.1)
Advance payments received on contracts	0.7	1.5
Accrued interest	6.0	7.9
Billings in excess of costs and earnings on uncompleted contracts	_	6.7
Income tax receivable and payable	0.5	0.4
Other liabilities	(0.9)	1.4
Net cash used in operating activities from continuing operations	(8.7)	(11.5)
Investing activities:		
Capital expenditures	(5.2)	(2.1)
Net cash used in investing activities from continuing operations	(5.2)	(2.1)
Financing activities:		
Extinguishment of long-term debt	(64.0)	_
Proceeds from the issuance of common stock	81.9	_
Repayment of debt	(0.3)	(0.3)
Proceeds from exercise of restricted stock units, employee stock options, and employee stock purchase plan		1.2
Net cash provided by financing activities from continuing operations	18.4	0.9
Net cash flows of continuing operations	4.5	(12.7)
Net operating cash flows of discontinued operations	_	(0.4)
Net investing cash flows of discontinued operations	(0.2)	4.7
Net increase (decrease) in cash and cash equivalents	4.3	(8.4)
Cash and cash equivalents at beginning of period	69.1	28.5
Cash and cash equivalents at end of period	\$ 73.4	\$ 20.1

The accompanying notes are an integral part of these condensed consolidated financial statements.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

Note 1. Summary of Significant Accounting Policies

All references to the "Company" and "Kratos" refer to Kratos Defense & Security Solutions, Inc., a Delaware corporation, and its subsidiaries.

(a) Basis of Presentation

The information as of March 26, 2017 and for the three months ended March 26, 2017 and March 27, 2016 is unaudited. The condensed consolidated balance sheet as of December 25, 2016 was derived from the Company's audited consolidated financial statements at that date. In the opinion of management, these unaudited condensed consolidated financial statements include all adjustments, consisting of normal recurring adjustments necessary for a fair presentation of the Company's financial position, results of operations and cash flows for the interim periods presented. The results have been prepared in accordance with the instructions to Form 10-Q and do not necessarily include all information and footnotes necessary for presentation in accordance with accounting principles generally accepted in the U.S. These unaudited condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and the related notes included in the Company's audited annual consolidated financial statements for the fiscal year ended December 25, 2016, included in the Company's Annual Report on Form 10-K filed with the U.S. Securities and Exchange Commission on February 27, 2017 (the "Form 10-K"). Interim operating results are not necessarily indicative of operating results expected in subsequent periods or for the year as a whole.

(b) Principles of Consolidation

The condensed consolidated financial statements include the accounts of the Company and its 100% owned subsidiaries for which all inter-company transactions have been eliminated in consolidation.

(c) Fiscal Year

The Company has a 52/53 week fiscal year ending on the last Sunday of the calendar year, with interim fiscal periods ending on the last Sunday of each calendar quarter. The three month periods ended March 26, 2017 and March 27, 2016 consisted of 13-week periods. There are 53 calendar weeks in the fiscal year ending on December 31, 2017 and 52 calendar weeks in the fiscal year ending on December 25, 2016.

(d) Accounting Estimates

There have been no significant changes in the Company's accounting estimates for the three months ended March 26, 2017 as compared to the accounting estimates described in the Form 10-K.

(e) Accounting Standards Updates

In January 2017, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2017-04 ("ASU 2017-04"), Intangibles - Goodwill and Other (Topic 350): Simplifying the Test for Goodwill Impairment. ASU 2017-04 amends the guidance to simplify the subsequent measurement of goodwill by removing Step 2 of the goodwill impairment test. Instead, an entity should perform its annual, or interim, goodwill impairment test by comparing the fair value of a reporting unit with its carrying amount. An entity should recognize an impairment charge for the amount by which the carrying amount exceeds the reporting unit's fair value. ASU 2017-04 is effective for annual reporting periods beginning after December 15, 2019, with early adoption permitted. The Company is currently evaluating the impact of the new guidance and timing of adoption, but does not expect that the standard will have a material impact on its Consolidated Financial Statements.

In October 2016, the FASB issued ASU 2016-16 ("ASU 2016-16"), *Income Taxes (Topic 740), Intra-Entity Transfers of Assets Other Than Inventory.* ASU 2016-16 requires that entities recognize the income tax consequences of an intra-entity transfer of an asset other than inventory when the transfer occurs, rather than when the asset is sold to an outside party. ASU 2016-16 is effective for annual reporting periods beginning after December 15, 2017, including interim periods within those annual reporting periods. Early adoption is permitted as of the beginning of an annual reporting period (as of the first interim period if an entity issues interim financial statements). ASU 2016-16 requires adoption on a modified retrospective basis through a cumulative-effect adjustment directly to retained earnings as of the beginning of the period of adoption. The Company early adopted this standard on December 26, 2016. The adoption of this guidance did not have a material impact on the Company's consolidated financial statements.

In August 2016, the FASB issued ASU 2016-15 ("ASU 2016-15"), *Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments.* The objective of ASU 2016-15 is to reduce existing diversity in practice by addressing eight specific cash flow issues related to how certain cash receipts and cash payments are presented and classified in the statement of cash flows. ASU 2016-15 is effective for fiscal years beginning after December 15, 2017, including interim periods within those fiscal years. Early adoption is permitted. If early adopted, an entity must adopt all of the amendments in the same period. The Company does not believe that the adoption of this guidance will have a material impact on its consolidated financial statements.

In March 2016, the FASB issued ASU 2016-09 ("ASU 2016-09"), *Compensation — Stock Compensation (Topic 718): Improvements to Employee Share-Based Payment Accounting*. ASU 2016-09 simplifies several aspects of the accounting for employee share-based payments, including accounting for income taxes, forfeitures, statutory tax withholding requirements, and classification on the statement of cash flows. The amendments in this ASU are effective for annual periods beginning after December 15, 2016. Early adoption was permitted. The Company early adopted this standard in the quarter ended December 25, 2016, which did not have a material impact on its consolidated financial statements.

In February 2016, the FASB issued ASU 2016-02 ("ASU 2016-02"), *Leases*. ASU 2016-02 requires that lessees recognize assets and liabilities for the rights and obligations for leases with a lease term of more than one year. The amendments in this ASU are effective for annual periods ending after December 15, 2018. Early adoption is permitted. The Company is currently evaluating the impact of the adoption of ASU 2016-02 on its consolidated financial statements.

In May 2014, the FASB issued ASU 2014-09 ("ASU 2014-09"), *Revenue from Contracts with Customers*. ASU 2014-09 establishes a broad principle that would require an entity to recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. To achieve this principle, an entity identifies the contract with a customer, identifies the separate performance obligations in the contract, determines the transaction price, allocates the transaction price to the separate performance obligations and recognizes revenue when each separate performance obligation is satisfied. ASU 2014-09 was further updated in March, April, May, and December 2016 to provide clarification on a number of specific issues as well as requiring additional disclosures. ASU 2014-09 may be applied either retrospectively or through the use of a modified-retrospective method. The full retrospective method requires companies to recast each prior reporting period presented as if the new guidance had always existed. Under the modified retrospective method, companies would recognize the cumulative effect of initially applying the standard as an adjustment to opening retained earnings at the date of initial application. On July 9, 2015, the FASB approved a one year deferral of the effective date of ASU 2014-09 to annual reporting periods beginning after December 15, 2017. The Company plans to adopt the new revenue standard effective January 1, 2018.

While the Company has not yet completed its evaluation of the impact of ASU 2014-09 upon adoption, based upon an assessment of material active contracts, the Company does not expect the impact on the results of operations or cash flows in the periods after adoption to be material. Under ASU 2014-09, revenue is recognized as control transfers to the customer. As such, revenue for the Company's contracts will generally be recognized over time using the cost-to-cost method, which is consistent with the revenue recognition model currently in use for the majority of contracts. For those contracts where revenue is currently recognized as units are delivered, in most cases the accounting for those contracts will change under ASU 2014-09 such that revenue will be recognized as costs are incurred. This change will generally result in an acceleration of revenue as compared with the current revenue recognition method for those contracts. The Company expects to adopt the new standard using the modified retrospective method.

There have been no changes in the Company's significant accounting policies for the three months ended March 26, 2017 as compared to the significant accounting policies described in the Form 10-K.

(f) Fair Value of Financial Instruments

The carrying amounts and the related estimated fair values of the Company's long-term debt financial instruments not measured at fair value on a recurring basis at March 26, 2017 and December 25, 2016 are presented in Note 7. The carrying value of all other financial instruments, including cash equivalents, accounts receivable, accounts payable, accrued expenses,

billings in excess of cost and earnings on uncompleted contracts, income taxes payable and short-term debt, approximated their estimated fair values at March 26, 2017 and December 25, 2016 due to the short-term nature of these instruments.

Note 2. Goodwill and Intangible Assets

(a) Goodwill

The carrying amounts of goodwill as of March 26, 2017 and December 25, 2016 by reportable segment are as follows (in millions):

	Gove	Kratos Government Public Safety & Solutions Security			nmanned Systems	Total
Gross value	\$	567.8	\$	53.9	\$ 111.1	\$ 732.8
Less accumulated impairment		215.3		18.3	13.8	247.4
Net	\$	352.5	\$	35.6	\$ 97.3	\$ 485.4

(b) Purchased Intangible Assets

The following table sets forth information for finite-lived and indefinite-lived intangible assets (in millions):

	As of March 26, 2017							As of December 25, 2016							
		Gross Value	Accumulated Amortization			Net Value	Gross Value		Accumulated Amortization			Net Value			
Acquired finite-lived intangible assets:															
Customer relationships	\$	53.7	\$	(46.2)	\$	7.5	\$	53.7	\$	(44.9)	\$	8.8			
Contracts and backlog		30.8		(24.2)		6.6		30.8		(23.7)		7.1			
Developed technology and technical know how	-	25.2		(16.5)		8.7		25.2		(15.7)		9.5			
Trade names		1.4		(1.2)		0.2		1.4		(1.1)		0.3			
Total finite-lived intangible assets		111.1		(88.1)		23.0		111.1		(85.4)		25.7			
Indefinite-lived trade names		6.9		_		6.9		6.9				6.9			
Total intangible assets	\$	118.0	\$	(88.1)	\$	29.9	\$	118.0	\$	(85.4)	\$	32.6			

Consolidated amortization expense related to intangible assets subject to amortization was \$2.7 million and \$2.7 million for the three months ended March 26, 2017 and March 27, 2016, respectively.

Note 3. Inventoried Costs

Inventoried costs consisted of the following components (in millions):

		ch 26, 17	Dec	ember 25, 2016
Raw materials	\$	32.3	\$	31.9
Work in process		30.3		22.1
Finished goods		1.7		1.4
Supplies and other		3.0		1.8
Subtotal inventoried costs	<u>.</u>	67.3		57.2
Less: Customer advances and progress payments		(2.5)		(1.8)
Total inventoried costs	\$	64.8	\$	55.4

Note 4. Stockholders' Equity

A summary of the changes in stockholders' equity is provided below (in millions):

	For the Three Months End				
	Mar	ch 26, 2017	Marc	ch 27, 2016	
Stockholders' equity at beginning of period	\$	276.4	\$	254.2	
Comprehensive loss:					
Net loss		(10.0)		(22.2)	
Change in cumulative translation adjustment		0.1		—	
Total comprehensive loss		(9.9)		(22.2)	
Stock-based compensation		2.1		1.5	
Issuance of common stock for cash		81.9		_	
Issuance of common stock for employee stock purchase plan		1.4		1.3	
Restricted stock units exchanged for taxes		(0.5)		(0.1)	
Stockholders' equity at end of period	\$	351.4	\$	234.7	

The components of accumulated other comprehensive loss are as follows (in millions):

	March 26	, 2017	March 27, 201		
Cumulative translation adjustment	\$	(1.0)	\$	(0.6)	
Post-retirement benefit reserve adjustment net of tax expense		(0.6)		(0.8)	
Total accumulated other comprehensive loss	\$	(1.6)	\$	(1.4)	

There were no reclassifications from accumulated other comprehensive loss to net loss for the three months ended March 26, 2017 and March 27, 2016.

Common stock issued by the Company for the three months ended March 26, 2017 and March 27, 2016 was as follows (in millions):

	For the Three	Months Ended
	March 26, 2017	March 27, 2016
Shares outstanding at beginning of the period	73.9	59.1
Stock issued for cash	11.9	_
Stock issued for employee stock purchase plan, stock options and restricted stock units exercised	0.6	0.5
Shares outstanding at end of the period	86.4	59.6

Note 5. Net Loss Per Common Share

The Company calculates net loss per share in accordance with FASB Accounting Standards Codification Topic 260, *Earnings per Share (Topic 260)*. Under Topic 260, basic net loss per common share is calculated by dividing net loss by the weighted-average number of common shares outstanding during the reporting period. Diluted net loss per common share reflects the effects of potentially dilutive securities.

Shares from stock options and awards, excluded from the calculation of diluted net loss per share because their inclusion would have been antidilutive, were 0.2 million and 2.5 million for the three months ended March 26, 2017 and March 27, 2016, respectively.

Note 6. Income Taxes

A reconciliation of the income tax benefit from continuing operations computed by applying the statutory federal income tax rate of 35% to loss from continuing operations before income taxes to the income tax provision for the three months ended March 26, 2017 and March 27, 2016 was as follows (in millions):

	М	March 26, 2017		rch 27, 2016
Income tax benefit at federal statutory rate	\$	(2.9)	\$	(6.5)
State and foreign taxes, net of federal tax benefit and valuation allowance		0.2		0.4
Nondeductible expenses and other		0.4		0.4
Impact of deferred tax liabilities for indefinite-lived assets		1.2		1.5
Increase in reserves for uncertain tax positions		0.1		1.7
Increase in federal valuation allowance		2.5		6.1
Total income tax provision	\$	1.5	\$	3.6

In assessing the Company's ability to realize deferred tax assets, management considers, on a periodic basis, whether it is more likely than not that some portion or all of the deferred tax assets will not be realized. As such, management has determined that it is appropriate to maintain a full valuation allowance against the Company's U.S. federal, combined state and certain foreign deferred tax assets, with the exception of an amount equal to its deferred tax liabilities, which can be expected to reverse over a definite life.

Federal and state income tax laws impose restrictions on the utilization of net operating losses ("NOLs") and tax credit carryforwards in the event that an "ownership change" occurs for tax purposes, as defined by Section 382 of the Internal Revenue Code of 1986, as amended ("Section 382"). In general, an ownership change occurs when shareholders owning 5% or more of a "loss corporation" (a corporation entitled to use NOLs or other loss carryovers) have increased their ownership of stock in such corporation by more than 50 percentage points during any three-year period. The annual base Section 382 limitation is calculated by multiplying the loss corporation's value at the time of the ownership change by the greater of the long-term tax-exempt rate determined by the Internal Revenue Service in the month of the ownership change or the two preceding months. This base limitation is subject to adjustments, including an increase for built-in gains recognized in the five-year period after the ownership change.

In March 2010, an "ownership change" occurred that will limit the utilization of NOL carryforwards. In July 2011, another "ownership change" occurred. The March 2010 ownership change limitation is more restrictive. In prior years, the Company acquired corporations with NOL carryforwards at the date of acquisition ("Acquired NOLs"). The Acquired NOLs are subject to separate limitations that may further restrict the use of Acquired NOLs. As a result, the Company's federal annual utilization of NOL carryforwards was limited to at least \$27.0 million a year for the five years succeeding the March 2010 ownership change and at least \$11.6 million for each year thereafter subject to separate limitations for Acquired NOLs. If the entire limitation amount is not utilized in a year, the excess can be carried forward and utilized in future years.

For the three months ended March 26, 2017, there was no impact of such limitations on the income tax provision, since the amount of taxable income did not exceed the annual limitation amount. However, future equity offerings or acquisitions that have equity as a component of the purchase price could also cause an "ownership change." If and when any other "ownership change" occurs, utilization of the NOLs or other tax attributes may be further limited.

As discussed elsewhere, deferred tax assets relating to the NOL and credit carryforwards are offset by a full valuation allowance. In addition, utilization of state tax loss carryforwards is dependent upon sufficient taxable income apportioned to the states.

The Company is subject to taxation in the U.S. and various state and foreign tax jurisdictions. The Company's tax years for 2000 and later are subject to examination by the U.S. and state tax authorities due to the existence of the NOL carryforwards. Generally, the Company's tax years for 2002 and later are subject to examination by various foreign tax authorities as well.

As of December 25, 2016, the Company had \$18.6 million of unrecognized tax benefits that, if recognized, would impact the effective income tax rate, subject to possible offset by an increase in the deferred tax asset valuation allowance. During the three months ended March 26, 2017, unrecognized tax benefits remained unchanged.

The Company recognizes interest and penalties related to unrecognized tax benefits in its provision for income taxes. For the three months ended March 26, 2017 and March 27, 2016, a \$0.1 million expense and \$0.6 million expense, respectively, were recorded related to interest and penalties related to unrecognized tax benefits. For the three months ended March 26, 2017 and March 27, 2016, there was no material benefit recorded related to the removal of interest and penalties. The Company believes that it is reasonably possible that as much as \$1.1 million of the liabilities for uncertain tax positions will expire within twelve months of March 26, 2017 due to the expiration of various applicable statutes of limitations.

Note 7. Debt

(a) Issuance of 7.00% Senior Secured Notes due 2019

In May 2014, the Company refinanced its \$625.0 million 10% Senior Secured Notes due in 2017 (the "10% Notes") with \$625.0 million of newly issued 7.00% Senior Secured Notes due in 2019 (the "7% Notes"). The net proceeds from the issuance of the 7% Notes was \$618.5 million after an original issue discount of \$6.5 million. The Company incurred debt issuance costs of \$8.8 million associated with the 7% Notes. The Company utilized the net proceeds from the issuance of the 7% Notes, a \$41.0 million draw on its Credit Agreement (as defined below), as well as cash from operations to extinguish the 10% Notes. The total reacquisition price of the 10% Notes was \$661.5 million including a \$31.2 million early termination fee, the write-off of \$15.5 million of unamortized issue costs, \$12.9 million of unamortized premium, along with \$5.3 million of additional interest while in escrow, which resulted in a loss on extinguishment of debt of \$39.1 million.

The Company completed the offering of the 7% Notes in a private placement conducted pursuant to Rule 144A and Regulation S under the Securities Act of 1933, as amended (the "Act"). On October 16, 2014, the Company exchanged the outstanding 7% Notes for an equal amount of new 7.00% Senior Secured Notes due in 2019 (the "Notes") that had been registered under the Act. The terms of the Notes issued in the exchange offer were identical in all material respects to the terms of the 7% Notes, except the Notes issued in the exchange offer had been registered under the Act.

The Notes are governed by an Indenture, dated May 14, 2014 (the "Indenture"), among the Company, certain of the Company's subsidiaries (the "Subsidiary Guarantors") and Wilmington Trust, National Association, as Trustee and Collateral Agent. A Subsidiary Guarantor can be released from its Guarantee (as defined in the Indenture) if (a) all of the capital stock issued by such Subsidiary Guarantor or all or substantially all of the assets of such Subsidiary Guarantor are sold or otherwise disposed of; (b) the Company designates such Subsidiary Guarantor as an Unrestricted Subsidiary (as defined in the Indenture); (c) if the Company exercises its legal defeasance option or its covenant defeasance option; or (d) upon satisfaction and discharge of the

Indenture or payment in full in cash of the principal of, premium, if any, and accrued and unpaid interest on the Notes.

The holders of the Notes have a first priority lien on substantially all of the Company's assets and the assets of the Subsidiary Guarantors, except with respect to accounts receivable, inventory, deposit accounts, securities accounts, cash, securities and general intangibles (other than intellectual property), on which the holders of the Notes have a second priority lien to the \$110.0 million Credit Agreement.

The Company pays interest on the Notes semi-annually, in arrears, on May 15 and November 15 of each year.

The Notes include customary covenants and events of default as well as a consolidated fixed charge ratio of 2.0:1 for the incurrence of additional indebtedness. Negative covenants include, among other things, limitations on additional debt, liens, negative pledges, investments, dividends, stock repurchases, asset sales and affiliate transactions. Events of default include, among other events, non-performance of covenants, breach of representations, cross-default to other material debt, bankruptcy, insolvency, material judgments and changes in control. As of March 26, 2017, the Company was in compliance with the covenants contained in the Indenture governing the Notes.

The Company may redeem some or all of the Notes at 105.25% of the aggregate principal amount of such Notes through May 15, 2017, 102.625% of the aggregate principal amount of such Notes if redeemed on or after May 16, 2017 but on or before May 15, 2018 and 100% of the aggregate principal amount of such Notes if redeemed on or after May 16, 2018, plus accrued and unpaid interest to the date of redemption.

The terms of the Indenture require that the net cash proceeds from asset dispositions be utilized to (i) repay or prepay amounts outstanding under the Credit Agreement unless such amounts are reinvested in similar collateral, (ii) make an investment in assets that replace the collateral of the Notes or (iii) a combination of both clauses (i) and (ii). To the extent there are any remaining net proceeds from the asset disposition after application of clauses (i) and (ii), such amounts are required to be utilized to repurchase Notes at par after 360 days following the asset disposition.

Following the sale of its U.S. and U.K. Electronic Products Divisions (the "Herley Entities") the Company paid down \$41.0 million outstanding under the Credit Agreement and repurchased \$175.0 million of the Notes at par, in accordance with the terms of the Indenture. The total reacquisition price of the Notes was \$178.4 million including the write off of \$1.8 million of unamortized issue costs, \$1.4 million of unamortized discount, along with \$0.2 million of legal fees, which resulted in a loss on extinguishment of debt of \$3.4 million.

The Company reinvested all net proceeds remaining after the repurchase of the \$175.0 million of Notes in replacement collateral under the Indenture within 360 days following the asset disposition, in accordance with the terms of the Indenture.

During the quarter ended December 25, 2016, the Company repurchased and extinguished \$14.5 million of the outstanding Notes, which resulted in a gain of \$0.4 million of \$0.1 million of unamortized issuance cost and \$0.1 million of unamortized discount resulting in a gain on extinguishment of debt of \$0.2 million.

During the quarter ended March 26, 2017, the Company repurchased and extinguished \$62.7 million of the outstanding Notes, which resulted in a loss of \$1.4 million and the realization of \$0.4 million of unamortized issuance cost and \$0.3 million of unamortized discount resulting in a loss on extinguishment of debt of \$2.1 million.

As of March 26, 2017, there was \$372.8 million in Notes outstanding.

(b) Other Indebtedness

\$110.0 Million Credit Agreement

On May 14, 2014, the Company entered into a \$110.0 million Credit and Security Agreement, dated May 14, 2014 (the "Credit Agreement"), with the lenders from time to time party thereto, SunTrust Bank, as Agent (the "Agent"), PNC Bank, National Association, as Joint Lead Arranger and Documentation Agent, and SunTrust Robinson Humphrey, Inc., as Joint Lead Arranger and Sole Book Runner. The Credit Agreement established a five-year senior secured revolving credit facility in the maximum amount of \$110.0 million (subject to a potential increase of the maximum principal amount to \$135.0 million, subject to the Agent's and applicable lenders' approval as described therein), consisting of a subline for letters of credit in an amount not to exceed \$50.0 million, as well as a swingline loan in an aggregate principal amount at any time outstanding not to exceed \$10.0 million. The obligations under the Credit Agreement are secured by a first priority lien on the Company's accounts receivable, inventory, deposit accounts, securities accounts, cash, securities and general intangibles (other than intellectual property). The obligations under the

Credit Agreement are secured by a second priority lien, junior to the lien securing the Notes, on all of the Company's other assets.

The Credit Agreement contains certain covenants, which include, but are not limited to, restrictions on indebtedness, liens, and investments, and limits on other various payments, as well as a financial covenant relating to a minimum fixed charge coverage ratio of 1.15:1 (as modified per the Third Amendment and the Fourth Amendment, as defined and discussed below). Events of default under the terms of the Credit Agreement include, but are not limited to: failure of the Company to pay any principal of any loans in full when due and payable; failure of the Company to pay any interest on any loan or any fee or other amount payable under the Credit Agreement within three business days after the date when due and payable; failure of the Company or any of its subsidiaries to comply with certain covenants and agreements, subject to applicable grace periods and/or notice requirements; any representation, warranty or statement made in or pursuant to the Credit Agreement or any related writing or any other material information furnished by the Company or any of its subsidiaries to the Agent or the lenders proving to be false or erroneous; and the occurrence of an event or condition having or reasonably likely to have a material adverse effect, which includes a material adverse effect on the business, operations, condition (financial or otherwise) or prospects of the Company or the ability of the Company to repay its obligations. Where an event of default arises from certain bankruptcy events, the commitments will automatically and immediately terminate and the principal of, and interest then outstanding on, all of the loans will become immediately due and payable. Subject to certain notice requirements and other conditions, upon the occurrence of an event of default, including the occurrence of a condition having or reasonably likely to have a material adverse effect, commitments may be terminated and the principal of, and interest then outstanding on, all of the loans may become immediately due and payable. As of March 26, 2017, no event of default had occurred and the Company believes that events or conditions having a material adverse effect, giving rise to an acceleration of any amounts outstanding under the Credit Agreement, have not occurred and the likelihood of such events or conditions occurring is remote.

Borrowings under the Credit Agreement may take the form of a base rate revolving loan, Eurodollar revolving loan or swingline loan. Base rate revolving loans and swingline loans will bear interest at a rate per annum equal to the sum of the applicable margin from time to time in effect plus the highest of (i) the Agent's prime lending rate, as in effect at such time, (ii) the federal funds rate, as in effect at such time, plus 0.50% per annum, and (iii) the adjusted London Interbank Offered Rate ("LIBOR") determined at such time for an interest period of one month, plus 1.00% per annum. Eurodollar revolving loans will bear interest at a rate per annum equal to the sum of the applicable margin from time to time in effect plus the adjusted LIBOR. The applicable margin varies between 1.50% and 2.00% for base rate revolving loans and swingline loans and 2.50% and 3.00% for Eurodollar loans, and is based on several factors including the Company's then-existing borrowing base and the lender's total commitment amount and revolving credit exposure. The calculation of the Company's borrowing base takes into account several items relating to the Company and its subsidiaries, including, without limitation, amounts due and owing under billed and unbilled accounts receivables, then-held eligible raw materials inventory, work-in-process inventory, and applicable reserves.

On May 31, 2015, the Company entered into a third amendment (the "Third Amendment") to the Credit Agreement. Under the terms of the Third Amendment, the definitions of certain terms of the Credit Agreement were modified, the disposition of the Herley Entities was approved by the lenders, a minimum \$175.0 million repurchase of the Notes by the Company was required and the payment in full of the outstanding balance of the Credit Agreement was required. Additionally, the measurement of the fixed charge coverage ratio of 1.15:1 was modified as follows: (i) the fixed charge coverage ratio will not be measured as of the end of any quarterly reporting period ending after June 30, 2015, if on such date (a) there are no outstanding revolving loans or swingline loans and (b) the aggregate amount outstanding under letters of credit is less than or equal to \$17.0 million, and (ii) as to any subsequent quarterly reporting period ending after June 30, 2015, and not covered by clause (i) above, a fixed charge coverage ratio of at least 1.05:1 must be maintained if the percentage of (a) outstanding revolving loans plus the sum of the outstanding swingline loans and outstanding letters of credit that are in excess of \$17.0 million, to (b) the revolving credit commitment, minus the Herley Disposition Proceeds Reinvestment Reserve (as defined below) is greater than 0.00% but less than 15.00% or a fixed charge coverage ratio of at least 1.10:1 must be maintained if the aforementioned percentage is equal to or greater than 15.00% but less than 25.00%. In all other instances, a fixed charge coverage ratio of at least 1.15:1 must be maintained. For purposes of computing the fixed charge coverage ratio, the associated reduction in consolidated interest expense in connection with the repurchase of Notes with proceeds from the sale of the Herley Entities shall be deemed to have occurred on the first day of the most recently completed four quarterly reporting periods prior to the sale.

The terms of the Third Amendment also included the establishment of a reserve (the "Herley Disposition Proceeds Reinvestment Reserve") that reduced the maximum \$110.0 million total borrowing base on the Credit Agreement. With the sale of the Herley Entities, a \$50.8 million reserve was established based upon the collateral carrying value under the Credit Agreement of the Herley Entities disposed. The reserve and therefore the maximum borrowing base were adjusted monthly for the subsequent cumulative reinvestment in similar collateral assets over a period not to have exceeded 360 days from the date of sale of the Herley Entities. As of March 26, 2017, there was no reserve on the maximum borrowings, resulting from a cumulative reinvestment in similar collateral assets since the sale of the Herley Entities in excess of the \$50.8 million reserve

established at the date of the sale of the Herley Entities. The Company made investments in assets that replaced the collateral, which reinstated the maximum facility to the full \$110.0 million as of the end of the first quarter of 2016.

On August 20, 2015, the Company entered into a fourth amendment (the "Fourth Amendment") to the Credit Agreement. Among other things, the Fourth Amendment provides for a modification of the Third Amendment as it relates to when the minimum fixed charge coverage ratio will be measured based upon the Company's outstanding borrowings. Outstanding borrowings for purposes of computing the applicable minimum fixed charge coverage ratio exclude any letter of credit exposure outstanding of \$17.0 million plus the amount of letters of credit outstanding for the divested Herley Entities for which a cash deposit has been placed in escrow by the Buyer to cover the amount of such outstanding letters of credit, should the letters of credit be pulled.

As of March 26, 2017, there were no borrowings outstanding on the Credit Agreement and \$10.1 million outstanding on letters of credit, resulting in net borrowing base availability of \$57.4 million. The Company was in compliance with the financial covenants of the Credit Agreement and its amendments as of March 26, 2017.

Debt Acquired in Acquisition

The Company has a \$10.0 million ten-year term loan with a bank in Israel entered into on September 16, 2008 in connection with the acquisition of one of its wholly owned subsidiaries. The balance under the term loan as of March 26, 2017 was \$1.5 million, and the loan is payable in quarterly installments of \$0.3 million plus interest at LIBOR plus a margin of 1.5%. The loan agreement governing the term loan contains various covenants, including a minimum net equity covenant as defined in the loan agreement. The Company was in compliance with all covenants contained in the loan agreement as of March 26, 2017.

Fair Value of Long-term Debt

Carrying amounts and the related estimated fair values of the Company's long-term debt financial instruments not measured at fair value on a recurring basis at March 26, 2017 and December 25, 2016 are presented in the following table:

	As of March 26, 2017							As of December 25, 2016				
\$ in millions		Carrying Principal Amount Fair			air Value	r Value Principal Am						
Total long-term debt including current portion	\$	374.3	\$	370.3	\$	374.3	\$	437.3	\$	432.0	\$	423.6

The fair value of the Company's long-term debt was based upon actual trading activity (Level 1, Observable inputs -quoted prices in active markets).

As of March 26, 2017, the difference between the carrying amount of \$370.3 million and the principal amount of \$374.3 million presented in the table above is the net unamortized original issue discount of \$1.8 million and the unamortized debt issuance costs of \$2.2 million, which are being accreted to interest expense over the term of the related debt. As of December 25, 2016, the difference between the carrying amount of \$432.0 million and the principal amount of \$437.3 million presented in the table above is the net unamortized original issue discount of \$2.4 million and the unamortized debt issuance costs of \$2.9 million, which are being accreted to interest expense over the term of the related debt.

Note 8. Segment Information

The Company operates in three reportable segments. The Kratos Government Solutions ("KGS") reportable segment is comprised of an aggregation of KGS operating segments, including the microwave electronic products, satellite communications, modular systems and defense and rocket support services operating segments. The Unmanned Systems ("US") reportable segment consists of its unmanned aerial system and unmanned ground and seaborne system businesses. The KGS and US segments provide products, solutions and services for mission critical national security programs. KGS and US customers primarily include national security related agencies, the U.S. Department of Defense (the "DoD"), intelligence agencies and classified agencies, and to a lesser degree, international government agencies and domestic and international commercial customers. The Public Safety & Security ("PSS") reportable segment provides independent integrated solutions for advanced homeland security, public safety, critical infrastructure, and security and surveillance systems for government and commercial applications. PSS customers include those in the critical infrastructure, power generation, power transport, nuclear energy, financial, IT, healthcare, education, transportation and petro-chemical industries, as well as certain government and military customers.

The Company organizes its reportable segments based on the nature of the products, solutions and services offered. Transactions between segments are generally negotiated and accounted for under terms and conditions similar to other government and commercial contracts. This presentation is consistent with the Company's operating structure. In the following table total operating income (loss) from continuing operations of the reportable business segments is reconciled to the corresponding consolidated amount. The reconciling item "unallocated corporate expense, net" includes costs for certain stock-based compensation programs (including stock-based compensation costs for stock options, employee stock purchase plan and restricted stock units), the effects of items not considered part of management's evaluation of segment operating performance, merger and acquisition expenses, corporate costs not allocated to the segments, and other miscellaneous corporate activities.

During the three months ended March 27, 2016, the PSS reportable segment recorded a \$1.9 million charge related to a litigation settlement of a contract dispute and the KGS reportable segment recorded a \$3.0 million impairment charge of the carrying value of inventories as a result of the decision to close one of its manufacturing facilities and the exit of certain lower margin product business lines.

Revenues, depreciation and amortization, and operating income (loss) generated by the Company's reportable segments for the three month periods ended March 26, 2017 and March 27, 2016 are as follows (in millions):

	Th	Three Months Ended		
	March 26, 20	6, 2017		ch 27, 2016
Revenues:				
Kratos Government Solutions				
Service revenues	\$	48.2	\$	52.4
Product sales	(57.2		56.2
Total Kratos Government Solutions	1	15.4		108.6
Public Safety & Security				
Service revenues	2	36.8		30.2
Product sales		_		_
Total Public Safety & Security		36.8		30.2
Unmanned Systems				
Service revenues		—		—
Product sales	·	15.6		14.2
Total Unmanned Systems		15.6		14.2
Total revenues	\$ 10	57.8	\$	153.0
Depreciation & amortization:				
Kratos Government Solutions	\$	3.7	\$	4.1
Public Safety & Security		0.1		0.1
Unmanned Systems		1.8		1.9
Total depreciation and amortization	\$	5.6	\$	6.1
Operating income (loss) from continuing operations:				
Kratos Government Solutions	\$	9.6	\$	(1.8)
Public Safety & Security		(0.2)		(2.7)
Unmanned Systems		(5.0)		(4.2)
Total segment operating income (loss)		4.4		(8.7)
Unallocated corporate expense, net		(2.7)		(1.5)
Total operating income (loss) from continuing operations	\$	1.7	\$	(10.2)

Note 9. Significant Customers

Revenue from the U.S. Government, which includes foreign military sales, includes revenue from contracts for which the Company is the prime contractor as well as those for which the Company is a subcontractor and the ultimate customer is the U.S. Government. The KGS and US segments have substantial revenue from the U.S. Government. Sales to the U.S. Government amounted to approximately \$96.3 million and \$95.4 million, or 57% and 62% of total Kratos revenue, for the three months ended March 26, 2017 and March 27, 2016, respectively.

Note 10. Commitments and Contingencies

In addition to commitments and obligations in the ordinary course of business, the Company is subject to various claims, pending and potential legal actions for damages, investigations relating to governmental laws and regulations and other matters arising out of the normal conduct of the Company's business. The Company assesses contingencies to determine the degree of probability and range of possible loss for potential accrual in its condensed consolidated financial statements. An estimated loss contingency is accrued in the Company's condensed consolidated financial statements if it is probable that a liability has been incurred and the amount of the loss can be reasonably estimated. Because litigation is inherently unpredictable and unfavorable resolutions could occur, assessing litigation contingencies is highly subjective and requires judgments about future events. When evaluating contingencies, the Company may be unable to provide a meaningful estimate due to a number of factors, including but not limited to the procedural status of the matter in question, the presence of complex or novel legal theories, and the ongoing discovery and development of information important to the matters. In addition, damage amounts claimed in litigation against it may be unsupported, exaggerated or unrelated to possible outcomes and, as such, are not meaningful indicators of its potential liability. The Company regularly reviews contingencies to determine the adequacy of its accruals and related disclosures. The amount of ultimate loss may differ from these estimates. It is possible that cash flows or results of operations could be materially affected in any particular period by the unfavorable resolution of one or more of these contingencies. Whether any losses finally determined in any claim, action, investigation or proceeding could reasonably have a material effect on the Company's business, financial condition, results of operations or cash flows will depend on a number of variables, including the timing and amount of such losses; the structure and type of any remedies; the monetary significance any such losses. damages or remedies may have on the Company's condensed consolidated financial statements; and the unique facts and circumstances of the particular matter that may give rise to additional factors.

Legal and Regulatory Matters

U.S. Government Cost Claims.

The Company's contracts with the DoD are subject to audit by the Defense Contract Audit Agency ("DCAA"). As a result of these audits, from time to time the Company is advised of claims concerning potential disallowed, overstated or disputed costs. For example, during the course of recent audits of the Company's contracts, the DCAA is closely examining and questioning certain of the established and disclosed practices that it had previously audited and accepted. The Company's personnel regularly scrutinizes costs incurred and allocated to contracts with the U.S. Government for compliance with regulatory standards. On July 28, 2015, the Company received a determination letter from the Defense Contract Management Agency ("DCMA") regarding what the DCMA believed were certain unallowable costs for one of the Company's subsidiaries with respect to fiscal years 2007. In April 2016, the Company reached an agreement with the DCAA to settle matters related to unallowable costs for this subsidiary for fiscal years 2007 and 2008 for approximately \$0.2 million. For those Company subsidiaries and fiscal years which have not yet been audited by the DCAA or for those audits which are in process which have not been completed by the DCAA, the Company cannot reasonably estimate the range of loss, if any, that may result from audits and reviews in which it is currently involved given the inherent difficulty in predicting regulatory action, fines and penalties, if any, and the various remedies and levels of judicial review available to the Company in the event of an adverse finding. As a result, the Company has not recorded any liability related to these matters.

Other Litigation Matters.

The Company is subject to normal and routine litigation arising from the ordinary course and conduct of business and, at times, as a result of acquisitions and dispositions. Such disputes include, for example, commercial, employment, intellectual property, environmental and securities matters. The aggregate amounts accrued related to these matters are not material to the total liabilities of the Company. The Company intends to defend itself in any such matters and does not currently believe that the outcome of any such matters will have a material adverse impact on the Company's financial condition, results of operations or cash flows. In the first quarter of fiscal 2016, the Company recorded a charge of \$1.9 million related to a litigation settlement of a contract dispute in the PSS segment.



Note 11. Condensed Consolidating Financial Statements

The Company has \$372.8 million in outstanding Notes (see Note 7). The Notes are guaranteed by the Subsidiary Guarantors and are collateralized by the assets of all of the Company's 100% owned subsidiaries. The Notes are fully and unconditionally guaranteed on a joint and several basis by each Subsidiary Guarantor and the Company. There are no contractual restrictions with respect to the Notes limiting cash transfers from Subsidiary Guarantors by dividends, loans or advances to the Company. The Notes are not guaranteed by the Company's foreign subsidiaries (the "Non-Guarantor Subsidiaries").

The following tables present condensed consolidating financial statements for the parent company, the Subsidiary Guarantors and the Non-Guarantor Subsidiaries, respectively. The condensed consolidating financial information below follows the same accounting policies as described in the condensed consolidated financial statements, except for the use of the equity method of accounting to reflect ownership interests in 100% owned subsidiaries, which are eliminated upon consolidation.

Condensed Consolidating Balance Sheet March 26, 2017 (Unaudited) (in millions) Subsidiary

	Parent Company		Subsidiary Guarantors on a Combined Basis]	Non-Guarantors on a Combined Basis		Eliminations	Consolidated
Assets								
Current Assets:								
Cash and cash equivalents	\$ 68.9	\$	(1.1)	\$	5.6	\$	—	\$ 73.4
Accounts receivable, net	—		198.1		29.8		—	227.9
Amounts due from affiliated companies	219.8		—		—		(219.8)	—
Inventoried costs	—		45.9		18.9		—	64.8
Other current assets	4.7		13.8		2.7			21.2
Total current assets	293.4		256.7		57.0		(219.8)	387.3
Property, plant and equipment, net	1.5		42.8		6.5			50.8
Goodwill			442.5		42.9			485.4
Intangible assets, net	_		22.3		7.6		_	29.9
Investment in subsidiaries	460.6		67.2				(527.8)	_
Other assets	0.3		7.9					8.2
Total assets	\$ 755.8	\$	839.4	\$	114.0	\$	(747.6)	\$ 961.6
Liabilities and Stockholders' Equity		_		_		_		
Current liabilities:								
Accounts payable	\$ 3.6	\$	43.1	\$	5.1	\$	_	\$ 51.8
Accrued expenses	11.3		44.3		3.8			59.4
Accrued compensation	4.0		26.8		4.4		_	35.2
Billings in excess of costs and earnings on uncompleted contracts			36.9		4.9		_	41.8
Amounts due to affiliated companies			192.0		27.8		(219.8)	_
Other current liabilities	0.9		3.8		1.7		_	6.4
Current liabilities of discontinued operations	1.3		_		0.1		_	1.4
Total current liabilities	21.1	_	346.9	_	47.8		(219.8)	 196.0
Long-term debt, net of current portion	368.8		_		0.5		_	369.3
Other long-term liabilities	10.8		19.9		10.5		_	41.2
Non-current liabilities of discontinued operations	3.7		—		_		—	3.7
Total liabilities	404.4		366.8		58.8		(219.8)	 610.2
Total stockholders' equity	351.4		472.6		55.2		(527.8)	351.4
Total liabilities and stockholders' equity	\$ 755.8	\$	8 839.4	\$	114.0	\$	(747.6)	\$ 961.6

Condensed Consolidating Balance Sheet December 25, 2016 (Unaudited) (in millions)

	Pare	nt Company	Subsidiary Guarantors on a Combined Basis]	Non-Guarantors on a Combined Basis	Eliminations	Consolidated
Assets		<u> </u>					
Current Assets:							
Cash and cash equivalents	\$	67.2	\$ (3.3)	\$	5.2	\$ —	\$ 69.1
Accounts receivable, net		_	197.9		31.5	—	229.4
Amounts due from affiliated companies		204.6	—		—	(204.6)	—
Inventoried costs		_	37.2		18.2	—	55.4
Other current assets		6.3	11.6		1.3	—	19.2
Total current assets		278.1	243.4		56.2	(204.6)	 373.1
Property, plant and equipment, net		1.6	41.7		6.5	—	49.8
Goodwill		—	442.5		42.9	—	485.4
Intangible assets, net		—	24.5		8.1	—	32.6
Investment in subsidiaries		458.0	67.5		_	(525.5)	—
Other assets		0.4	7.3		_	—	7.7
Total assets	\$	738.1	\$ 826.9	\$	5 113.7	\$ (730.1)	\$ 948.6
Liabilities and Stockholders' Equity							
Current liabilities:							
Accounts payable	\$	4.5	\$ 43.7	\$	4.5	\$ —	\$ 52.7
Accrued expenses		5.6	44.5		3.5	—	53.6
Accrued compensation		4.0	31.2		3.9	—	39.1
Billings in excess of costs and earnings on uncompleted contracts		_	38.9		2.9	_	41.8
Amounts due to affiliated companies		—	174.6		30.0	(204.6)	
Other current liabilities		1.4	4.1		2.2	_	7.7
Current liabilities of discontinued operations		1.5	_		0.1	_	1.6
Total current liabilities		17.0	 337.0	_	47.1	 (204.6)	196.5
Long-term debt, net of current portion		430.2	_		0.8	_	431.0
Other long-term liabilities		10.8	19.9		10.3	_	41.0
Non-current liabilities of discontinued operations		3.7	_		_	_	3.7
Total liabilities		461.7	 356.9	_	58.2	 (204.6)	672.2
Total stockholders' equity		276.4	470.0		55.5	(525.5)	276.4
Total liabilities and stockholders' equity	\$	738.1	\$ 826.9	\$	5 113.7	\$ (730.1)	\$ 948.6

Condensed Consolidating Statement of Operations and Comprehensive Income (Loss) Three Months Ended March 26, 2017 (Unaudited) (in millions)

	Parent Company	Subsidiary Guarantors on a Combined Basis	Non-Guarantors on a Combined Basis	Eliminations	Consolidated
Service revenues	\$ —	\$ 82.4	\$ 2.6	\$ —	\$ 85.0
Product sales	—	73.3	12.7	(3.2)	82.8
Total revenues		155.7	15.3	(3.2)	167.8
Cost of service revenues		59.9	1.9		61.8
Cost of product sales	—	54.0	10.1	(3.2)	60.9
Total costs		113.9	12.0	(3.2)	122.7
Gross profit		41.8	3.3		45.1
Selling, general and administrative expenses	2.1	33.8	3.1	—	39.0
Research and development expenses	—	4.0	0.4	_	4.4
Operating income (loss) from continuing operations	(2.1)	4.0	(0.2)		1.7
Other income (expense):					
Interest income (expense), net	(8.2)	_	_	_	(8.2)
Loss on extinguishment of debt	(2.1)	_	_	—	(2.1)
Other income (expense), net	_	_	0.2	_	0.2
Total other income (expense), net	(10.3)	_	0.2		(10.1)
Income (loss) from continuing operations before income taxes	(12.4)	4.0		_	(8.4)
Provision for income taxes from continuing operations	0.1	1.1	0.3	_	1.5
Income (loss) from continuing operations	(12.5)	2.9	(0.3)		(9.9)
Loss from discontinued operations	(0.1)	_		—	(0.1)
Equity in net income (loss) of subsidiaries	2.6	(0.3)	_	(2.3)	
Net income (loss)	\$ (10.0)	\$ 2.6	\$ (0.3)	\$ (2.3)	\$ (10.0)
Comprehensive income (loss)	\$ (9.9)	\$ 2.6	\$ (0.2)	\$ (2.4)	\$ (9.9)

Condensed Consolidating Statement of Operations and Comprehensive Income (Loss) Three Months Ended March 27, 2016 (Unaudited) (in millions)

		Subsidiary Guarantors on a	Non-Guarantors on		
	Parent Company	Combined Basis	a Combined Basis	Eliminations	Consolidated
Service revenues	\$ —	\$ 77.4	\$ 5.2	\$ —	\$ 82.6
Product sales		59.0	13.1	(1.7)	70.4
Total revenues		136.4	18.3	(1.7)	153.0
Cost of service revenues	—	56.7	3.6	—	60.3
Cost of product sales		48.0	10.5	(1.7)	56.8
Total costs		104.7	14.1	(1.7)	117.1
Gross profit	—	31.7	4.2	—	35.9
Selling, general and administrative expenses	1.3	39.6	2.3	_	43.2
Research and development expenses		2.9			2.9
Operating income (loss) from continuing operations	(1.3)	(10.8)	1.9		(10.2)
Other income (expense):					
Interest income (expense), net	(8.6)	(0.1)	—	—	(8.7)
Other income (expense), net			0.3		0.3
Total other income (expense), net	(8.6)	(0.1)	0.3	—	(8.4)
Income (loss) from continuing operations before income taxes	(9.9)	(10.9)	2.2	_	(18.6)
Provision (benefit) for income taxes from continuing operations	0.1	3.2	0.3	_	3.6
Income (loss) from continuing operations	(10.0)	(14.1)	1.9		(22.2)
Equity in net income (loss) of subsidiaries	(12.2)	1.9		10.3	_
Net income (loss)	\$ (22.2)	\$ (12.2)	\$ 1.9	\$ 10.3	\$ (22.2)
Comprehensive income (loss)	\$ (22.2)	\$ (12.2)	\$ 1.9	\$ 10.3	\$ (22.2)

Condensed Consolidating Statement of Cash Flows Three Months Ended March 26, 2017 (Unaudited) (in millions)

	Paron	t Company	Gu	Subsidiary arantors on a mbined Basis	on-Guarantors on a Combined Basis	Eliminations	Consolidated
Net cash provided by (used in) operating activities from continuing operations	\$	(1.8)	\$	(7.7)	\$ 	\$ —	\$ (8.7)
Investing activities:							
Investment in affiliated companies		(15.0)		_	_	15.0	—
Capital expenditures		_		(5.0)	(0.2)	_	(5.2)
Net cash provided by (used in) investing activities from continuing operations		(15.0)		(5.0)	(0.2)	15.0	(5.2)
Financing activities:							
Extinguishment of long-term debt		(64.0)		—	—	—	(64.0)
Repayment of debt				—	(0.3)	—	(0.3)
Proceeds from the issuance of common stock		81.9		—	_	—	81.9
Proceeds from the sale of employee stock purchase plan shares		0.8		_	_	_	0.8
Financings from affiliated companies		—		15.0	—	(15.0)	—
Net cash provided by (used in) financing activities from continuing operations		18.7		15.0	 (0.3)	(15.0)	18.4
Net cash flows of continuing operations		1.9		2.3	 0.3		4.5
Net investing cash flows from discontinued operations		(0.2)		_		_	(0.2)
Net increase in cash and cash equivalents	\$	1.7	\$	2.3	\$ 0.3	\$ _	\$ 4.3

Condensed Consolidating Statement of Cash Flows Three Months Ended March 27, 2016 (Unaudited) (in millions)

	Parent Company	Subsidiary Guarantors on a Combined Basis	Non-Guarantors on a Combined Basis	Eliminations	Consolidated
Net cash provided by (used in) operating activities from continuing operations	\$ 3.9	\$ (15.3)	\$ (0.1)	\$ —	\$ (11.5)
Investing activities:					
Investment in affiliated companies	(15.8)	—	—	15.8	_
Capital expenditures	(0.3)	(1.4)	(0.4)	—	(2.1)
Net cash provided by (used in) investing activities from continuing operations	(16.1)	(1.4)	(0.4)	15.8	(2.1)
Financing activities:					
Repayment of debt	—	—	(0.3)	—	(0.3)
Proceeds from the sale of employee stock purchase plan shares	1.2	_	_	_	1.2
Financing from affiliated companies	—	15.8		(15.8)	—
Net cash provided by (used in) financing activities from continuing operations	1.2	15.8	(0.3)	(15.8)	0.9
Net cash flows of continuing operations	(11.0)	(0.9)	(0.8)	_	(12.7)
Net operating cash flows from discontinued operations	(0.4)	—		—	(0.4)
Net investing cash flows from discontinued operations	4.7	—	—	—	4.7
Net decrease in cash and cash equivalents	\$ (6.7)	\$ (0.9)	\$ (0.8)	\$	\$ (8.4)

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

This Ouarterly Report on Form 10-0 (this "Quarterly Report") contains "forward-looking statements" relating to our future financial performance, the market for our services and our expansion plans and opportunities. In some cases, you can identify forward-looking statements by terminology such as "may," "will," "should," "expect," "plan," "anticipate," "believe," "estimate," "predict," "potential," or "continue," the negative of such terms or other comparable terminology. These forward-looking statements reflect our current beliefs, expectations and projections, are based on assumptions, and are subject to known and unknown risks and uncertainties that could cause our actual results or achievements to differ materially from any future results or achievements expressed in or implied by our forward-looking statements. Many of these factors are beyond our ability to control or predict. As a result, you should not place undue reliance on forward-looking statements. Important risks and uncertainties that could cause our actual results or achievements to differ materially from the results or achievements reflected in our forward-looking statements include, but are not limited to: changes or cutbacks in spending or the appropriation of funding by the federal government, including the U.S. Department of Defense (the "DoD"), which could cause delays, cancellations or reductions of key government contracts; bid protests; changes in the scope or timing of our projects; the timing, rescheduling or cancellation of significant customer contracts and agreements, or consolidation by or the loss of key customers; risks of adverse regulatory action or litigation; risks associated with debt leverage; failure to successfully achieve our integration, cost reduction or divestiture strategies; risks related to security breaches, cybersecurity attacks or other significant disruptions of our information systems; and competition in the marketplace, which could reduce revenues and profit margins, as well as the additional risks and uncertainties described in this Ouarterly Report, in "Item 1A-Risk Factors" of our Annual Report on Form 10-K for the fiscal year ended December 25, 2016 filed with the U.S. Securities and Exchange Commission (the "SEC") on February 27, 2017 (the "Form 10-K"), and in other reports that we have filed with the SEC. These forward-looking statements reflect our views and assumptions only as of the date such forward-looking statements are made. Except as required by law, we assume no responsibility for updating any forward-looking statements, whether as a result of new information, future events or otherwise.

All references to "us," "we," "our," the "Company" and "Kratos" refer to Kratos Defense & Security Solutions, Inc., a Delaware corporation, and its wholly owned subsidiaries.

Overview

We are a mid-tier government contractor at the forefront of the DoD's Innovation Initiative and Third Offset Strategy. We are a leading technology, intellectual property and proprietary product and solution company focused on the U.S. and its allies' national security. We are an industry leader in high performance unmanned aerial drone target systems used to test weapon systems and to train the warfighter, and a provider of high performance unmanned combat aerial systems for force multiplication and amplification. We are also an industry leader in satellite communications, microwave electronics, cyber security/warfare, missile defense and combat systems. We have primarily an engineering and technically oriented work force of approximately 2,900 with a significant number of employees holding national security clearances. Substantially all of our work is performed on a military base, in a secure facility or at a critical infrastructure location. Our primary end customers are national security related agencies. We believe that our technology, intellectual property, proprietary products and designed-in positions on our customers' platforms and systems gives us a competitive advantage and creates a high barrier to entry into our markets. Our entire organization is focused on executing our strategy of becoming the leading technology and intellectual property based company in our industry.

Industry Update

Faced with significant budget pressures, in recent years the U.S. Government has implemented reductions in government spending, including reductions in appropriations for the DoD and other federal agencies, pursuant to the Budget Control Act of 2011 ("BCA"), as amended by the American Taxpayer Relief Act of 2012 and the Bipartisan Budget Act of 2013. Pursuant to the terms of the BCA, a sequestration went into effect in March 2013 resulting in a 7.8% reduction to the DoD budget for fiscal year (the period running from October 1st to September 30th, a "FY") 2013 to \$495.5 billion, excluding funding for military personnel. The DoD budget was approximately \$496.0 billion in FY 2014 and remained at a similar level in FY 2015. The DoD base budget excludes funding for Overseas Contingency Operations ("OCO"), such as those in Afghanistan, Iraq and Syria, which are appropriated separately and are not currently subject to the BCA.

On November 2, 2015, Barack Obama signed the Bipartisan Budget Act of 2015 ("BBA"), formalizing the terms of a two-year budget agreement which raised the U.S. debt ceiling and lifted the sequestration spending caps by \$80.0 billion. Under the budget agreement, the total federal spending increase over the BCA topline funding caps was \$50.0 billion in FY 2016 and \$30.0 billion in FY 2017, with the amounts divided equally between defense and domestic priorities. The overall

discretionary budget was set at \$1.067 trillion in FY 2016 and \$1.07 trillion in FY 2017. The FY 2016 discretionary defense budget was \$548.1 billion, a \$25.0 billion increase over the BCA topline funding caps.

Under the BBA, the Obama Administration received \$33.0 billion of the \$38.0 billion national defense spending increase it sought in FY 2016. In summary the budget agreement:

- extended the BCA out to 2025;
- suspended the U.S. debt limit/ceiling until March 2017;
- increased spending caps for FY 2016 and FY 2017 by \$80.0 billion, including \$50.0 billion in FY 2016 and \$30.0 billion in FY 2017, split evenly between defense and domestic priorities; and
- included a FY 2016 DoD base budget of \$548.0 billion and a FY 2016 OCO budget of \$59.0 billion.

On December 18, 2015, Congress passed and Mr. Obama signed the Consolidated Appropriations Act of 2016, which provided funding for the U.S. Government for FY 2016, providing \$1.1 trillion in discretionary funding for federal agencies through September 2016. Mr. Obama signed a continuing resolution in September 2016, which was extended in December 2016, and provides funding for the U.S. Government at FY 2016 levels through April 28, 2017. On April 28, 2017, Congress passed a one-week stop gap funding bill through May 5, 2017. On May 3, 2017, the U.S. House of Representatives passed a \$1.16 trillion budget bill which proposes a \$15 billion supplement to the DoD base budget for the government fiscal year ended September 30, 2017. The bill was approved by the Senate on May 4, 2017 and is required to be signed by the President by May 5, 2017 before the continuing resolution expires.

Failure to enact appropriations or to extend a continuing resolution would result in a government shutdown of unknown duration. If a prolonged government shutdown were to occur, it could result in program cancellations and stop work orders and could limit our ability to perform on our U.S. Government contracts and the U.S. Government's ability to make timely payments.

In March 2017, the debt ceiling was reached and the Treasury Department began taking "extraordinary measures" to finance the government. It is expected that in mid-to-late 2017 the U.S. Government will exhaust these available measures and Congress will need to raise the debt limit in order for the U.S. Government to continue borrowing money. If the debt ceiling is not raised, we may be required to continue to perform for some period of time on certain of our U.S. Government contracts even if the U.S. Government is unable to make timely payments.

In March 2017, President Trump released a blueprint of his FY 2018 budget request, which seeks \$574 billion for the DoD's base budget and \$65 billion for OCO. The requested budget exceeds the caps established under the BCA. It is unclear whether an annual appropriations bill will be enacted for FY 2018 at the levels proposed by the President and the government may once again at least begin its fiscal year operating under a continuing resolution.

Reportable Segments

The Company operates in three reportable segments. The Kratos Government Solutions ("KGS") reportable segment is comprised of an aggregation of KGS operating segments, including our microwave electronic products, satellite communications, modular systems and defense and rocket support services operating segments. The Unmanned Systems ("US") reportable segment consists of our unmanned aerial system and unmanned ground and seaborne system businesses. The Public Safety & Security ("PSS") reportable segment provides independent integrated solutions for advanced homeland security, public safety, critical infrastructure, and security and surveillance systems for government and commercial applications. We organize our business segments based primarily on the nature of the products, solutions and services offered. Transactions between segments are negotiated and accounted for under terms and conditions similar to other government and commercial contracts, and these intercompany transactions are eliminated in consolidation. For additional information regarding our reportable segments, see Note 8 of the Notes to Condensed Consolidated Financial Statements. From a customer and solutions perspective, we view our business as an integrated whole, leveraging skills and assets wherever possible.

Key Financial Statement Concepts

There have been no changes to our key financial statement concepts for the three months ended March 26, 2017. For a complete description of our business and a discussion of our critical accounting matters, please refer to Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations" in the Form 10-K.

Table of Contents

Comparison of Results for the Three Months Ended March 26, 2017 to the Three Months Ended March 27, 2016

Revenues. Revenues by operating segment for the three months ended March 26, 2017 and March 27, 2016 are as follows (dollars in millions):

	March 26, 2017		1	March 27, 2016	\$ change		% change
Kratos Government Solutions							
Service revenues	\$	48.2	\$	52.4	\$	(4.2)	(8.0)%
Product sales		67.2		56.2		11.0	19.6 %
Total Kratos Government Solutions		115.4		108.6		6.8	6.3 %
Public Safety & Security Service revenues		36.8		30.2		6.6	21.9 %
Unmanned Systems Product sales	_	15.6		14.2		1.4	9.9 %
Total revenues	\$	167.8	\$	153.0	\$	14.8	
Total service revenues	\$	85.0	\$	82.6	\$	2.4	2.9 %
Total product sales		82.8		70.4		12.4	17.6 %
Total revenues	\$	167.8	\$	153.0	\$	14.8	9.7 %

Revenues increased \$14.8 million to \$167.8 million for the three months ended March 26, 2017 from \$153.0 million for the three months ended March 27, 2016. Revenues in our KGS segment increased due to recent contract awards in the satellite communications and training solutions businesses. Revenues in our US segment increased due to recent contract awards for our aerial target products. Growth in our PSS business revenues was driven by security system and related communication equipment integration under a security system deployment for a mass transit authority in a large metropolitan area.

Product sales increased \$12.4 million to \$82.8 million for the three months ended March 26, 2017 from \$70.4 million for the three months ended March 27, 2016, primarily as a result of an increase in product shipments in our KGS segment. As a percentage of total revenue, product sales were 49.3% for the three months ended March 26, 2017 as compared to 46.0% for the three months ended March 27, 2016. Service revenues increased by \$2.4 million to \$85.0 million for the three months ended March 26, 2017 from \$82.6 million for the three months ended March 27, 2016. The increase was primarily related to the work performed in our PSS segment offset by reductions in certain of our services businesses within KGS.

Cost of Revenues. Cost of revenues increased \$5.6 million to \$122.7 million for the three months ended March 26, 2017 from \$117.1 million for the three months ended March 27, 2016. The increase in cost of revenues was primarily a result of the increase in revenues discussed above.

Gross margin increased to 26.9% for the three months ended March 26, 2017 from 23.5% for the three months ended March 27, 2016. Margins on services increased to 27.3% for the three months ended March 26, 2017 from 27.0% for the three months ended March 27, 2016, due primarily to a more favorable mix of revenues. Margins on products increased to 26.4% for the three months ended March 26, 2017 from 19.3% for the three months ended March 27, 2016, primarily due to the mix of products shipped. Margins in the KGS segment increased to 30.1% for the three months ended March 26, 2017 from 23.7% for the three months ended March 27, 2016, primarily as a result of a more favorable mix of revenues, and due to the restructuring actions taken in the first quarter of 2016 in our modular systems business. Margins in the US segment decreased to 9.0% for the three months ended March 26, 2017 from 12.7% for the three months ended March 27, 2016, primarily due to a less favorable mix of products produced and shipped during the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 27.8% for the three months ended March 27, 2016, primarily due to a less favorable mix of products produced and shipped during the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 27.8% for the three months ended March 27, 2016, primarily due to a less favorable mix of products produced and shipped during the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 27.8% for the three months ended March 27, 2016, primarily due to a less favorable mix of products produced and shipped during the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 24.5% for the three months ended March 26, 2017 from 24.5% for the three months ended March 27, 2016, d

Selling, General and Administrative ("SG&A") Expenses. SG&A expense was \$38.7 million for the three months ended March 26, 2017 and \$37.7 million for the three months ended March 27, 2016. As a percentage of revenues, SG&A decreased to 23.1% at March 26, 2017 from 24.6% at March 27, 2016, which is primarily due to the increase in revenues discussed above.

Research and Development ("R&D") Expenses. R&D expenses increased \$1.5 million to \$4.4 million for the three months ended March 26, 2017 from \$2.9 million for the three months ended March 27, 2016, with the primary increases in the Company's satellite and communications business. As a percentage of revenues, R&D increased to 2.6% for the three months

ended March 26, 2017 from 1.9% of revenues in the three months ended March 27, 2016. R&D expenditures are primarily related to investments we are making in conjunction with our customers, with the objectives of the Company's products being the new platform for or "designed-in" to certain new long-term program opportunities and the Company owning certain intellectual property rights for products that support these programs as well as technology upgrades and refresh activities that are necessary for the next generation of our existing product lines.

Unused Office Space, Restructuring Expenses, and Other: The expense of \$0.3 million for the three months ended March 26, 2017 was primarily due to employee termination costs related to personnel reduction actions taken in the first quarter of 2017. The expense of \$5.5 million for the three months ended March 27, 2016 was primarily due to a \$3.0 million impairment charge of the carrying value of inventories that was recorded in the Company's modular systems business as a result of the closure of one of its manufacturing facilities and the exit of certain lower margin product business lines, a \$1.9 million charge related to a litigation settlement of a contract dispute in our PSS business, and employee termination costs related to personnel reduction actions taken in the first quarter of 2016.

Other Expense, Net. Other expense, net increased to \$10.1 million from \$8.4 million for the three months ended March 26, 2017 and March 27, 2016, respectively. The increase in expense of \$1.7 million is primarily related to the loss of \$1.4 million and the realization of \$0.4 million of unamortized issuance cost and \$0.3 million of unamortized discount from the repurchase and extinguishment of \$62.7 million of our Notes (as defined below), resulting in a \$2.1 million loss on extinguishment of debt in the first quarter of 2017.

Provision for Income Taxes. Income tax expense for the three months ended March 26, 2017 and March 27, 2016 was \$1.5 million and \$3.6 million, respectively. For the three months ended March 26, 2017 and March 27, 2016, the expense was a function of the estimated effective tax rate for the year. The estimated effective tax rate is driven by estimated foreign taxes, estimated federal and state taxes, permanent book/tax differences, tax amortization of intangible assets that have an indefinite life under accounting principles generally accepted in the U.S. and the projected income or loss for the year. The income tax expense for the three months ended March 27, 2016 includes a discrete tax expense of \$1.7 million related to an increase in uncertain tax positions.

Loss from Discontinued Operations. The loss from discontinued operations was \$0.1 million for the three months ended March 26, 2017. There was no significant activity for the three months ended March 27, 2016.

Backlog

As of March 26, 2017 and March 27, 2016, our backlog was approximately \$878.3 million and \$909.1 million, respectively, of which \$616.2 million was funded in 2017 and \$553.9 million was funded in 2016. Backlog is our estimate of the amount of revenue we expect to realize over the remaining life of awarded contracts and task orders that we have in hand as of the measurement date. Our total backlog consists of funded and unfunded backlog. We define funded backlog as estimated future revenue under government contracts and task orders for which funding has been appropriated by Congress and authorized for expenditure by the applicable agency, plus our estimate of the future revenue we expect to realize from our commercial contracts that are under firm orders. Our funded backlog does not include the full potential value of our contracts because Congress often appropriates funds to be used by an agency for a particular program of a contract on a yearly or quarterly basis even though the contract may call for performance over a number of years. As a result, contracts typically are only partially funded at any point during their term, and all or some of the work to be performed under the contracts may remain unfunded unless and until Congress makes subsequent appropriation and the procuring agency allocates funding to the contract.

Unfunded backlog reflects our estimate of future revenue under awarded government contracts and task orders for which either funding has not yet been appropriated or the expenditure has not yet been authorized. Our total backlog does not include estimates of revenue from government wide acquisition contracts or General Services Administration schedules beyond awarded or funded task orders, but our unfunded backlog does include estimates of revenue beyond awarded or funded task orders for other types of indefinite delivery or indefinite quantity contracts based on our experience under such contracts and similar contracts. Unfunded backlog also includes priced options, which consist of the aggregate contract revenues expected to be earned as a result of a customer exercising an option period that has been specifically defined in the original contract award.

Contracts undertaken by us may extend beyond one year. Accordingly, portions are carried forward from one year to the next as part of backlog. Because many factors affect the scheduling of projects, no assurance can be given as to when revenue will be realized on projects included in our backlog. Although funded backlog represents only business that is considered to be firm, we cannot guarantee that cancellations or scope adjustments will not occur. The majority of funded backlog represents contracts with terms that would entitle us to all or a portion of our costs incurred and potential fees upon cancellation by the customer.

Management believes that year-to-year comparisons of backlog are not necessarily indicative of future revenues. The actual timing of receipt of revenues, if any, on projects included in backlog could change because many factors affect the scheduling of projects. In addition, cancellation or adjustments to contracts may occur. Backlog is typically subject to large variations from quarter-to-quarter as existing contracts are renewed or new contracts are awarded. Additionally, all U.S. Government contracts included in backlog, whether or not funded, may be terminated at the convenience of the U.S. Government.

Liquidity and Capital Resources

As of March 26, 2017, we had cash and cash equivalents of \$73.4 million compared with cash and cash equivalents of \$69.1 million as of December 25, 2016, which includes \$5.6 million and \$5.2 million, respectively, of cash and cash equivalents held by our foreign subsidiaries. We are not presently aware of any restrictions on the repatriation of these funds; however, earnings of these foreign subsidiaries are considered permanently invested in these foreign subsidiaries. If these funds were needed to fund our operations or satisfy obligations in the U.S. they could be repatriated, and their repatriation into the U.S. may cause us to incur additional U.S. income taxes or foreign withholding taxes. Any additional U.S. income taxes could be offset, in part or in whole, by foreign tax credits. The amount of such taxes and application of tax credits would be dependent on the income tax laws and other circumstances at the time these amounts are repatriated. Based on these variables, it is not practicable to determine the income tax liability that might be incurred if these earnings were to be repatriated. We do not currently intend to repatriate these earnings.

Our total debt, including capital lease obligations, principal due on our Notes, and other term debt decreased from \$432.0 million at December 25, 2016 to \$370.3 million at March 26, 2017, due to the redemption of \$62.7 million of our Notes during the three months ended March 26, 2017, in addition to the principal payment required on our ten-year term loan with a bank in Israel, offset partially by the amortization of the discount on our Notes and the amortization of deferred financing costs.

We use our operating cash flow to finance trade accounts receivable, fund necessary increases in inventory, fund capital expenditures, our internal research and development investments, and our ongoing operations, service our debt and make strategic acquisitions. Financing trade accounts receivable is necessary because, on average, our customers do not pay us as quickly as we pay our vendors and employees for their goods and services since a number of our receivables are contractually billable and due to us only when certain contractual milestones are achieved. Financing increases in inventory balances is necessary to fulfill shipment requirements to meet delivery schedules of our customers. Cash from continuing operations is primarily derived from our customer contracts in progress and associated changes in working capital components. Our days sales outstanding have increased to 124 days at March 26, 2017 from 115 days as of December 25, 2016, primarily as a result of certain contractual billing milestones that have not yet been attained, such as equipment shipments and deliveries on certain products, and for certain flight requirements that must be fulfilled on certain aerial target programs, or final billings which are not due until completion on certain of our large critical infrastructure deployment projects, and therefore we are unable to contractually bill for amounts outstanding related to those milestones at this time. In addition, we used our working capital during the three months ended March 26, 2017 to build inventory of approximately \$9.4 million in anticipation of future scheduled product deliveries.

A summary of our net cash provided by (used in) operating activities from continuing operations, investing activities from continuing operations, and financing activities from continuing operations and our cash flows from discontinued operations from our condensed consolidated statements of cash flows is as follows (in millions):

		Three Months Ended		
	March	26, 2017	Marc	h 27, 2016
Net cash used in operating activities from continuing operations	\$	(8.7)	\$	(11.5)
Net cash used in investing activities from continuing operations		(5.2)		(2.1)
Net cash provided by financing activities from continuing operations		18.4		0.9
Net operating cash flows of discontinued operations		_		(0.4)
Net investing cash flows of discontinued operations	\$	(0.2)	\$	4.7

Net cash used in operating activities from continuing operations for the three months ended March 26, 2017 was positively impacted by increased operating income as compared to the three months ended March 27, 2016, offset by changes in working capital accounts. The net use in working capital accounts for the three months ended March 26, 2017 includes approximately \$2.0 million of internal development investments we are making related to the Low Cost Attritable Unmanned Aerial System Demonstration ("LCASD"). For fiscal 2017, we expect to spend \$7.0 million to \$10.0 million on internal development efforts that are non-capital expenditure related for the LCASD program.

Net cash used in investing activities from continuing operations for the three months ended March 26, 2017 is comprised of capital expenditures, which consist primarily of investments in machinery, computer hardware and software and improvement of our physical properties in order to maintain suitable conditions in which to conduct our business, as well as expenditures related to internal capital investments to build company-owned capital aerial targets and related support equipment and tooling related to the LCASD and UTAP-22 platforms for the three months ended March 26, 2017. Net cash used in investing activities for the three months ended March 27, 2016 is comprised of capital expenditures, which consist primarily of investment in machinery, computer hardware and software and improvement of our physical properties in order to maintain suitable conditions in which to conduct our business. During the three months ended March 26, 2017, capital expenditures of approximately \$2.3 million were incurred in our US business, primarily related to our unmanned combat target initiative. We expect our capital expenditures for fiscal 2017 to be significant due primarily to the capital aerial targets and related support equipment and tooling that we are building related to the LCASD and UTAP-22 platforms in our US business. These capital aerial targets are expected to equal an aggregate of approximately \$18.0 million to \$23.0 million for fiscal 2017.

Net cash provided by financing activities from continuing operations for the three months ended March 26, 2017 was primarily net proceeds of \$81.9 million from our recent completed equity offering of 11.9 million shares of common stock, partially offset by \$64.0 million used to retire approximately \$62.7 million in principal amount of outstanding Notes. Net cash provided by financing activities from continuing operations for the three months ended March 27, 2016 was primarily proceeds from our employee stock purchase plan.

The operating cash flows of discontinued operations for the three month period ended March 27, 2016 is substantially related to the discontinued operations of our U.S. and U.K. Electronic Products Divisions (the "Herley Entities"). The investing cash flows of discontinued operations for the three months ended March 26, 2017 reflects cash used of \$0.2 million, primarily related to the discontinued operations of the Herley Entities. The investing cash flows of discontinued operations for the three months ended March 27, 2016 reflects cash provided of \$4.7 million, primarily related to the reimbursement of taxes related to the sale of the Herley Entities.

Contractual Obligations and Commitments

7.00% Senior Secured Notes due 2019

In May 2014, we refinanced our \$625.0 million of 10% Senior Secured Notes due in 2017 (the "10% Notes") with \$625.0 million of newly issued 7.00% Senior Secured Notes due in 2019 (the "7% Notes"). The net proceeds from the issuance of the 7% Notes was \$618.5 million after an original issue discount of \$6.5 million. We incurred debt issuance costs of \$8.8 million associated with the 7% Notes. We utilized the net proceeds from the issuance of the 7% Notes, a \$41.0 million draw on our Credit Agreement (as defined below), as well as cash from operations to extinguish the 10% Notes. The total reacquisition price of the 10% Notes was \$661.5 million including a \$31.2 million early termination fee, the write-off of \$15.5 million of unamortized issue costs, \$12.9 million of unamortized premium, along with \$5.3 million of additional interest while in escrow, which resulted in a loss on extinguishment of debt of \$39.1 million.

We completed the offering of the 7% Notes in a private placement conducted pursuant to Rule 144A and Regulation S under the Securities Act of 1933, as amended (the "Act"). On October 16, 2014, the Company exchanged the outstanding 7% Notes for an equal amount of new 7.00% Senior Secured Notes due in 2019 (the "Notes") that had been registered under the Act. The terms of the Notes issued in the exchange offer were identical in all material respects to the terms of the 7% Notes, except the Notes issued in the exchange offer had been registered under the Act.

The Notes are governed by an Indenture, dated May 14, 2014 (the "Indenture"), among the Company, certain of the Company's subsidiaries (the "Subsidiary Guarantors") and Wilmington Trust, National Association, as Trustee and Collateral Agent. A Subsidiary Guarantor can be released from its Guarantee (as defined in the Indenture) if (a) all of the capital stock issued by such Subsidiary Guarantor or all or substantially all of the assets of such Subsidiary Guarantor are sold or otherwise disposed of; (b) the Company designates such Subsidiary Guarantor as an Unrestricted Subsidiary (as defined in the Indenture); (c) if the Company exercises its legal defeasance option or its covenant defeasance option; or (d) upon satisfaction and discharge of the Indenture or payment in full in cash of the principal of, premium, if any, and accrued and unpaid interest on the Notes.

The holders of the Notes have a first priority lien on substantially all of the Company's assets and the assets of the Subsidiary Guarantors, except with respect to accounts receivable, inventory, deposit accounts, securities accounts, cash, securities and general intangibles (other than intellectual property), on which the holders of the Notes have a second priority lien to our \$110.0 million Credit Agreement.

We pay interest on the Notes semi-annually, in arrears, on May 15 and November 15 of each year.

The Notes include customary covenants and events of default as well as a consolidated fixed charge ratio of 2.0:1 for the incurrence of additional indebtedness. Negative covenants include, among other things, limitations on additional debt, liens, negative pledges, investments, dividends, stock repurchases, asset sales and affiliate transactions. Events of default include, among other events, non-performance of covenants, breach of representations, cross-default to other material debt, bankruptcy, insolvency, material judgments and changes in control. As of March 26, 2017, we were in compliance with the covenants contained in the Indenture governing the Notes.

We may redeem some or all of the Notes at 105.25% of the aggregate principal amount of such Notes through May 15, 2017, 102.625% of the aggregate principal amount of such Notes if redeemed on or after May 16, 2017 but on or before May 15, 2018 and 100% of the aggregate principal amount of such Notes if redeemed on or after May 16, 2018, plus accrued and unpaid interest to the date of redemption.

The terms of the Indenture require that the net cash proceeds from asset dispositions be utilized to (i) repay or prepay amounts outstanding under our Credit Agreement unless such amounts are reinvested in similar collateral, (ii) make an investment in assets that replace the collateral of the Notes or (iii) a combination of both clauses (i) and (ii). To the extent there are any remaining net proceeds from the asset disposition after application of clauses (i) and (ii), such amounts are required to be utilized to repurchase Notes at par after 360 days following the asset disposition.

Following the sale of the Herley Entities we paid down the \$41.0 million outstanding under the Credit Agreement and repurchased \$175.0 million of the Notes at par, in accordance with the terms of the Indenture.

The total reacquisition price of the Notes was \$178.4 million including the write off of \$1.8 million of unamortized issue costs, \$1.4 million of unamortized discount, along with \$0.2 million of legal fees, which resulted in a loss on extinguishment of debt of \$3.4 million.

The Company reinvested all net proceeds remaining after the repurchase of the \$175.0 million of Notes in replacement collateral under the Indenture within 360 days following the asset disposition, in accordance with the terms of the Indenture.

During the quarter ended December 25, 2016, the Company repurchased and extinguished \$14.5 million of the outstanding Notes, which resulted in a gain of \$0.4 million offset by \$0.1 million of unamortized issuance cost and \$0.1 million of unamortized discount resulting in a net gain of \$0.2 million.

During the quarter ended March 26, 2017, the Company repurchased and extinguished \$62.7 million of the outstanding Notes, which resulted in a loss of \$1.4 million and the realization of \$0.4 million of unamortized issuance cost and \$0.3 million of unamortized discount resulting in a net loss of \$2.1 million.

As of March 26, 2017, there was \$372.8 million in Notes outstanding.

Other Indebtedness

\$110.0 Million Credit Agreement

On May 14, 2014, we entered into a \$110.0 million Credit and Security Agreement, dated May 14, 2014 (the "Credit Agreement"), with the lenders from time to time party thereto, SunTrust Bank, as Agent (the "Agent"), PNC Bank, National Association, as Joint Lead Arranger and Documentation Agent, and SunTrust Robinson Humphrey, Inc., as Joint Lead Arranger and Sole Book Runner. The Credit Agreement established a five-year senior secured revolving credit facility in the maximum amount of \$110.0 million (subject to a potential increase of the maximum principal amount to \$135.0 million, subject to the Agent's and applicable lenders' approval as described therein), consisting of a subline for letters of credit in an amount not to exceed \$50.0 million, as well as a swingline loan in an aggregate principal amount at any time outstanding not to exceed \$10.0 million. The obligations under the Credit Agreement are secured by a first priority lien on our accounts receivable, inventory, deposit accounts,

securities accounts, cash, securities and general intangibles (other than intellectual property). The obligations under the Credit Agreement are secured by a second priority lien, junior to the lien securing the Notes, on all of our other assets.

The Credit Agreement contains certain covenants, which include, but are not limited to, restrictions on indebtedness, liens, and investments, and limits on other various payments, as well as a financial covenant relating to a minimum fixed charge coverage ratio of 1.15:1 (as modified per the Third Amendment and the Fourth Amendment, as defined and discussed below). Events of default under the terms of the Credit Agreement include, but are not limited to: our failure to pay any principal of any loans in full when due and payable; our failure to pay any interest on any loan or any fee or other amount payable under the Credit Agreement within three business days after the date when due and payable; our failure or the failure of any of our subsidiaries to comply with certain covenants and agreements, subject to applicable grace periods and/or notice requirements; any representation, warranty or statement made in or pursuant to the Credit Agreement or any related writing or any other material information furnished by us or any of our subsidiaries to the Agent or the lenders proving to be false or erroneous; and the occurrence of an event or condition having or reasonably likely to have a material adverse effect, which includes a material adverse effect on the business, operations, condition (financial or otherwise) or our prospects or our ability to repay our obligations. Where an event of default arises from certain bankruptcy events, the commitments will automatically and immediately terminate and the principal of, and interest then outstanding on, all of the loans will become immediately due and payable. Subject to certain notice requirements and other conditions, upon the occurrence of a event of default, including the occurrence of a condition having or reasonably likely to have a material adverse effect, commitments may be terminated and the principal of, and interest then outstanding on, all of the loans will become immediately due and payable. Subject to certain notice requirements and other conditions, upon

Borrowings under the Credit Agreement may take the form of a base rate revolving loan, Eurodollar revolving loan or swingline loan. Base rate revolving loans and swingline loans will bear interest at a rate per annum equal to the sum of the applicable margin from time to time in effect plus the highest of (i) the Agent's prime lending rate, as in effect at such time, (ii) the federal funds rate, as in effect at such time, plus 0.50% per annum, and (iii) the adjusted London Interbank Offered Rate ("LIBOR") determined at such time for an interest period of one month, plus 1.00% per annum. Eurodollar revolving loans will bear interest at a rate per annum equal to the sum of the applicable margin from time to time in effect plus the adjusted LIBOR. The applicable margin varies between 1.50% and 2.00% for base rate revolving loans and swingline loans and 2.50% and 3.00% for Eurodollar loans, and is based on several factors including our then-existing borrowing base and the lender's total commitment amount and revolving credit exposure. The calculation of our borrowing base takes into account several items relating to us and our subsidiaries, including, without limitation, amounts due and owing under billed and unbilled accounts receivables, then-held eligible raw materials inventory, work-in-process inventory, and applicable reserves.

On May 31, 2015, we entered into a third amendment (the "Third Amendment") to the Credit Agreement. Under the terms of the Third Amendment, the definitions of certain terms of the Credit Agreement were modified, the disposition of the Herley Entities was approved by the lenders, a minimum \$175.0 million repurchase of the Notes by us was required, and the payment in full of the outstanding balance of the Credit Agreement was required. Additionally, the measurement of the fixed charge coverage ratio of 1.15:1 was modified as follows: (i) the fixed charge coverage ratio will not be measured as of the end of any quarterly reporting period ending after June 30, 2015, if on such date (a) there are no outstanding revolving loans or swingline loans and (b) the aggregate amount outstanding under letters of credit is less than or equal to \$17.0 million, and (ii) as to any subsequent quarterly reporting period ending after June 30, 2015, and not covered by clause (i) above, a fixed charge coverage ratio of at least 1.05:1 must be maintained if the percentage of (a) outstanding revolving loans plus the sum of the outstanding swingline loans and outstanding letters of credit that are in excess of \$17.0 million, to (b) the revolving credit commitment, minus the Herley Disposition Proceeds Reinvestment Reserve (as defined below) is greater than 0.00% but less than 15.00% or a fixed charge coverage ratio of at least 1.10:1 must be maintained if the aforementioned percentage is equal to or greater than 15.00% but less than 25.00%. In all other instances, a fixed charge coverage ratio of at least 1.15:1 must be maintained. For purposes of computing the fixed charge coverage ratio, the associated reduction in consolidated interest expense in connection with the repurchase of Notes with proceeds from the sale of the Herley Entities shall be deemed to have occurred on the first day of the most recently completed four quarterly reporting periods prior to the sale.

The terms of the Third Amendment also included the establishment of a reserve (the "Herley Disposition Proceeds Reinvestment Reserve") that reduced the maximum \$110.0 million total borrowing base on the Credit Agreement. With the sale of the Herley Entities, a \$50.8 million reserve was established based upon the collateral carrying value under the Credit Agreement of the Herley Entities disposed. The reserve and therefore the maximum borrowing base were adjusted monthly for the subsequent cumulative reinvestment in similar collateral assets over a period not to have exceeded 360 days from the date of sale of the Herley Entities. As of March 26, 2017, there was no reserve on the maximum borrowings, resulting from a cumulative reinvestment in similar collateral assets since the sale of the Herley Entities in excess of the \$50.8 million reserve

established at the date of the sale of the Herley Entities. The Company made investments in assets that replaced the collateral, which reinstated the maximum facility to the full \$110.0 million as of the end of the first quarter of 2016.

On August 20, 2015, we entered into a fourth amendment (the "Fourth Amendment") to the Credit Agreement. Among other things, the Fourth Amendment provides for a modification of the Third Amendment as it relates to when the minimum fixed charge coverage ratio will be measured based upon our outstanding borrowings. Outstanding borrowings for purposes of computing the applicable minimum fixed charge coverage ratio exclude any letter of credit exposure outstanding of \$17.0 million plus the amount of letters of credit outstanding for the divested Herley Entities for which a cash deposit has been placed in escrow by the Buyer to cover the amount of such outstanding letters of credit, should the letters of credit be pulled.

As of March 26, 2017, there were no borrowings outstanding on the Credit Agreement and \$10.1 million was outstanding on letters of credit, resulting in net borrowing base availability of \$57.4 million. We were in compliance with the financial covenants of the Credit Agreement and its amendments as of March 26, 2017.

Debt Acquired in Acquisition

We assumed a \$10.0 million ten-year term loan with a bank in Israel entered into on September 16, 2008 in connection with the acquisition of one of its wholly owned subsidiaries. The balance under the term loan as of March 26, 2017 was \$1.5 million, and the loan is payable in quarterly installments of \$0.3 million plus interest at LIBOR plus a margin of 1.5%. The loan agreement governing the term loan contains various covenants, including a minimum net equity covenant as defined in the loan agreement. We were in compliance with all covenants contained in the loan agreement as of March 26, 2017.

Other Liquidity Matters

We believe that our cash on hand, together with funds available under the Credit Agreement and cash expected to be generated from operating activities, will be sufficient to fund our anticipated working capital and other cash needs for at least the next 12 months.

As discussed in Part I, Item 1A, "Risk Factors" of the Form 10-K, our quarterly and annual operating results have fluctuated in the past and may vary in the future due to a variety of factors, many of which are external to our control. If the conditions in our industry deteriorate or our customers cancel or postpone projects or if we are unable to sufficiently increase our revenues or further reduce our expenses, we may experience, in the future, a significant long-term negative impact to our financial results and cash flows from operations. In such a situation, we could fall out of compliance with our financial and other covenants, which, if not waived, could limit our liquidity and capital resources.

Critical Accounting Principles and Estimates

The foregoing discussion of our financial condition and results of operations is based on the condensed consolidated financial statements included in this Quarterly Report. The preparation of these condensed consolidated financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, sales and expenses, and the related disclosures of contingencies. We base these estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities. Actual results may differ from these estimates.

There have been no significant changes to our "Critical Accounting Policies or Estimates" as compared to the significant accounting policies described in the Form 10-K.

Item 3. Quantitative and Qualitative Disclosures About Market Risk.

We are exposed to market risk, primarily related to interest rates and foreign currency exchange rates.

Exposure to market risk for changes in interest rates relates to our outstanding debt. We are exposed to interest rate risk, primarily through our borrowing activities under the Credit Agreement discussed under "Contractual Obligations and Commitments" above. As we do not have a current outstanding balance, a 1% change in the LIBOR would not significantly impact our financial position. We manage exposure to these risks through our operating and financing activities and, when deemed appropriate, through the use of derivative financial instruments. Derivative financial instruments are viewed as risk management tools and are not used for speculation or for trading purposes. Derivative financial instruments would be contracted with investment grade counterparties to reduce exposure to nonperformance.

Exposure to market risk for foreign currency exchange rate risk is related to receipts from customers, payments to suppliers and intercompany loans denominated in foreign currencies. We currently enter into limited foreign currency forward contracts to manage foreign currency exchange rate risk because to date exchange rate fluctuations have had minimal impact on our operating results and cash flows.

Our cash and cash equivalents as of March 26, 2017 were \$73.4 million and are primarily invested in money market interest bearing accounts. A hypothetical 10% adverse change in the average interest rate on our money market cash investments and short-term investments would have had no material effect on our net loss for the three months ended March 26, 2017.

Item 4. Controls and Procedures.

Conclusions Regarding the Effectiveness of Disclosure Controls and Procedures

We maintain disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) promulgated under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), designed to ensure that information required to be disclosed in our reports filed under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and that such information is accumulated and communicated to our management, including our Principal Executive Officer and Principal Financial Officer, as appropriate, to allow timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures, management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management necessarily was required to apply its judgment in evaluating the cost benefit relationship of possible controls and procedures.

As required by Rule 13a-15(b) promulgated under the Exchange Act, we carried out an evaluation, under the supervision and with the participation of our management, including our Principal Executive Officer and Principal Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures as of the end of the period covered by this report.

Based on the foregoing, our Principal Executive Officer and Principal Financial Officer concluded that our disclosure controls and procedures were effective at the reasonable assurance level as of March 26, 2017.

Changes in Internal Control Over Financial Reporting

We operate under the COSO (Committee of Sponsoring Organizations) 2013 Framework. There was no change in our internal control over financial reporting during the three months ended March 26, 2017 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings.

See Note 10 of the Notes to the Condensed Consolidated Financial Statements contained within this Quarterly Report for a discussion of our legal proceedings.

Item 1A. Risk Factors.

In evaluating us and our common stock, we urge you to carefully consider the risks and other information in this Quarterly Report, as well as the risk factors disclosed in Item 1A. to Part I of the Form 10-K, and other reports that we have filed with the SEC. Any of the risks discussed in such reports, as well as additional risks and uncertainties not currently known to us or that we currently deem immaterial, could materially and adversely affect our results of operations, financial condition or prospects. There have been no material changes in our risk factors as previously disclosed in the Form 10-K during the period covered by this Quarterly Report.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults Upon Senior Securities.

None.

Item 4. Mine Safety Disclosures.

Not applicable.

Item 5. Other Information.

On May 1, 2017, the Company entered into an office lease (the "Lease") with TPP 212 Scripps, LLC (the "Landlord") regarding the Company's new corporate headquarters in San Diego, California.

The premises being rented by the Company consist of approximately 26,192 square feet of office space. The Lease has an initial term of 10 years and is anticipated to commence in October 2018. The Lease can be extended for one 5-year term, which must be exercised by giving written notice to the Landlord not earlier than 12 months and not later than 10 months prior to the expiration of the initial term.

The base rent under the Lease is initially approximately \$817,200 per year, escalating over the 10 years at approximately 3.0% per annum. In addition to the base rent, the Company will pay additional rent for the Company's proportionate share of operating expenses, taxes, utilities and insurance expenses for the complex in which the premises are located. The Lease includes standard contract terms and conditions, including without limitation, provisions relating to insurance, indemnity, casualty, defaults and assignment and subletting.

The foregoing description of the Lease is not complete and is qualified in its entirety by reference to the full text of such agreement, a complete copy of which is filed herewith as Exhibit 10.2 and incorporated herein by reference.

Table of Contents

Item 6. Exhibits.

			Incorporated by Reference		
xhibit umber	Exhibit Description	Form	Filing Date/ Period End Date	Exhibit	Filed- Furnished Herewith
2.1#†	Stock Purchase Agreement, dated May 31, 2015, among Kratos Defense & Security Solutions, Inc., Herley Industries, Inc., Ultra Electronics Defense Inc. and Ultra Electronics Holdings plc.	10-Q	08/06/2015 (001-34460)	2.4	
3.1	Amended and Restated Certificate of Incorporation of Kratos Defense & Security Solutions, Inc., as amended.	10-K	02/27/2017 (001-34460)	3.1	
3.2	Second Amended and Restated Bylaws of Kratos Defense & Security Solutions, Inc., as amended.	10-К	02/27/2017 (001-34460)	3.2	
4.1	Specimen Stock Certificate.	10-К	02/27/2017 (001-34460)	4.1	
4.2	Indenture, dated as of May 14, 2014, among Kratos Defense & Security Solutions, Inc., as Issuer, the Guarantors party thereto, and Wilmington Trust, National Association, as Trustee and Collateral Agent (including the Form of 7.00% Senior Secured Notes due 2019).	8-K	05/15/2014 (001-34460)	4.1	
4.3	Registration Rights Agreement, dated as of May 14, 2014, among Kratos Defense & Security Solutions, Inc., as Issuer, the Guarantors party thereto, and SunTrust Robinson Humphrey, Inc., as Representative of the Initial Purchasers.	8-K	05/15/2014 (001-34460)	10.1	
10.1	Employment Agreement, effective January 1, 2017, by and between Kratos Defense & Security Solutions, Inc. and Phil Carrai.	8-K	12/08/2016 (001-34460)	10.1	
10.2	Office Lease, dated as of May 1, 2017, by and between Kratos Defense & Security Solutions, Inc. and TPP 212 Scripps, LLC.				*
31.1	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes Oxley Act of 2002.				*
31.2	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes Oxley Act of 2002.				*
32.1	Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 for Eric M. DeMarco.				*
32.2	Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 for Deanna Lund.				*

	_		Incorporated by Reference Filing Date/		Filed-
Exhibit Number	Exhibit Description	Form	Period End Date	Exhibit	Furnished Herewith
101	Financial statements from the Quarterly Report on Form 10-Q of Kratos Defense & Security Solutions, Inc. for the quarter ended March 26, 2017 formatted in XBRL: (i) the Condensed Consolidated Balance Sheets, (ii) the Condensed Consolidated Statements of Operations and Comprehensive Loss, (iii) the Condensed Consolidated Statements of Cash Flows, (iv) the Notes to the Condensed Consolidated Financial Statements.				*

[#] Certain schedules and exhibits referenced in this document have been omitted in accordance with Item 601(b)(2) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished supplementally to the Securities and Exchange Commission upon request.

[†] This Exhibit has been filed separately with the Secretary of the Securities and Exchange Commission without the redaction pursuant to a Confidential Treatment Request under Rule 24b-2 of the Securities Exchange Act of 1934, as amended.

SIGNATURES

By:

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

/s/ ERIC M. DEMARCO

Eric M. DeMarco Chief Executive Officer, President (Principal Executive Officer)

By: /s/ DEANNA H. LUND, CPA

Deanna H. Lund Executive Vice President, Chief Financial Officer (Principal Financial Officer)

By: /s/ MARIA CERVANTES DE BURGREEN, CPA

Maria Cervantes de Burgreen Vice President and Corporate Controller (Principal Accounting Officer)

Date: May 4, 2017

SCRIPPS PLAZA

OFFICE LEASE

TPP 212 SCRIPPS, LLC, a Delaware limited liability company,

as Landlord,

and

KRATOS DEFENSE & SECURITY SOLUTIONS, INC., a Delaware corporation,

as Tenant.

SUMMARY OF BASIC LEASE INFORMATION

The parties hereto agree to the following terms of this Summary of Basic Lease Information (the "Summary"). This Summary is hereby incorporated into and made a part of the attached Office Lease (this Summary and the Office Lease to be known collectively as the "Lease") which pertains to the office building located at 10680 Treena Street, San Diego, California 92131. Each reference in the Office Lease to any term of this Summary shall have the meaning as set forth in this Summary for such term. In the event of a conflict between the terms of this Summary and the Office Lease, the terms of the Office Lease shall prevail. Any capitalized terms used herein and not otherwise defined herein shall have the meaning as set forth in the Office Lease.

TERMS OF LEASE (References are to the Office Lease)

1. Date:

2. Landlord:

3. Address of Landlord (Section 30.11):

4. Tenant:

DESCRIPTION

May 1, 2017. TPP 212 SCRIPPS, LLC, a Delaware limited liability company TPP 212 Scripps, LLC c/o Cypress Asset Management, Inc. 10650 Treena Street, Suite 109 San Diego, California 92131 Attention: Ron Lack

With a copy to:

Cushman & Wakefield 10650 Treena Street, Suite 101 San Diego, California 92131 Attention: Property Manager

With a copy to:

TPP 212 Scripps, LLC c/o TriGate Capital, LLC 750 N. St. Paul Street, Suite 900 Dallas, Texas 75201 Attention: Asset Manager KRATOS DEFENSE & SECURITY SOLUTIONS, INC., a Delaware corporation.

5. Address of Tenant (Section 30.11):	4820 Eastgate Mall, Suite 200 San Diego, California 92121 Attention: General Counsel
	(Prior to Lease Commencement Date) and
	10680 Treena Street, Suite 600 San Diego, California 92131 Attention: General Counsel (After Lease Commencement Date)
6. Premises (Article 1):	Approximately 26,192 rentable square feet of space located in Suite 600 on the sixth (6^{th}) floor of the Building, as set forth in <u>Exhibit A</u> attached hereto.
7. Term (Article 2).	
7.1 Lease Term:	Ten (10) years.
7.2 Lease Commencement	October 1, 2018, subject to the Beneficial Occupancy Period as set forth in Section 2.3 below.
Date:	

8. Base Rent (Article 3):

Monthly <u>Installment of Base Rent</u>	Monthly Rental Rate <u>per Rentable Square Foot</u>
\$68,099.20	\$2.60
\$70,194.56	\$2.68
\$72,289.92	\$2.76
\$74,385.28	\$2.84
\$76,742.56	\$2.93
\$78,837.92	\$3.01
\$81,195.20	\$3.10
\$83,814.40	\$3.20
\$86,171.68	\$3.29
\$88,790.88	\$3.39
	Installment of Base Rent \$68,099.20 \$70,194.56 \$72,289.92 \$74,385.28 \$76,742.56 \$78,837.92 \$81,195.20 \$83,814.40 \$86,171.68

*In addition to Base Rent, Tenant will pay for electricity supplied to the Premises on a net basis as more fully described in Section 6.1.2 below.

9. Additional Rent (Article 4).
9.1 Base Year:
9.2 Tenant's Share:
10. Security Deposit (Article 21):

11. Parking Pass Ratio (Article 28):

12. Broker (Section 30.15): Calendar year 2019. Approximately 17.06%. \$88,790.88.

Four (4) parking passes for every 1,000 rentable square feet of the Premises (i.e., one hundred five (105) unreserved parking passes); provided that Tenant shall have the right to convert up to ten (10) of Tenant's parking passes into reserved parking spaces designated for Tenant's use at a location within the covered parking structure designated by Landlord from time to time (which initial locations are shown on Exhibit E attached hereto).

Kidder Matthews (for Tenant) Jones Lang LaSalle (for Landlord)

TABLE OF CONTENTS

<u>Page</u>

1	REAL PROPERTY, BUILDING AND PREMISES	1
2	LEASE TERM	4
3	BASE RENT	6
4	ADDITIONAL RENT	7
5	USE OF PREMISES	13
6	SERVICES AND UTILITIES	14
7	REPAIRS	18
8	ADDITIONS AND ALTERATIONS	19
9	COVENANT AGAINST LIENS	21
10	INSURANCE	21
11	DAMAGE AND DESTRUCTION	23
12	NONWAIVER	25
13	CONDEMNATION	25
14	ASSIGNMENT AND SUBLETTING	26
15	SURRENDER OF PREMISES AND REMOVAL OF TENANT'S PROPERTY	29
16	HOLDING OVER	29
17	ESTOPPEL CERTIFICATES	29
18	SUBORDINATION	29
19	DEFAULTS	30
20	COVENANT OF QUIET ENJOYMENT	32
21	SECURITY DEPOSIT	33
22	INTENTIONALLY OMITTED	33
23	SIGNS	33

i

TABLE OF CONTENTS (cont.)

		<u>Page</u>
24	COMPLIANCE WITH LAW	35
25	LATE CHARGES	36
26	LANDLORD'S RIGHT TO CURE DEFAULT	36
27	ENTRY BY LANDLORD	36
28	TENANT PARKING	37
29	HAZARDOUS MATERIALS	38
30	MISCELLANEOUS PROVISIONS	39

ii

<u>INDEX</u>

	Pages
Abatement Event	17
Abatement Notice	17
ACMs	20
Additional Rent.	7
Affiliate	28
Affiliated Assignee	28
Alterations	19
Applicable Laws	13
Approved Working Drawings	Exhibit B
Architect	Exhibit B
Base Rent	6
Base Year	7
Base, Shell and Core	Exhibit B
Beneficial Occupancy Period	6
Brokers	41
Building	1
Calendar Year	7
CASp Report	36
CC&Rs	1
Code	Exhibit B
Common Areas	1
Construction Drawings	Exhibit B
Contract	Exhibit B
Contractor	Exhibit B
Control	28
Cosmetic Alterations	19
Damage Repair Estimate	24
Election Date	3
Eligibility Period	18
Engineers	Exhibit B
Environmental Laws	38
Estimate	12
Estimate Statement	11
Estimated Excess	12
Excess	7
Expense Year	7
FF&E Allowance	Exhibit B
Final Costs	Exhibit B
Final Costs Statement	Exhibit B
Final Retention	Exhibit B
Final Space Plan	Exhibit B
Final Working Drawings	Exhibit B
First Refusal Notice	2
First Refusal Space	2

	Pages
Force Majeure	40
Generator	16
Hazardous Materials	38
Holidays	14
HVAC	14
Improvement Allowance	Exhibit B
Improvement Allowance Items	Exhibit B
Improvements	Exhibit B
Indemnified Claims	21
Insurance Expenses	7
Interest Notice	5
Landlord	1
Landlord Indemnified Parties	21
Lease	1
Lease Commencement Date	4
Lease Term	4
Lease Year	4
Notices	40
Objectionable Name	35
Operating Expenses	7
Option Notice	4
Option Rent	5
Option Rent Notice	5
Option Term	4
Original Tenant	2
Other Building	1
Outside Agreement Date	5
Over-Allowance Amount	Exhibit B
Package Units	16
Parking Facilities	1
Permits	Exhibit B
Premises	1
Project Manager Allowance	Exhibit B
Real Property.	1
Refurbished Improvements	20
Refurbishment Allowance	20
Renovations	42
Rent	7
Review Period	12
Rules and Regulations	13
Security Deposit	32
Signage	33
Signage Specifications	34
SNDA	30
Specifications	Exhibit B

	Pages
Standard Improvement Package	Exhibit B
Statement	11
Subject Space	26
Subleasing Costs	27
Summary	1
Superior Leases	2
Superior Rights	2
Systems and Equipment	10
Tax Expenses	11
Tenant	1
Tenant Improvements.	1
Tenant Parties	21
Tenant's Agents	Exhibit B
Tenant's Election Notice	3
Tenant's Share	11
Terms	2
Test Fit Allowance	Exhibit B
Transfer Notice	26
Transfer Premium	27
Transferee	26
Transfers	26
Utility Expenses	11

v

OFFICE LEASE

This Office Lease, which includes the preceding Summary of Basic Lease Information (the "Summary") attached hereto and incorporated herein by this reference (the Office Lease and Summary to be known sometimes collectively hereafter as the "Lease"), dated as of the date set forth in Section 1 of the Summary, is made by and between TPP 212 SCRIPPS, LLC, a Delaware limited liability company ("Landlord"), and KRATOS DEFENSE & SECURITY SOLUTIONS, INC., a Delaware corporation ("Tenant").

1. REAL PROPERTY, BUILDING AND PREMISES

1.1 <u>Real Property, Building and Premises</u>. Upon and subject to the terms, covenants and conditions hereinafter set forth in this Lease, Landlord hereby leases to Tenant and Tenant hereby leases from Landlord the premises set forth in Section 6 of the Summary (the "**Premises**"), which Premises are located in the "Building," as that term is defined in this Section 1.1. The outline of the floor plan of the Premises is set forth in Exhibit A attached hereto. The Premises are a part of the building known as Scripps Plaza located and addressed at 10680 Treena Street, San Diego, California 92131 ("**Building**"). The Building, the building located at 10650 Treena Street (the "**Other Building**"), the parking structure and facilities serving the Building and the Other Building ("**Parking Facilities**"), the outside plaza areas, land and other improvements surrounding the Building and the Other Building which are designated from time to time by Landlord as common areas appurtenant to or servicing the Building and the Other Building, and the land upon which any of the foregoing are situated, are herein sometimes collectively referred to as the "**Real Property**." Tenant is hereby granted the right to the nonexclusive use of the common corridors and hallways, stairwells, elevators, restrooms and other public or common areas located on the Real Property ("**Common Areas**"), subject to compliance with any reciprocal easement agreements, covenants, conditions and restrictions or similar restrictions on the Real Property. Landlord reserves the right to make alterations or additions to or to change the location of elements of the Real Property.

1.2 <u>Condition of the Premises</u>. Except as specifically set forth in this Lease and in the Tenant Work Letter attached hereto as <u>Exhibit B</u>, Landlord shall not be obligated to provide or pay for any improvement work or services related to the improvement of the Premises. Tenant also acknowledges that Landlord has made no representation or warranty (express or implied) regarding (i) the condition of the Premises or the Real Property except as specifically set forth in this Lease and the Tenant Work Letter or (ii) the suitability or fitness of the Premises or the Real Property for the conduct of Tenant's business. Any existing leasehold improvements in the Premises as of the date of this Lease, together with the Improvements to be constructed pursuant to the Tenant Work Letter, may be collectively referred to herein as the "**Tenant Improvements**." Notwithstanding the foregoing, Landlord shall deliver the Premises to Tenant with the HVAC, electrical and plumbing systems and Building structure in good working order. If, upon Landlord's delivery of the Premises to Tenant, such systems or structure are not in good working order and Tenant notifies Landlord on or before the earlier of (i) that date which is thirty (30) days after Landlord's delivery of the Premises to Tenant or (ii) the date Tenant first commences physical construction of its Improvements

(including demolition work), then Landlord shall, at Landlord's sole cost and expense and as Tenant's sole remedy therefor, put such systems or structure in good working order.

1.3 <u>Verification of Rentable Square Feet of Premises and Building</u>. For purposes of this Lease, "rentable square feet" shall mean "rentable area" calculated pursuant to Landlord's standard method of measurement for the Building. The parties hereby stipulate to the square footage listed above and agree that the Premises will not be remeasured during the initial Lease Term unless the physical size of the Premises changes. In the event that after the initial Lease Term, Landlord's space measurement consultant determines that the space measurement shall be different from that set forth in this Lease, Landlord shall modify all amounts, percentages and figures appearing or referred to in this Lease to conform to such corrected rentable square footage (including, without limitation, the amount of the "Rent," as that term is defined in Article 4 of this Lease). If such modification is made, it will be confirmed in writing by Landlord to Tenant.

1.4 <u>Right of First Refusal</u>. Landlord hereby grants to the original Tenant named in this Lease (the "**Original Tenant**"), during the initial Lease Term only, a right of first refusal with respect to any space located on the fifth (5th) floor of the Building (the "**First Refusal Space**"). Notwithstanding the foregoing, (i) such first refusal right shall commence only following the expiration or earlier termination of any existing lease pertaining to the First Refusal Space (collectively, the "**Superior Leases**"), including any renewal or extension of such existing or future lease pursuant to an option granted in such existing lease, and (ii) such first refusal right shall be subordinate and secondary to all rights of expansion, first refusal, first offer or similar rights granted to the tenant(s) of the Superior Leases or any other leases in existence as of the date of this Lease (the rights described in items (i) and (ii), above to be known collectively, for purposes of this Section 1.4 only, as "**Superior Rights**"). Tenant's right of first refusal shall be on the terms and conditions set forth in this Section 1.4.

1.4.1 <u>Procedure</u>. Landlord shall notify Tenant (the "**First Refusal Notice**") from time to time when Landlord receives a proposal or prepares a response to a request for proposal that Landlord would consider for all or any portion of the First Refusal Space, where no holder of a Superior Right desires to lease such space. The First Refusal Notice shall describe the space which is the subject of the proposal (which may include space outside of the First Refusal Space) and shall set forth the material economic terms and conditions (including the proposed lease term) as proposed by Landlord and as set forth in the proposal (collectively, the "**Terms**"); provided that if there are at least five (5) years remaining in the initial Lease Term measured from the commencement date of Tenant's lease of the applicable First Refusal Space, then the Lease Term for the First Refusal Space will be co-terminus with the Lease Term for the initial Premises. Notwithstanding the foregoing, Landlord's obligation to deliver the First Refusal Notice shall not apply during the last ten (10) months of the initial Lease Term unless Tenant has delivered an Interest Notice pursuant to Section 2.2.2 of this Lease, nor during the period following Landlord's delivery of the Option Rent Notice to Tenant pursuant to Section 2.2.2 unless and until Tenant has delivered to Landlord the Option Notice pursuant to Section 2.2 of this Lease is fully executed, regardless of whether the lease based on such proposal or response to request for proposal is entered after the date hereof.

1.4.2 Procedure for Acceptance. If Tenant wishes to exercise Tenant's right of first refusal with respect to the space described in the First Refusal Notice, then within seven (7) business days after delivery of the First Refusal Notice to Tenant (the "Election Date"), Tenant shall deliver written notice to Landlord ("Tenant's Election Notice") pursuant to which Tenant shall elect either to (i) lease the entire space described in the First Refusal Notice upon the Terms set forth in the First Refusal Notice to the proposed third party tenant who was the subject of the initial proposal or (ii) refuse to lease such space identified in the First Refusal Notice, in which event Landlord may lease such space to the third party who was the subject of the proposal and/or request for proposal; provided that in the event the Terms are altered so as to reduce the Net Effective Rental Rate (as defined below) by more than five percent (5%) of the Net Effective Rental Rate listed in the Terms offered to Tenant, Landlord will again be obligated to offer the First Refusal Space to Tenant on such revised Terms, in which case Tenant will have three (3) business days to deliver Tenant's Election Notice as set forth above. The term "Net Effective Rental Rate" shall mean the rental rate, as adjusted to reflect the value of any free rent, tenant improvement allowance or similar monetary concessions contained in the First Refusal Notice. Further, in the event Landlord does not enter into a lease for the First Refusal Space with the third party who was the subject of the prior proposal, Landlord will be obligated to deliver a new First Refusal Notice to Tenant upon receipt of any proposal or response to proposal with a new prospective tenant, and in such instance such revised First Refusal Notice will be subject to the terms of this Section 1.4 to the same extent as the initial First Refusal Notice. Notwithstanding anything contained herein, Landlord will only be obligated to deliver First Refusal Notices to Tenant with respect to three (3) separate third party offers (i.e., any second notice provided with respect to the same third party offer will not count toward the foregoing three (3) instances), and thereafter this right of first refusal will terminate and be of no further force or effect. If Tenant does not so respond in writing to Landlord's First Refusal Notice by the Election Date, Tenant shall be deemed to have elected the option described in clause (ii) above. Notwithstanding anything herein to the contrary. Tenant may only exercise its right of first refusal with respect to all of the space described in the First Refusal Notice, and not a portion thereof.

1.4.3 <u>Lease of First Refusal Space</u>. If Tenant timely exercises Tenant's right to lease the First Refusal Space as set forth herein, Landlord and Tenant shall execute an amendment to this Lease incorporating into this Lease the Terms applicable to such First Refusal Space.

1.4.4 <u>Termination of Right of First Refusal</u>. The right of first refusal granted herein shall terminate as to a particular First Refusal Space upon the failure by Tenant to exercise its right of first refusal with respect to such First Refusal Space as offered by Landlord (subject to Landlord obligation to submit a repeat First Refusal Notice for any new third party as provided in Section 1.4.2 above) but shall remain in effect for any subsequent availability of all or any portion of the remaining First Refusal Space. Landlord shall not have any obligation to deliver the First Refusal Notice if, as of the date Landlord would otherwise deliver the First Refusal Notice to Tenant, Tenant is in default under the Lease after any applicable notice and cure periods, Tenant does not physically occupy the entire Premises, if any portion of the Premises is subject to a sublease, or if the Lease has been assigned (other than an assignment or sublease to an Affiliate of Tenant or a sublease of not more than twenty five percent (25%) of the Premises to a subcontractor of Tenant as part of Tenant's ordinary course of business). In addition, at Landlord's option, if Tenant has previously

delivered Tenant's Election Notice in accordance with Section 1.4.2 and, as of the scheduled date of delivery of such First Refusal Space to Tenant, Tenant is in default under the Lease after any applicable notice and cure periods, Tenant does not physically occupy the entire Premises, if any portion of the Premises is subject to a sublease, or if the Lease has been assigned (other than an assignment or sublease to an Affiliate of Tenant or a sublease of not more than twenty five percent (25%) of the Premises to a subcontractor of Tenant as part of Tenant's ordinary course of business), Tenant shall not have the right to lease the First Refusal Space.

2. LEASE TERM

2.1 Initial Term. The terms and provisions of this Lease shall be effective as of the date of this Lease except for the provisions of this Lease relating to the payment of Rent. The term of this Lease (the "Lease Term") shall be for the period of time set forth in Section 7.1 of the Summary and shall commence on the date (the "Lease Commencement Date") set forth in Section 7.2 of the Summary (subject, however, to the terms of the Tenant Work Letter, if applicable), and shall terminate upon the expiration of the Lease Term, unless this Lease is sooner terminated as hereinafter provided. For purposes of this Lease, the term "Lease Year" shall mean each consecutive twelve (12) month period during the Lease Term; provided, however, that if the Lease Commencement Date is not the first day of the month, then the first Lease Year shall commence on the Lease Commencement Date and end on the last day of the twelfth month thereafter and the second and each succeeding Lease Year shall commence on the first day of the next calendar month; and further provided that the last Lease Year shall end on the last day of the Lease Term (for example, if the Lease Commencement Date is April 15, the first Lease Year will be April 15 through April 30 of the following year, and each succeeding Lease Year will be May 1 through April 30). If Landlord is unable to deliver possession of the Premises to Tenant on or before the anticipated Lease Commencement Date as set forth in Section 7.2 of the Summary, Landlord shall not be subject to any liability for its failure to do so and such failure shall not affect the validity of this Lease nor the obligations of Tenant hereunder (unless the cause of Landlord's delay in delivering possession of the Premises is the result of Landlord's negligence or misconduct). At any time during the Lease Term, Landlord may deliver to Tenant a notice of Lease Term dates substantially in the form of Exhibit C attached hereto, which notice Tenant shall execute and return to Landlord within ten (10) days of receipt thereof; if Tenant fails to execute and return such notice within such time period, the information contained in such notice shall be deemed correct and binding upon Tenant. Landlord will not be obligated to deliver possession of the Premises to Tenant until Landlord has received from Tenant all of the following: (i) a copy of this Lease fully executed by Tenant, (ii) any Security deposit, Guaranty and/or Letter of Credit required hereunder and the first installment of Monthly Base Rent; and (iii) copies of Tenant's certificates of insurance as required hereunder. Landlord shall grant Tenant early access to the Premises for the purpose of installing fixtures, furniture, and equipment for up to thirty (30) days prior to the Lease Commencement Date. Prior to Tenant's entry into the Premises as permitted by the terms of this Section 2.1, Tenant shall submit a schedule to Landlord for approval, which schedule shall detail the timing and purpose of Tenant's entry. Tenant shall hold Landlord harmless from and indemnify, protect and defend Landlord against any loss or damage to the Project or Premises and against injury to any persons caused by Tenant's actions pursuant to this Section 2.1.

2.2 <u>Option Term</u>. Landlord hereby grants to Tenant, one (1) option to extend the Lease Term for a period of five (5) years ("**Option Term**"), which option shall be exercisable only by written notice ("**Option Notice**") delivered by Tenant to Landlord as provided in Section 2.2.2 below. Tenant shall not have the rights contained in this Section 2.2 if, as of the date of the Option Notice, Tenant is in default under this Lease after any applicable notice and cure period, Tenant or an Affiliate Assignee does not physically occupy at least 75% of the Premises, more than 25% of the Premises is subject to a sublease (other than to an Affiliate), this Lease has been assigned (other than to an Affiliate), or any portion of the Premises has been recaptured pursuant to Section 14.4 below).

2.2.1 <u>Option Rent</u>. The Rent payable by Tenant during the Option Term (the "**Option Rent**") shall be equal to the fair market rent for the Premises and the parking passes as of the commencement date of the Option Term. The fair market rent shall be the rental rate, including all escalations, at which tenants, as of the commencement of the Option Term, are leasing non-sublease, non-encumbered space (other than loans on the applicable building) comparable in size, location and quality to the Premises for a term of five (5) years, under new leases or renewal amendments which comparable space is located in comparable office buildings in the I-15 corridor of San Diego County.

2.2.2 Exercise of Option. The option contained in this Section 2.2 shall be exercised by Tenant, if at all, only in the following manner: (i) Tenant shall deliver written notice ("Interest Notice") to Landlord no sooner than twelve (12) months and no later than ten (10) months prior to the expiration of the then current Lease Term, stating that Tenant is interested in exercising its option; (ii) Landlord, after receipt of the Interest Notice, shall deliver notice (the "Option Rent Notice") to Tenant setting forth Landlord's determination of the Option Rent; and (iii) if Tenant wishes to exercise such option, Tenant shall, within thirty (30) days after Tenant's receipt of the Option Rent Notice, exercise the option by delivering the Option Notice to Landlord and upon, and concurrent with such exercise, Tenant may, at its option, object to the Option Rent Notice, then the Option Rent shall be determined as set forth in Section 2.2.3 below. Failure of Tenant to deliver the Interest Notice to Landlord on or before the date specified in (ii) above or to deliver the Option Notice to Landlord on or before the date specified in (ii) above shall be determed to constitute Tenant's failure to exercise its option to extend. If Tenant timely and properly exercises its option to extend, the Lease Term, subject to Section 2.2.3 below, shall be extended for the Option Term upon all of the terms and conditions set forth in this Lease, except that the Rent shall be as indicated in the Option Rent Notice or as determined in accordance with Section 2.2.3, as applicable, and all references herein to the Lease Term shall include the Option Term.

2.2.3 <u>Determination of Option Rent</u>. In the event Tenant exercises its option to extend but objects to Landlord's determination of the Option Rent concurrently with its exercise of the option to extend, Landlord and Tenant shall attempt to agree in good faith upon the Option Rent. If Landlord and Tenant fail to reach agreement within twenty (20) days following Tenant's delivery of the Option Notice (the "**Outside Agreement Date**"), then each party shall make a separate determination of the Option Rent within five (5) business days after the Outside Agreement Date

and will concurrently exchange such determinations. Such determinations shall then be submitted to arbitration in accordance with Sections 2.2.3.1 through 2.2.3.7 below.

2.2.3.1 Landlord and Tenant shall each appoint one arbitrator who shall by profession be a real estate broker who shall have been active over a five (5) year period ending on the date of such appointment in the leasing of comparable properties on the I-15 Corridor of San Diego County. The determination of the arbitrators shall be limited solely to the issue of whether Landlord's or Tenant's submitted Option Rent is the closest to the actual fair market rent, as determined by the arbitrators, taking into account the requirements of Section 2.2.1 of this Lease. Each such arbitrator shall be appointed within fifteen (15) business days after the applicable Outside Agreement Date.

2.2.3.2 The two (2) arbitrators so appointed shall within five (5) days of the date of the appointment of the last appointed arbitrator agree upon and appoint a third arbitrator who shall be qualified under the same criteria set forth hereinabove for qualification of the initial two (2) arbitrators.

2.2.3.3 The three (3) arbitrators shall within five (5) days of the appointment of the third arbitrator reach a decision as to whether the parties shall use Landlord's or Tenant's submitted Option Rent and shall notify Landlord and Tenant thereof.

2.2.3.4 The decision of the majority of the three (3) arbitrators shall be binding upon Landlord and Tenant.

2.2.3.5 If either Landlord or Tenant fails to appoint an arbitrator within fifteen (15) business days after the applicable Outside Agreement Date, the arbitrator appointed by one of them shall reach a decision, notify Landlord and Tenant thereof, and such arbitrator's decision shall be binding upon Landlord and Tenant.

2.2.3.6 If the two (2) arbitrators fail to agree upon and appoint a third arbitrator, or both parties fail to appoint an arbitrator, then the appointment of the third arbitrator or any arbitrator shall be dismissed and the Option Rent to be decided shall be forthwith submitted to arbitration under the provisions of the American Arbitration Association, but subject to the instruction set forth in this Section 2.2.3.

2.2.3.7 The cost of arbitration shall be paid by Landlord and Tenant equally.

2.3 <u>Beneficial Occupancy</u>. Notwithstanding anything to the contrary contained herein, Tenant shall have the right to commence business from the Premises during the period (the "**Beneficial Occupancy Period**") from September 15, 2018 until the Commencement Date, provided that (i) Tenant shall give Landlord at least ten (10) days prior notice of any such occupancy of the Premises, (ii) a certificate of occupancy (or its equivalent) shall have been issued by the appropriate governmental authorities for the Premises, and (iii) all of the terms and conditions of this Lease shall apply during the Beneficial Occupancy Period (if any), except that Tenant's obligation to pay monthly Base Rent shall not apply during the Beneficial Occupancy Period.

3. BASE RENT

Tenant shall pay, without notice or demand, to Landlord or Landlord's agent at the management office of the Building, or at such other place as Landlord may from time to time designate in writing, monthly installments of base rent ("**Base Rent**") as set forth in Section 8 of the Summary, in advance on or before the first day of each and every month during the Lease Term, without any setoff or deduction whatsoever, except as expressly set forth in this Lease (i.e., Section 6.10 below). The Base Rent for the first full month of the Lease Term (or if the first full month of the Lease Term is within a free rent period, then the Base Rent for the first full month which occurs after the expiration of any free rent period) shall be paid at the time of Tenant's execution of this Lease. If any rental or other payment date (including the Lease Commencement Date) falls on a day of the month other than the first day of such month or if any rental or other payment is for a period which is shorter than one month, then the rental or other payment for any such fractional month shall be a proportionate amount of a full calendar month's rental or other payment based on the proportion that the number of days in such fractional month bears to the number of days in the calendar month during which such fractional month occurs.

4. ADDITIONAL RENT

4.1 Additional Rent. In addition to paying the Base Rent specified in Article 3 of this Lease, and separately metered electricity charges pursuant to Section 6.1.2 below, Tenant shall pay as additional rent Tenant's Share of the annual Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses that are in excess of the amount of Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses, respectively applicable to the Base Year (the "Excess"). Such additional rent, together with any and all other amounts payable by Tenant to Landlord pursuant to the terms of this Lease, shall be hereinafter collectively referred to as the "Additional Rent." The Base Rent and Additional Rent are herein collectively referred to as the "Rent." All amounts due under this Article 4 as Additional Rent shall be payable for the same periods and in the same manner, time and place as the Base Rent. Without limitation on other obligations of Tenant which shall survive the expiration of the Lease Term, the obligations of Tenant to pay the Additional Rent provided for in this Article 4 shall survive the expiration of the Lease Term.

4.2 <u>Definitions</u>. As used in this Article 4, the following terms shall have the meanings hereinafter set forth:

4.2.1 "Base Year" shall mean the year set forth in Section 9.1 of the Summary.

4.2.2 "Calendar Year" shall mean each calendar year in which any portion of the Lease Term falls, through and including the calendar year in which the Lease Term expires.

4.2.3 **"Expense Year**" shall mean each Calendar Year, provided that Landlord, upon notice to Tenant, may change the Expense Year from time to time to any other twelve (12) consecutive-month period, and, in the event of any such change, Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses shall be equitably adjusted for any Expense Year involved in any such change.

4.2.4 "**Insurance Expenses**" shall mean the cost of insurance carried by Landlord, in such amounts as Landlord may reasonably determine or as may be required by any mortgagees or the lessor of any underlying or ground lease affecting the Real Property, including any deductibles thereunder.

425 "Operating Expenses" shall mean all expenses, costs and amounts which Landlord incurs or which accrue during any Expense Year because of or in connection with the ownership, management, maintenance, repair, restoration or operation of the Real Property (other than Insurance Expenses, Tax Expenses and Utility Expenses) and the cost of any capital improvements or other costs which (A) are intended to reduce Operating Expenses, Insurance Expenses, Utility Expenses or Tax Expenses, (B) are required under any and all Applicable Laws or (C) are reasonably necessary for the health and safety of the occupants of the Real Property; provided, however, that if any such cost described in (A), (B) or (C) above is a capital expenditure, such cost shall be amortized over its useful life as Landlord shall reasonably determine in accordance with generally accepted accounting principles, together with interest on the unamortized balance equal to the Prime Rate plus two percent (2%); provided, however, that cost savings resulting from capital improvements or capital assets may be included in Operating Expenses without regard to the foregoing amortization requirement. If the Building is less than ninety-five percent (95%) occupied during any portion of the Base Year or any Expense Year, Landlord shall make an appropriate adjustment to the variable components of Operating Expenses or Utility Expenses (as defined below) for such year, employing sound accounting and management principles, to determine the amount of Operating Expenses or Utility Expenses that would have been paid had the Building been ninety-five percent (95%) occupied. In no event will Operating Expenses for purposes of calculating Tenant's Share pursuant to this Lease include more than one hundred percent (100%) of the actual Operating Expenses accrued in any Expense Year.

Notwithstanding anything in the Lease to the contrary, Operating Expenses shall not include:

(1) the cost of providing any service directly to and paid directly by any tenant (outside of such tenant's Operating Expenses payments);

(2) the cost of any items for which Landlord is reimbursed by insurance proceeds, condemnation awards, a tenant of the Building, or otherwise to the extent so reimbursed;

(3) any real estate brokerage commissions or other costs incurred in procuring tenants, or any fee in lieu of commissions;

(4) ground lease payments (if any);

(5) costs incurred by Landlord due to the violation by Landlord or any tenant of the terms and conditions of any lease of space in the Building that would not have been incurred but for such violation;

(6) Landlord's general corporate overhead (as opposed to overhead expenses related to the Building or Real Property) and costs for the operation of the entity which constitutes Landlord as the same are distinguished from the costs of the operation of the Real Property, such as accounting

and legal matters unrelated to the Building or Real Property, costs of defending any lawsuits with any mortgagee, costs of selling or mortgaging Landlord's interest in the Real Property, costs of any disputes between Landlord and its employees (if any) who do not engage in work relating to the Building or Real Property, lawsuits between Landlord and its property management provider;

(7) any compensation paid to clerks, attendants or other persons in commercial concessions operated by Landlord (other than in the parking facility for the Building);

(8) bad debt expenses and interest, principal, points and fees on debts (except in connection with the financing of items which may be included in Operating Expenses) or amortization on any ground lease, mortgage or mortgages or any other debt instrument encumbering the Building (including the Real Property on which the Building is situated);

(9) marketing costs, including leasing commissions and attorneys' fees in connection with the negotiation and preparation of letters, deal memos, letters of intent, leases, subleases and/or assignments, space planning costs, and other costs and expenses incurred in connection with lease, sublease and/or assignment negotiations and transactions with present or prospective tenants or other occupants of the Building;

(10) costs, including permit, license and inspection costs, including any landlord supervision fees, incurred with respect to the installation of other tenants' or occupants' improvements made for tenants or other occupants in the Building or incurred in renovating or otherwise improving, decorating, painting or redecorating vacant space for tenants or other occupants in the Building;

(11) any costs expressly excluded from Operating Expenses elsewhere in this Lease;

(12) costs of any items (including, but not limited to, costs incurred by Landlord for the repair of damage to the Building) to the extent Landlord receives reimbursement from insurance proceeds or from a third party (except that any deductible amount under any insurance policy shall be included within Operating Expenses);

(13) expenses in connection with services or other benefits which are not offered to Tenant or for which Tenant is charged for directly but which are provided to another tenant or occupant of the Building, without charge;

(14) electric power costs or other utility costs for which any tenant directly contracts with the local public service company (but Landlord shall have the right to "gross up" as if such space was vacant);

(15) costs (including in connection therewith all attorneys' fees and costs of settlement, judgments and/or payments in lieu thereof) arising from claims, disputes or potential disputes in connection with potential or actual claims, litigation or arbitrations pertaining to another tenant or occupant of the Building;

(16) costs incurred in connection with the original construction of the Building;

(17) costs incurred to (i) comply with laws relating to the removal of any "Hazardous Material," as that term is defined in Article 29 of this Lease, which was in existence on the Real Property prior to the Commencement Date, and was of such a nature that a federal, state or municipal governmental authority, if it had then had knowledge of the presence of such Hazardous Material, in the state, and under the conditions that it then existed on the Real Property, would have then required the removal of such Hazardous Material or other remedial or containment action with respect thereto, and (ii) remove, remedy, contain, or treat any Hazardous Material, which Hazardous Material is brought onto the Real Property after the date hereof by Landlord or any other tenant of the Project and is of such a nature, at that time, that a federal, state or municipal governmental authority, if it had then had knowledge of the presence of such Hazardous Material, in the state, and under the conditions, that it then exists on the Real Property after the conditions, that it then exists on the Real Property, would have then required the removal of such Hazardous Material, in the state, and under the conditions, that it then exists on the Real Property, would have then required the removal of such Hazardous Material or other remedial or containment action with respect thereto; provided, however, Operating Expenses shall include costs incurred in connection with the clean-up, remediation, monitoring, management and administration of (and defense of claims related to) the presence of Hazardous Materials used by Landlord (provided such use is not negligent and is in compliance with Applicable Laws) in connection with the operation, repair and maintenance of the Real Property to perform Landlord's obligations under this Lease (such as, without limitation, fuel oil for generators, cleaning solvents, and lubricants) and which are customarily found or used in comparable buildings;

(18) wages, salaries, including employment taxes and benefits payable to employees above the level of Property Manager;

(19) any management fee in excess of three percent (3%) of gross receipts for the Real Property;

(20) costs and expenses (which would otherwise be includable in Operating Expenses) to the extent the same arise from the gross negligence or tortious acts of Landlord or any of Landlord's agents, employees or contractors;

(21) any overhead and/or profit increment paid to Landlord or to subsidiaries or affiliates or Landlord for services in the Real Property to the extent the same exceed the amount which would generally be expected to be the cost of such services rendered by comparably qualified unaffiliated third parties;

(22) rental associated with the Real Property management office to the extent such costs exceed fair market rent;

(23) costs associated with Landlord's political or charitable contributions;

(24) the cost to purchase art work (provided that costs to maintain art work may be included in Operating Expenses);

(25) fines, interest or penalties due to Landlord's failure to pay any Operating Expenses or Real Estate Taxes when due; and

(26) capital expenditures except to the extent expressly included herein.

4.2.6 "Systems and Equipment" shall mean any plant, machinery, transformers, duct work, cable, wires, and other equipment, facilities, and systems designed to supply heat, ventilation, air conditioning and humidity or any other services or utilities, or comprising or serving as any component or portion of the electrical, gas, steam, plumbing, sprinkler, communications, alarm, security, or fire/life safety systems or equipment, or any other mechanical, electrical, electronic, computer or other systems or equipment which serve the Real Property in whole or in part.

4.2.7 "**Tax Expenses**" shall mean all federal, state, county, or local governmental or municipal taxes, fees, charges or other impositions of every kind and nature, whether general, special, ordinary or extraordinary, (including, without limitation, real estate taxes, general and special assessments, transit taxes, leasehold taxes or taxes based upon the receipt of rent, including gross receipts or sales taxes applicable to the receipt of rent, unless required to be paid by Tenant, personal property taxes imposed upon the fixtures, machinery, equipment, apparatus, systems and equipment, appurtenances, furniture and other personal property used in connection with the Real Property), which Landlord shall pay during any Expense Year because of or in connection with the ownership, leasing and operation of the Real Property or Landlord's interest therein.

4.2.8 "Utility Expenses" shall mean the cost of supplying all utilities to the Real Property (other than utilities for which tenants of the Building are separately metered), including utilities for the heating, ventilation and air conditioning system for the Building and Common Areas.

4.2.9 **"Tenant's Share**" shall mean the percentage set forth in Section 9.2 of the Summary. Tenant's Share was calculated by multiplying the number of rentable square feet of the Premises by 100 and dividing the product by the total rentable square feet in the Building. The total rentable square feet in the Building as of the date hereof is 153,550 (subject to adjustment pursuant to Section 1.3 above).

4.3 Payment of Additional Rent.

4.3.1 <u>Allocation of Expenses</u>. The Building is part of a multi-building project, and Landlord will allocate Tax Expenses, Insurance Expenses, Operating Expenses and Utility Expenses which are attributable to the Real Property as a whole among the Building and the Other Building on an equitable basis.

4.3.2 <u>Statement of Actual Expenses and Payment by Tenant</u>. Landlord shall give to Tenant following the end of each Expense Year, a statement (the "**Statement**") which shall state the Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses incurred or accrued for such preceding Expense Year, and which shall indicate the amount, if any, of any Excess. Upon receipt of the Statement for each Expense Year commencing or ending during the Lease Term, if an Excess is present, Tenant shall pay, with its next installment of Base Rent due (or within thirty (30) days of receipt if the Lease Term has expired prior to Tenant's receipt of the Statement), the full amount of the Excess for such Expense Year, less the amounts, if any, paid during such Expense Year as Estimated Excess. The failure of Landlord to timely furnish the Statement or the Estimated Statement for any Expense Year shall not prejudice Landlord from enforcing its rights under this Article 4. The provisions of this Section 4.3.2 shall survive the expiration or earlier termination of

the Lease Term. In the event Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses are lower than Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses, respectively, for the Base Year, Tenant will not be entitled to any credit against Base Rent, but Tenant's Share shall be zero as to the applicable category of expenses.

4.3.3 <u>Statement of Estimated Expenses</u>. In addition, Landlord shall endeavor to give Tenant a yearly expense estimate statement (the "**Estimate Statement**") which shall set forth Landlord's reasonable estimate (the "**Estimate**") of what the total amount of Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses for the then-current Expense Year shall be and the estimated Excess (the "**Estimated Excess**"). If pursuant to the Estimate Statement an Estimated Excess is calculated for the then-current Expense Year, Tenant shall pay, with its next installment of Base Rent due, a fraction of the Estimated Excess for the then-current Expense Year (reduced by any amounts paid pursuant to the last sentence of this Section 4.3.3). Such fraction shall have as its numerator the number of months which have elapsed in such current Expense Year to the month of such payment, both months inclusive, and shall have twelve (12) as its denominator (for example, if Landlord issues an Estimate Statement in April, Tenant will pay 4/12 of the Estimated Excess as the initial payment of Estimated Excess which is owed by Tenant pursuant to such Estimate Statement for the applicable Expense Year). Until a new Estimate Statement is furnished, Tenant shall pay monthly, with the monthly Base Rent installments, an amount equal to one-twelfth (1/12) of the total Estimated Excess for an Expense Year, then the difference shall be credited against future expenses charged to Tenant or refunded to Tenant if the Lease has expired.

4.3.4 Audit Right. Within one hundred twenty (120) days after receipt of a Statement by Tenant ("Review Period"), if Tenant disputes the amount set forth in the Statement, Tenant's employees or an independent certified public accountant (which accountant is a member of a nationally or regionally recognized accounting firm and is not paid on a contingency basis), designated by Tenant, may, after reasonable notice to Landlord and at reasonable times, inspect Landlord's records (pertaining to Landlord's calculation of Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses) at Landlord's offices, provided that Tenant is not then in default after expiration of all applicable cure periods and provided further that Tenant and such accountant or representative shall, and each of them shall cause their respective agents and employees to, maintain all information contained in Landlord's records in strict confidence. Notwithstanding the foregoing. Tenant shall only have the right to review Landlord's records one (1) time during any twelve (12) month period. Tenant's failure to dispute the amounts set forth in any Statement within the Review Period shall be deemed to be Tenant's approval of such Statement and Tenant, thereafter, waives the right or ability to dispute the amounts set forth in such Statement. If after such inspection, but within thirty (30) days after the Review Period, Tenant notifies Landlord in writing that Tenant still disputes such amounts, a certification as to the proper amount shall be made, at Tenant's expense, by an independent certified public accountant selected by Landlord and reasonably approved by Tenant and who is a member of a nationally or regionally recognized accounting firm. Landlord shall cooperate in good faith with Tenant and the accountant to provide Tenant and the accountant with the information upon which the certification is to be based. However, if such certification by the accountant proves that the total amount of Operating Expenses, Insurance Expenses, Utility

Expenses and Tax Expenses set forth in the Statement were overstated by more than five percent (5%), then the actual, documented cost of the accountant, as well as Tenant's reasonable out of pocket cost incurred for its audit shall be paid for by Landlord. Promptly following the parties receipt of such certification, the parties shall make such appropriate payments or reimbursements, as the case may be, to each other, as are determined to be owing pursuant to such certification. In no event shall Landlord or its property manager be required to (i) photocopy any accounting records or other items or contracts, (ii) create any ledgers or schedules not already in existence, (iii) incur any costs or expenses relative to such inspection, or (iv) perform any other tasks other than making available such accounting records as are described in this paragraph. Landlord shall not be liable for the payment of any contingency fee payments to any auditor or consultant of Tenant. The provisions of this Section shall be the sole method to be used by Tenant to dispute the amount of Operating Expenses, Insurance Expenses, Utility Expenses and Tax Expenses payable by Tenant under this Lease, and Tenant waives any other rights or remedies relating thereto.

4.4 <u>Taxes and Other Charges for Which Tenant Is Directly Responsible</u>. Tenant shall reimburse Landlord upon demand for any and all taxes or assessments required to be paid by Landlord (except to the extent included in Tax Expenses by Landlord), excluding state, local and federal personal or corporate income taxes measured by the net income of Landlord from all sources and estate and inheritance taxes, whether or not now customary or within the contemplation of the parties hereto, when: (i) said taxes are measured by or reasonably attributable to the cost or value of Tenant's equipment, furniture, fixtures and other personal property located in the Premises, or by the cost or value of any leasehold improvements made in or to the Premises by or for Tenant, to the extent the cost or value of such leasehold improvements exceeds the cost or value of a building standard build-out as determined by Landlord regardless of whether title to such improvements shall be vested in Tenant or Landlord; (ii) said taxes are assessed upon or with respect to the possession, leasing, operation, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises or any portion of the Real Property (including the Parking Facilities); or (iii) said taxes are assessed upon this transaction or any document to which Tenant is a party creating or transferring an interest or an estate in the Premises.

5. USE OF PREMISES

5.1 <u>Permitted Use</u>. Tenant shall use the Premises solely for general office purposes and ancillary uses related thereto which have been reasonably approved by Landlord, but in all cases consistent with the character of the Building as a first-class office building and in compliance with all applicable laws, codes (including zoning codes) and ordinances, and Tenant shall not use or permit the Premises to be used for any other purpose or purposes whatsoever. Tenant will procure and maintain, at its sole cost and expense, any and all permits or approvals required for Tenant's use and occupancy of the Premises.

5.2 <u>Prohibited Uses</u>. Tenant further covenants and agrees that it shall not use, or suffer or permit any person or persons to use, the Premises, the Parking Facilities or any other Common Areas or any part of the Real Property for any use or purpose contrary to the rules and regulations

reasonably established by Landlord for the Real Property (the "**Rules and Regulations**"), attached hereto as <u>Exhibit D</u> and made a part hereof, or in violation of any federal, state or local laws, codes or ordinances or any recorded covenants, conditions and restrictions or ground or underlying leases affecting the Real Property or Landlord's sustainability programs or any Environmental Laws ("**Applicable Laws**"); provided that Tenant will be permitted to have *de minimis* quantities of household cleaning products and office supplies used in the ordinary course of Tenant's business in the Premises and kept in compliance with Applicable Laws. Tenant shall not use or allow Tenant Parties to use any part of the Premises or the Real Property for the storage, use, treatment, manufacture or sale of any hazardous or toxic material, and Tenant will be responsible for any damage caused by unauthorized hazardous or toxic material in the Premises or Real Property in violation of this sentence. Tenant will not do or permit to be done anything that will obstruct or interfere with the rights of other tenants or occupants of the Building.

6. SERVICES AND UTILITIES

6.1 <u>Standard Tenant Services</u>. Tenant will at all times operate within the Premises in a manner which is consistent with responsible energy and water use and which minimizes the energy and water use in the Premises, including use of proven energy and carbon reducing measures, such as use of energy efficient light bulbs, use of lighting controls (e.g., automatic shut offs or dimmers) and purchasing ENERGY STAR® qualified equipment and products certified by the EPA's Water Sense® program. Landlord shall provide the following services on all days during the Lease Term, unless otherwise stated below.

6.1.1 Subject to all governmental rules, regulations and guidelines applicable thereto, Landlord shall provide heating, ventilation and air conditioning ("**HVAC**") when necessary for normal comfort for normal office use in the Premises, from Monday through Friday, during the period from 7:00 a.m. to 6:00 p.m., and on Saturday during the period from 8:00 a.m. to 1:00 p.m., except for the date of observation of New Year's Day, Presidents' Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day and other locally or nationally recognized holidays (collectively, the "**Holidays**").

6.1.2 Landlord shall provide adequate electrical wiring and facilities for normal general office use, as determined by Landlord. Tenant will pay for all electricity supplied to the Premises on a net basis as Additional Rent. The Premises is separately metered and Tenant shall make payment directly to the entity providing such electricity.

6.1.3 Landlord shall provide city water from the regular Building outlets for drinking, lavatory and toilet purposes.

6.1.4 Landlord shall provide janitorial services five (5) days per week, except the date of observation of the Holidays, in and about the Premises and window washing services in a manner consistent with other comparable buildings in the vicinity of the Building. Landlord agrees that it will require that any janitorial contractors providing services to the Premises only use employees who are legal United States residents to service the Premises. Notwithstanding the foregoing, at any time during the Lease Term, as it may be extended, Tenant may elect to provide its own janitorial services by providing written notice thereof to Landlord, in which event Landlord

will have the right to reasonably approve Tenant's janitorial contractor, Tenant shall be obligated to keep and maintain the Premises in a clean and sanitary condition and will pay for such services directly to the provider, and the costs of janitorial services shall be removed from Operating Expenses (both in the Base Year and all subsequent Expense Years) owed by Tenant.

6.1.5 Landlord shall provide nonexclusive automatic passenger elevator service at all times.

6.1.6 Landlord shall provide nonexclusive freight elevator service subject to scheduling by Landlord.

6.1.7 Landlord shall provide for the replacement of all Building standard fluorescent bulbs and ballasts and Building standard incandescent bulbs in the Premises and Common Areas in a manner and to the extent deemed by Landlord to be reasonably required.

Overstandard Tenant Use. Tenant shall not, without Landlord's prior written consent, use heat-generating machines, 6.2 machines other than normal fractional horsepower office machines, or equipment or lighting other than building standard lights in the Premises, which may affect the temperature otherwise maintained by the air conditioning system or increase the electricity or water normally furnished for the Premises by Landlord pursuant to the terms of Section 6.1 of this Lease. If Tenant uses water or heat or air conditioning in excess of that supplied by Landlord pursuant to Section 6.1 of this Lease, Tenant shall pay to Landlord, upon billing, the cost of such excess consumption, the cost of the installation, operation, and maintenance of equipment which is installed in order to supply such excess consumption, and the cost of the increased wear and tear on existing equipment caused by such excess consumption, and Landlord may install devices to separately meter any increased use and in such event Tenant shall pay the increased cost directly to Landlord, on demand, including the cost of such additional metering devices. If Tenant desires to use HVAC during hours other than those for which Landlord is obligated to supply such utilities pursuant to the terms of Section 6.1 of this Lease, Tenant shall give Landlord such prior notice, as Landlord shall from time to time establish as appropriate, of Tenant's desired use and Landlord shall supply such utilities to Tenant at such hourly cost to Tenant as Landlord shall from time to time establish, which rates are Thirty Five Dollars (\$35) per hour per zone as of the date hereof. Amounts payable by Tenant to Landlord for such use of additional utilities shall be deemed Additional Rent hereunder. Notwithstanding anything herein to the contrary, any HVAC or other service necessary to accommodate a computer server room will be deemed to constitute an overstandard use and will be subject to the provisions of this Section 6.2.

6.3 <u>Interruption of Use</u>. Except as provided in Section 6.10 below, Tenant agrees that Landlord shall not be liable for any damages incurred by Tenant, by abatement of Rent or otherwise, for failure to furnish or delay in furnishing any utility or service (including telephone and telecommunication services), or for any diminution in the quality or quantity thereof; and such failures or delays or diminution shall never be deemed to constitute an eviction or disturbance of Tenant's use and possession of the Premises or relieve Tenant from paying Rent or performing any of its obligations under this Lease.

6.4 <u>Additional Services</u>. Landlord shall also have the exclusive right, but not the obligation, to provide any additional services which may be required by Tenant, including, without limitation, locksmithing (which shall not include any card key access or security system installed by Tenant), additional janitorial service (only if Tenant has not elected to provide its own janitorial service as permitted by Section 6.1.4 above), and additional repairs and maintenance which Tenant requests Landlord to perform, provided that Tenant, as Additional Rent, shall pay to Landlord upon billing, the sum of all costs to Landlord of such additional services plus an administration fee not to exceed five percent (5%) of the cost of the work.

6.5 <u>Energy Usage Reporting</u>. If requested by Landlord, Tenant agrees to deliver to Landlord such information and/or documents as Landlord requires for Landlord to comply with California Public Resources Code Section 25402.10, or successor statute(s), and related California Code of Regulation, relating to commercial building energy ratings. Tenant will provide Landlord with the Energy Star or similar rating for any of Tenant's equipment in the Premises. Alternatively, Landlord may require Tenant to authorize the utility provider for the Premises to release Tenant's utility records to the appropriate government agency or to permit Landlord direct access to Tenant's utility records for the Premises, which Tenant will provide using the applicable utility provider's standard form authorization to receive customer information. Any equipment used by Tenant in the Premises which does not have an ENERGY STAR® or Water Sense® rating may be considered overstandard and be subject to the provisions of Section 6.2 above. Landlord reserves the right to designate the utility provider for the Building and Real Property in its sole and absolute discretion, and may elect to purchase green or renewable energy.

6.6 <u>Cabling</u>. Landlord shall have the right to require that Tenant use a qualified, licensed, bonded contractor reasonably approved by Landlord who meets all of Landlord's requirements for vendors performing work within the Building to install or connect any telecommunications or data cabling from the minimum point of entry (MPOE) in the Building to Tenant's Premises.

6.7 <u>Access</u>. Tenant shall, subject to Landlord's reasonable security requirements, Force Majeure and de minimus interruptions, have access to the Premises twenty-four (24) hours per day, seven (7) days per week.

6.8 <u>Package HVAC Units</u>. Tenant shall be entitled to install, as an initial Tenant Improvement or as an Alteration, dedicated heating, ventilation and air conditioning units ("**Package Units**") within the Premises at Tenant's sole cost and expense. The plans and specifications for any Package Units shall, as indicated in Article 8 below and the Tenant Work Letter (as applicable), be subject to Landlord's reasonable approval. If Tenant elects to install Package Units within the Premises, Tenant shall also install, at Tenant's sole cost and expense, separate meters or at Landlord's option, submeters, in order to measure the amount of electricity furnished to such units and Tenant shall be responsible for Landlord's actual cost of supplying electricity to such units as reflected by such meters or submeters, which amounts shall be payable on a monthly basis as Additional Rent. Tenant shall be solely responsible for maintenance and repair of the Package Units and such units shall be considered to be a fixture within the Premises and shall remain upon the Premises, unless Landlord requires Tenant to remove such Package Units, upon the expiration or earlier termination of the Lease Term or any applicable Option Term.

6.9 Backup Generator. Subject to Landlord's prior approval of all plans and specifications, which approval shall not be unreasonably withheld, and Tenant's receipt of any applicable governmental permits and approvals, Landlord shall permit Tenant to install and maintain, at Tenant's sole cost and expense, a backup diesel-powered generator and uninterrupted power supply (collectively, "Generator") at a location designated by Landlord. Such Generator shall be used by Tenant only during (i) testing and regular maintenance, and (ii) any period of electrical power outage in the Building. Tenant shall be entitled to operate the generator for testing and regular maintenance only upon notice to Landlord and at times reasonably approved by Landlord. Tenant shall submit the specifications for design, operation, installation and maintenance of the Generator for Landlord's consent, which consent shall not be unreasonably withheld or delayed and may be conditioned on Tenant complying with such reasonable requirements imposed by Landlord, based on the advice of Landlord's structural and mechanical engineers, so that the Building's systems and equipment are not adversely affected. In addition, Tenant shall ensure that the Generator does not result in any Hazardous Materials being introduced to the Real Property (other than fuel kept in compliance with Environmental Laws), and Article 29 will apply to Tenant's use of the Generator. Further, Tenant shall be responsible for ensuring that the Generator does not interfere with the use of the Real Property by Landlord or other tenants. In the event another tenant of the Real Property or of a neighboring project complains of problems caused by the Generator, Tenant shall take whatever steps are reasonably necessary to remedy the problem complained of, including removal of the Generator if another solution is not available. Tenant shall ensure that the design and installation of the Generator is performed in a manner so as to minimize or eliminate any noise or vibration cause by such Generator. The vent for the Generator must be higher than the roof line of the Building. Any repairs and maintenance of such Generator shall be the sole responsibility of Tenant and Landlord makes no representation or warranty with respect to such Generator. If Tenant is so notified by Landlord, Tenant shall, at Tenant's sole cost and expense, remove such Generator upon the expiration or earlier termination of the Lease Term and repair all damage to the Real Property resulting from such removal. Such Generator shall be deemed to be a part of the Premises for purposes of Article 10 of this Lease. Landlord will design and construct the necessary pad and enclosure for the Generator at a location designated by Landlord and reasonably approved by Tenant and the cost of such pad and enclosure shall be charged against the Improvement Allowance or paid by Tenant as an Over-Allowance Amount, as applicable.

6.10 <u>Abatement</u>. An "**Abatement Event**" shall be defined as an event that prevents Tenant from using the Premises or any portion thereof, as a result of any failure to provide utilities or services to the Premises, failure to perform any repairs which Landlord is required to perform under this Lease, or due to Renovations performed by Landlord, where (i) Tenant's use of the Premises is materially and adversely impaired or Tenant does not actually use the Premises or such portion thereof, and (ii) such event is caused by the negligence or willful misconduct of Landlord, its agents, employees or contractors. Tenant shall give Landlord and any mortgagee of Landlord (of whom Tenant is notified) notice ("**Abatement Notice**") of any such Abatement Event, and if such Abatement Event continues beyond the "Eligibility Period" (as that term is defined below), then the Base Rent and Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses shall be abated entirely or reduced, as the case may be, after expiration of the Eligibility Period for such time that Tenant continues to be so prevented from using, and does not use, the Premises or a portion thereof, in the proportion that the rentable area of the portion of the

Premises that Tenant is prevented from using, and does not use, bears to the total rentable area of the Premises; provided, however, in the event that Tenant is prevented from using, and does not use, a portion of the Premises for a period of time in excess of the Eligibility Period and the remaining portion of the Premises is not sufficient to allow Tenant to effectively conduct its business therein, and if Tenant does not conduct its business from such remaining portion, then for such time after expiration of the Eligibility Period during which Tenant is so prevented from effectively conducting its business therein, the Base Rent and Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses for the entire Premises shall be abated entirely for such time as Tenant continues to be so prevented from using, and does not use, the Premises. If, however, Tenant reoccupies any portion of the Premises during such period, the Base Rent and Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses allocable to such reoccupied portion, based on the proportion that the rentable area of such reoccupied portion of the Premises bears to the total rentable area of the Premises, shall be payable by Tenant from the date Tenant reoccupies such portion of the Premises. Notwithstanding anything to the contrary contained herein, if Landlord is diligently pursuing the repair of such utilities or services and Landlord provides substitute services reasonably suitable for Tenant's purposes as reasonably determined by Landlord, for example bringing in portable air conditioning or heating equipment, then there shall be no abatement of Base Rent or Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses. The term "Eligibility Period" shall mean a period of five (5) consecutive business days after Landlord's and Landlord's mortgagee's (if applicable), receipt of the applicable Abatement Notice (provided that Landlord will be provided additional time as required to remedy such event so long as Landlord is diligently attempting to remedy such Abatement Event and pursues such remedy to completion). Such right to abate Base Rent and Tenant's Share of Operating Expenses, Insurance Expenses, Utility Expenses and/or Tax Expenses shall be Tenant's sole remedy for an Abatement Event. This Section 6.10 shall not apply in case of damage to, or destruction of, the Premises or the Building, or any eminent domain proceedings which shall be governed by separate provisions of this Lease.

6.11 <u>Security System</u>. Tenant shall be entitled to install, at Tenant's sole cost and expense, a separate security system for the Premises as an Alteration or as a part of the Improvements; provided, however, that the plans and specifications for any such system shall be subject to Landlord's reasonable approval, and any such system must be compatible with the existing systems of the Building, Tenant's obligation to indemnify, defend and hold Landlord harmless as provided in, and subject to, Section 10.1 below shall also apply to Tenant's use and operation of any such system, and the installation of such system shall otherwise be subject to the terms and conditions of Article 8 below. At Landlord's option, upon the expiration or earlier termination of this Lease, Tenant shall remove such security system and repair any damage to the Premises resulting from such removal. Tenant shall at all times provide Landlord with a contact person who can disarm the security system and who is familiar with the functions of the alarm system in the event of a malfunction, and Tenant shall provide Landlord with the alarm codes or other necessary information required to disarm the alarm system in the event Landlord must enter the Premises.

7. REPAIRS

7.1 Landlord's Maintenance. Landlord shall (at Landlord's cost and expense, but subject to inclusion in Operating Expenses to the extent permitted by Article 4 above) maintain the structural portions of the Building including the foundation, floor/ceiling slabs, roof, curtain wall, exterior glass and mullions, columns, beams, shafts (including elevator shafts), stairs, stairwells, elevator cabs and all Common Areas and shall also maintain and repair the basic mechanical, electrical, life safety, plumbing, sprinkler systems (connected to the core but not any distribution throughout the Premises) and heating ventilating and air conditioning systems (including primary loops connected to the core but not any distribution throughout the Premises). Notwithstanding the foregoing, if any of the foregoing repairs are necessitated by the act or omission of Tenant Parties (as defined in Section 10.1 below), then the cost of such repairs, together with an administrative fee of five percent (5%) of such costs, shall be paid by Tenant to Landlord immediately following delivery of an invoice therefor.

7.2 <u>Tenant's Maintenance</u>. Tenant shall, at Tenant's own expense, keep the Premises, including all improvements, fixtures and furnishings therein, in good order, repair and condition at all times during the Lease Term, reasonable wear and tear excepted; provided however, that, at Landlord's option, or if Tenant fails to make such repairs, Landlord may, but need not, make such repairs and replacements, and Tenant shall pay Landlord's costs or expenses, including Landlord's overhead (which shall not exceed five percent (5%) of Landlord's costs or expenses), arising from Landlord's involvement with such repairs and replacements forthwith upon being billed for same. Tenant hereby waives and releases its right to make repairs at Landlord's expense and/or terminate this Lease or vacate the Premises under Section 1942 and Section 1932(1) of the California Civil Code and any other California law, statute, or ordinance now or hereafter in effect.

8. ADDITIONS AND ALTERATIONS

8.1 Landlord's Consent to Alterations. Tenant may not make any improvements, alterations, additions or changes to the Premises (collectively, the "Alterations") without first procuring the prior written consent of Landlord to such Alterations, which consent shall be requested by Tenant not less than thirty (30) days prior to the commencement thereof, and which consent shall not be unreasonably withheld by Landlord. The construction of the initial improvements to the Premises shall be governed by the terms of the Tenant Work Letter and not the terms of this Article 8. Notwithstanding anything to the contrary contained herein, Tenant may make strictly cosmetic changes to the finish work in the Premises (the "Cosmetic Alterations"), without Landlord's consent, provided that the aggregate cost of any such alterations does not exceed \$100,000 in any twelve (12) month period, and further provided that such alterations do not (i) require any structural or other substantial modifications to the Premises, (ii) require any changes to, or adversely affect, the Systems and Equipment of the Building, (iii) affect the exterior appearance of the Building or (iv) trigger any legal requirement which would require Landlord to make any alteration or improvement to the Premises, the Building or the Real Property. Tenant shall give Landlord at least thirty (30) days prior notice of such Cosmetic Alterations, which notice shall be accompanied by reasonably adequate evidence that such changes meet the criteria contained in this Article 8. Except as otherwise provided, the term "Alterations" shall include Cosmetic Alterations.

8.2 Manner of Construction. Tenant shall have obtained Landlord's approval of all plans, specifications, drawings, contractors and subcontractors prior to the commencement of Tenant's construction of the Alterations; provided, however, a contractor of Landlord's selection shall perform all mechanical, electrical, plumbing, structural, and heating, ventilation and air conditioning work (provided that any contractor previously approved by Landlord and used by Tenant for the foregoing trades as part of the initial Improvements will be perform future Alterations in the Premises), and such work shall be performed at Tenant's cost. Tenant agrees to carry "Builder's All Risk" insurance in an amount approved by Landlord covering the construction of such Alterations, and such other insurance as Landlord may require. In addition, Landlord may, in its discretion, for Alterations costing more than \$250,000 in the aggregate (i.e., based on the total cost of the Alterations project at issue), require Tenant to obtain a lien and completion bond or some alternate form of security satisfactory to Landlord in an amount sufficient to ensure the lien-free completion of such Alterations and naming Landlord as a co-obligee. Further, Tenant shall pay to Landlord or its agent a supervision fee not to exceed five percent (5%) based upon the cost of such work (which fee shall be based upon the amount charged to Landlord by its third party construction management consultant or if Landlord acts as its own construction manager, based upon the actual cost to Landlord to provide such construction supervision). Tenant shall construct such Alterations and perform such repairs in conformance with any and all applicable laws and pursuant to a valid building permit, issued by the appropriate governmental authorities, in conformance with Landlord's construction rules and regulations and in a manner which is in keeping with any third party rating system in effect at the Building (e.g., LEED) and in a diligent, good and workmanlike manner. If such Alterations trigger a legal requirement upon Landlord to make any Alterations or improvements to the Building or Common Areas, Tenant shall, as Additional Rent, reimburse Landlord for the cost thereof within thirty (30) days following receipt of an invoice therefor. Landlord's approval of the plans, specifications and working drawings for Tenant's Alterations shall create no responsibility or liability on the part of Landlord for their completeness, design sufficiency, or compliance with all laws, rules and regulations of governmental agencies or authorities. Upon completion of any Alterations, Tenant agrees to cause a Notice of Completion (or equivalent) to be recorded in the office of the Recorder of the County of San Diego in accordance with Section 3093 of the Civil Code of the State of California or any successor statute, and Tenant shall deliver to the Building management office a reproducible copy of the "as built" drawings of the Alterations. All Alterations will comply with Landlord's sustainability practices and any third party rating system concerning the Premises or the Building then in effect.

Tenant acknowledges that Landlord has advised Tenant that the Building may contain asbestos-containing materials ("ACMs"). If Tenant undertakes any Alterations as permitted by this Article 8, Tenant shall, in addition to complying with the requirements of this Article 8, shall undertake the Alterations in a manner that avoids disturbing any ACMs present in the Building. If ACMs are likely to be disturbed in the course of such work, Tenant shall encapsulate or remove the ACMs in accordance with an approved asbestos-removal plan and otherwise in accordance with all Applicable Laws, including giving all notices required by California Health and Safety Code Sections 25915-25919.7.

8.3 <u>Landlord's Property</u>. All Alterations and fixtures which may be made, installed or placed in or about the Premises from time to time, shall be at the sole cost of Tenant and shall be

and become the property of Landlord; however, Landlord may, by written notice to Tenant at the time of Landlord's consent to such Alterations, require Tenant at Tenant's expense to remove any such Alterations or fixtures. If Tenant fails to complete such removal and/or to repair any damage caused by the removal of any Alterations, Improvements or fixtures, Landlord may do so and may charge the cost thereof to Tenant. Tenant will not be required to remove or restore any of the initial Improvements installed in accordance with the Tenant Work Letter attached hereto as Exhibit B.

8.4 <u>Refurbishment</u>. Effective as of the first day of the sixth (6th) full Lease Year during the initial Lease Term only, Tenant shall be entitled to a one-time tenant refurbishment allowance (the "**Refurbishment Allowance**") in the amount of \$130,960.00 (i.e., \$5.00 per rentable square foot of the Premises) for the costs relating to the design and construction of certain renovations in the Premises that are to be permanently affixed to the Premises (the "**Refurbished Improvements**"). In no event shall Landlord be obligated to make disbursements under this Section 8.4 in a total amount which exceeds the Refurbishment Allowance and in no event shall Landlord be obligated to disburse the Refurbishment Allowance unless and until Tenant has completed the Refurbished Improvements. All Refurbished Improvements will be constructed in accordance with this Article 8. If Tenant fails to complete the Refurbished Improvements and request the Refurbishment Allowance within twelve (12) months after the first day of the sixth (6th) full Lease Year during the initial Lease Term, Tenant will be deemed to have waived the Refurbishment Allowance and Landlord shall have no further obligation to pay such Refurbishment Allowance.

9. COVENANT AGAINST LIENS

Tenant covenants and agrees not to suffer or permit any lien of mechanics or materialmen or others to be placed against the Real Property, the Building or the Premises with respect to work or services claimed to have been performed for or materials claimed to have been furnished to Tenant or the Premises, and, in case of any such lien attaching or notice of any lien, Tenant covenants and agrees to cause it to be immediately released and removed of record. Notwithstanding anything to the contrary set forth in this Lease, in the event that such lien is not released and removed within twenty (20) days after the date notice of such lien is delivered by Landlord to Tenant, Landlord, at its sole option, may immediately take all action necessary to release and remove such lien, without any duty to investigate the validity thereof, and all sums, costs and expenses, including reasonable attorneys' fees and costs, incurred by Landlord in connection with such lien shall be deemed Additional Rent under this Lease and shall immediately be due and payable by Tenant.

10. INSURANCE

10.1 <u>Indemnification and Waiver</u>. Tenant shall be liable for, and shall indemnify, defend, protect and hold Landlord and Landlord's partners, officers, directors, employees, agents, successors and assigns (collectively, "Landlord Indemnified Parties") harmless from and against, any and all claims, damages (including direct and consequential or indirect damages), judgments, suits, causes of action, losses, liabilities and expenses, including reasonable attorneys' fees and court costs (collectively, "Indemnified Claims"), arising or resulting from (a) any negligent or willful act or omission of Tenant or any of Tenant's agents, employees, contractors, subtenants, assignees, invitees or licensees in or about the Premises, the Building or the Real Property (collectively, "Tenant Parties"); (b) any occurrence within the Premises unless caused by the gross negligence or willful

misconduct of Landlord; (c) the use or occupancy of the Premises by Tenant or the Tenant Parties and/or (d) any default by Tenant of any obligations on Tenant's part to be performed under the terms of this Lease. Except for damage resulting from Landlord's gross negligence or material misconduct, Tenant hereby assumes all risk of damage to property or injury to persons in or about the Premises, the Building or elsewhere on the Real Property from any cause, and Tenant hereby waives all claims in respect thereof against Landlord. Notwithstanding anything to the contrary in this Lease, Landlord shall in no event be liable to Tenant or any Tenant Party for any consequential damages, special or punitive damages or for loss of business, revenue, profits or income and Tenant hereby waives any and all claims for any such damages. Landlord shall not be liable for interference with light or other intangible rights. Notwithstanding anything to the contrary in this Lease, Tenant shall not be liable under any circumstances for any consequential damages, special or punitive damages or for loss of business, revenue, profits or income, in each case, however occurring, except that the foregoing waiver will not apply to liability of Tenant pursuant to Article 16 or 29 of this Lease and the parties stipulate that the damages recoverable by Landlord pursuant to the California Civil Code in the event of a Tenant default shall not be deemed consequential damages and the foregoing waiver will not apply to such damages. The provisions of this Section 10.1 shall survive the expiration or sooner termination of this Lease with respect to any claims or liability occurring prior to such expiration or termination.

In all instances in this Lease where Tenant is obligated to defend Landlord as part of an indemnity obligation of Tenant, such defense obligation will be modified in accordance with the terms of this paragraph. In the event any Claim for which Tenant is required to indemnify Landlord is covered by insurance, the terms of the applicable indemnity will not be modified. However, for Claims which are not covered by insurance, (i) Tenant shall have the option to assume the defense with counsel selected (and reasonably approved by Landlord) and controlled by Tenant, or (ii) if Tenant does not elect to directly secure counsel for Landlord, Landlord will secure counsel of its choice to defend Landlord and will control all aspects of such defense, and Tenant will reimburse Landlord for the reasonable costs of such defense as such costs are incurred. Such defense costs will be reimbursed to Landlord within forty-five (45) days after invoicing, which invoices must contain a reasonably detailed explanation of the time expended on such defense for the applicable billing period.

10.2 <u>Tenant's Compliance with Landlord's Fire and Casualty Insurance</u>. Tenant shall, at Tenant's expense, comply with all insurance company requirements pertaining to the use of the Premises. If Tenant's conduct or use of the Premises causes any increase in the premium for such insurance policies, then Tenant shall reimburse Landlord for any such increase.

10.3 <u>Tenant's Insurance</u>. Tenant shall maintain the following coverages in the following amounts.

10.3.1 Commercial General Liability Insurance, on an occurrence basis, covering the insured against claims of bodily injury, personal injury and property damage arising out of Tenant's operations, assumed liabilities or use of the Premises, including a Broad Form Commercial General Liability endorsement covering the insuring provisions of this Lease and the performance by Tenant of the indemnity agreements set forth in Section 10.1 of this Lease, for limits of liability

not less than: (i) \$3,000,000 each occurrence and annual aggregate for bodily injury and property damage liability; and (ii) \$3,000,000 each occurrence and annual aggregate, with 0% insured's participation, for personal injury liability.

10.3.2 Physical Damage Insurance covering (i) all office furniture, trade fixtures, office equipment, merchandise and all other items of Tenant's property on the Premises installed by, for, or at the expense of Tenant, and (ii) the Tenant Improvements and Alterations. Such insurance shall be written on an "all risks" of physical loss or damage basis, for the full replacement cost value new without deduction for depreciation of the covered items and in amounts that meet any co-insurance clauses of the policies of insurance and shall include a vandalism and malicious mischief endorsement, sprinkler leakage coverage and earthquake sprinkler leakage coverage.

10.3.3 Loss of income and extra expense insurance in such amounts as will reimburse Tenant for direct or indirect loss of earnings attributable to all perils commonly insured against by prudent tenants or attributable to prevention of access to the Premises or to the Building as a result of such perils.

10.3.4 Worker's compensation insurance providing statutory benefits to Tenant's employees, with a waiver of subrogation in favor of Landlord, and employer's liability insurance with limits of not less than the statutory limits required for worker's compensation insurance.

10.4 Form of Policies. The minimum limits of policies of insurance required of Tenant under this Lease shall in no event limit the liability of Tenant under this Lease. All insurance shall (i) be issued by an insurance company having a rating of not less than A-X in Best's Insurance Guide or which is otherwise acceptable to Landlord and licensed to do business in the State of California; and (ii) provide that said insurance shall not be canceled or coverage changed unless prior written notice shall have been given to Landlord and the other additional insureds thereunder designated by Landlord. In addition, the insurance described in Section 10.3.1 above shall (a) name Landlord, and any other party specified by Landlord, as an additional insured; (b) specifically cover the liability assumed by Tenant under this Lease including, but not limited to, Tenant's obligations under Section 10.1 of this Lease; (c) be primary insurance as to all claims thereunder and provide that any insurance required by Landlord is excess and is non-contributing with any insurance requirement of Tenant; and (d) contain a cross-liability endorsement or severability of interest clause acceptable to Landlord. In addition, the insurance described in Section 10.3.2 shall name Landlord as loss-payee as to all items referred to in clause (ii) of Section 10.3.2 and the insurance described in Sections 10.3.2 and 10.3.3 shall have commercially reasonable deductibles. Tenant shall deliver all policies or certificates thereof to Landlord on or before Landlord's delivery of the Premises to Tenant or the Lease Commencement Date, whichever first occurs, and before the expiration dates thereof. In the event Tenant shall fail to procure such insurance, or to deliver such policies or certificate, Landlord may deny Tenant the right to occupy the Premises until such time as Tenant makes such deliveries (which denial shall have no effect upon the Lease Commencement Date) and/or procure such policies for the account of Tenant, and the cost thereof shall be paid to Landlord as Additional Rent within five (5) days after delivery to Tenant of the bills therefor.

10.5 <u>Subrogation</u>. Landlord and Tenant agree to have their respective insurance companies issuing property damage, worker's compensation insurance and loss of income and extra expense insurance waive any rights of subrogation that such companies may have against Landlord or Tenant, as the case may be. Notwithstanding anything in this Lease to the contrary, Landlord and Tenant hereby waive any right that either may have against the other on account of any loss or damage if such loss or damage is insurable under the property damage or loss of income and extra expense insurance required to be maintained hereunder (this waiver extends to deductibles under such insurance).

11. DAMAGE AND DESTRUCTION

11.1 Repair of Damage to Premises by Landlord. Tenant shall promptly notify Landlord of any damage to the Premises resulting from fire or any other casualty. If the Premises or any Common Areas of the Building serving or providing access to the Premises shall be damaged by fire or other casualty, Landlord shall promptly and diligently, subject to reasonable delays for insurance adjustment or other matters beyond Landlord's reasonable control, and subject to all other terms of this Article 11, restore the Base, Shell and Core of the Premises and such Common Areas to substantially the same condition as existed prior to the casualty, except for modifications required by law, the holder of a mortgage on the Real Property, the lessor of a ground or underlying lease, or any other modifications to the Common Areas deemed desirable by Landlord. Notwithstanding any other provision of this Lease, upon the occurrence of any damage to the Premises resulting from fire or other casualty, Tenant shall assign to Landlord all insurance proceeds payable to Tenant as to items of property described in clause (ii) of Section 10.3.2, and Landlord shall return the Tenant Improvements and Alterations to their original condition; provided that if the cost of such repair by Landlord exceeds the amount of insurance proceeds received by Landlord from Tenant's insurance carrier, as assigned by Tenant, the cost of such repairs shall be paid by Tenant to Landlord prior to Landlord's repair of the damage. In the event any damage to the Building or Common Area occurs as a result of the negligence or willful misconduct of Tenant and/or its agents, contractors, employees and/or invitees, Tenant shall reimburse Landlord, promptly on demand, for the cost of incurred by Landlord in repairing such damage and the provisions of Section 10.5 regarding Landlord's deductible shall not apply to such reimbursement obligation. Landlord shall not be liable for any inconvenience or annovance to Tenant or its visitors, or injury to Tenant's business resulting in any way from damage resulting from fire or other casualty or Landlord's repair thereof; provided however, that if such fire or other casualty shall have damaged the Premises or Common Areas necessary to Tenant's occupancy, and if such damage is not the result of the negligence or willful misconduct of Tenant or Tenant's employees, contractors, licensees, or invitees, Landlord shall allow Tenant a proportionate abatement of Rent to the extent Landlord is reimbursed from the proceeds of rental interruption insurance purchased by Landlord as part of Operating Expenses, during the time and to the extent the Premises are unfit for occupancy for the purposes permitted under this Lease, and not occupied by Tenant as a result thereof.

11.2 <u>Landlord's Option to Repair</u>. Notwithstanding the terms of Section 11.1 of this Lease, Landlord may elect not to rebuild and/or restore the Premises and/or Building and instead terminate this Lease by notifying Tenant in writing of such termination within sixty (60) days after the date Landlord learns of the necessity for repairs as the result of damage, such notice to include a

termination date giving Tenant ninety (90) days to vacate the Premises (provided that in the event it is permissible by Applicable Laws and will not impair Landlord's repair or demolition activities at the Building or result in any threat to persons or property, Landlord will permit Tenant up to an additional sixty (60) days to vacate if required by Tenant to find replacement Premises), but Landlord may so elect only if the Building shall be damaged by fire or other casualty or cause, whether or not the Premises are affected, and one or more of the following conditions is present: (i) repairs cannot reasonably be completed within one hundred twenty (120) days after the date Landlord learns of the necessity for repairs as the result of damage (when such repairs are made without the payment of overtime or other premiums); (ii) the holder of any mortgage on the Real Property or ground or underlying lessor with respect to the Real Property shall require that the insurance proceeds or any portion thereof be used to retire the mortgage debt, or shall terminate the ground or underlying lease, as the case may be; (iii) the damage is not fully covered, except for deductible amounts, by Landlord's insurance policies; or (iv) such damage occurs during the last twenty-four (24) months of the Lease Term. Within sixty (60) days after the date Landlord learns of the necessity for repairs as a result of damage, Landlord shall notify Tenant ("Damage Repair Estimate") of Landlord's estimated assessment of the period of time in which the repairs will be completed, which assessment shall be based upon the opinion of a contractor reasonably selected by Landlord and experienced in comparable repairs of first class office buildings. If Landlord does not elect to terminate this Lease pursuant to Landlord's termination right as provided above, and the Damage Repair Estimate indicates that repairs cannot be completed within one hundred eighty (180) days after being commenced, Tenant may elect, not later than thirty (30) days after Tenant's receipt of the Damage Repair Estimate, to terminate this Lease by written notice to Landlord effective as of the date specified in Tenant's notice. In the event that the Premises or the Project is destroyed or damaged to any substantial extent during the last twelve (12) months of the Lease Term and if such damage shall take longer than sixty (60) days to repair and if such damage is not the result of the negligence or willful misconduct of Tenant or Tenant's employees, licensees, invitees or agents, then notwithstanding anything in this Article 11 to the contrary, Tenant shall have the option to terminate this Lease by written notice to Landlord of the exercise of such option within sixty (60) days after Tenant learns of the necessity for repairs as the result of such damage.

11.3 <u>Waiver of Statutory Provisions</u>. The provisions of this Lease, including this Article 11, constitute an express agreement between Landlord and Tenant with respect to any and all damage to, or destruction of, all or any part of the Premises, the Building or any other portion of the Real Property, and any statute, regulation or case law of the State of California, including without limitation, Sections 1932(2) and 1933(4) of the California Civil Code, with respect to termination rights arising from damage or destruction shall have no application to this Lease or any damage or destruction to all or any part of the Premises, the Building or any other portion of the Real Property.

12. NONWAIVER

No waiver of any provision of this Lease shall be implied by any failure of either party to enforce any remedy on account of the violation of such provision, even if such violation shall continue or be repeated subsequently. Any waiver by either party of any provision of this Lease may only be in writing, and no express waiver shall affect any provision other than the one specified

in such waiver and that one only for the time and in the manner specifically stated. No receipt of monies by Landlord from Tenant after the termination of this Lease shall in any way alter the length of the Lease Term.

13. CONDEMNATION

If the whole or any part of the Premises, Building or Real Property shall be taken by power of eminent domain or condemned by any competent authority for any public or quasi-public use or purpose, or if any adjacent property or street shall be so taken or condemned, or reconfigured or vacated by such authority in such manner as to require the use, reconstruction or remodeling of any part of the Premises, Building or Real Property, or if Landlord shall grant a deed or other instrument in lieu of such taking by eminent domain or condemnation, Landlord shall have the option to terminate this Lease upon ninety (90) days' notice, provided such notice is given no later than one hundred eighty (180) days after the date of such taking, condemnation, reconfiguration, vacation, deed or other instrument. If more than fifteen percent (15%) of the rentable square feet of the Premises is taken, or if access to the Premises is substantially impaired, Tenant shall have the option to terminate this Lease upon ninety (90) days' notice, provided such notice is given no later than one hundred eighty (180) days after the date of such taking. Landlord shall be entitled to receive the entire award or payment in connection with such taking, except that Tenant shall have the right to file any separate claim available to Tenant for any taking of Tenant's personal property belonging to Tenant and removable by Tenant upon expiration of the Lease Term pursuant to the terms of this Lease, and for goodwill and moving expenses, so long as such claim does not diminish the award available to Landlord, its ground lessor with respect to the Real Property or its mortgagee, and such claim is payable separately to Tenant. If any part of the Premises shall be taken, and this Lease shall not be so terminated, the Rent shall be proportionately abated. Tenant hereby waives any and all rights it might otherwise have pursuant to Section 1265.130 of the California Code of Civil Procedure, or any other California law, statute or ordinance now or hereafter in effect, to seek termination of this Lease in the event of a taking.

14. ASSIGNMENT AND SUBLETTING

14.1 <u>Transfers</u>. Tenant shall not, without the prior written consent of Landlord, which shall not be unreasonably withheld if Landlord does not elect to proceed under Section 14.4 below, assign or otherwise transfer this Lease or any interest hereunder, permit any assignment or other such foregoing transfer of this Lease or any interest hereunder by operation of law, sublet the Premises or any part thereof, or permit the use of the Premises by any persons other than Tenant and its employees (all of the foregoing are hereinafter sometimes referred to collectively as "**Transfers**" and any person to whom any Transfer is made or sought to be made is hereinafter sometimes referred to as a "**Transferee**"). If Tenant shall desire Landlord's consent to any Transfer, Tenant shall notify Landlord in writing, which notice (the "**Transfer Notice**") shall include (i) the proposed effective date of the Transfer, which shall not be less than thirty (30) days nor more than one hundred eighty (180) days after the date of delivery of the Transfer Notice, (ii) a description of the portion of the Premises to be transferred (the "**Subject Space**"), (iii) all of the terms of the proposed Transfer and the consideration therefor, including a calculation of the "Transfer Premium," as that term is defined in Section 14.3 below, in connection with such Transfer, the name and address

of the proposed Transferee and a copy of all operative documents to be executed to evidence such Transfer or the agreements incidental or related to such Transfer, and (iv) current financial statements of the proposed Transferee and such other information as Landlord may reasonably require. If there are any changes in the terms and conditions from those specified in the Transfer Notice (i) such that Landlord would initially have been entitled to refuse its consent to such Transfer under this Section 14.1, or (ii) which would cause the proposed Transfer to be more favorable to the Transferee than the terms set forth in Tenant's original Transfer Notice, Tenant shall again submit the Transfer to Landlord for its approval and other action under this Article 14 (including Landlord's right of recapture, if any, under Section 14.4 of this Lease). As a condition to Landlord's consent to any proposed assignment of this Lease, Landlord may, at Landlord's sole option, require the assigning Tenant to guaranty the proposed assignee's obligations under this Lease, as assigned; and, if Landlord so elects, the assigning Tenant shall execute a Guaranty of Lease in a form reasonably acceptable to Landlord. Any Transfer made without Landlord's prior written consent shall, at Landlord's option, be null, void and of no effect, and shall, at Landlord's option, constitute a default by Tenant under this Lease. Whether or not Landlord shall grant consent, Tenant shall pay Landlord's review and processing fees, as well as any reasonable legal fees incurred by Landlord, (up to a total amount not to exceed \$1500) within thirty (30) days after written request by Landlord. Notwithstanding any contrary provision of this Lease, if Tenant or any proposed Transferee claims that Landlord has unreasonably withheld or delayed its consent to a proposed Transfer or otherwise has breached its obligations under this Article 14, Tenant's and such Transferee's only remedy shall be to seek a declaratory judgment and/or injunctive relief, and Tenant, on behalf of itself and, to the extent permitted by law, such proposed Transferee waives all other remedies against Landlord, including without limitation, the right to seek monetary damages or to terminate this Lease.

Landlord's Consent. Landlord will respond to any request for consent to a proposed Transfer within ten (10) days 14.2 after receipt of all information required by this Article 14 and shall not unreasonably withhold its consent to any proposed Transfer of the Subject Space to the Transferee on the terms specified in the Transfer Notice. The parties hereby agree that it shall be reasonable under this Lease and under any applicable law for Landlord to withhold consent to any proposed Transfer where one or more of the following apply, without limitation as to other reasonable grounds for withholding consent: (i) the Transferee is of a character or reputation or engaged in a business which is not consistent with the quality of the Building, or would be a significantly less prestigious occupant of the Building than Tenant; (ii) the Transferee intends to use the Subject Space for purposes which are not permitted under this Lease; (iii) the Transferee is either a governmental agency or instrumentality thereof; (iv) the Transfer will result in more than a reasonable and safe number of occupants per floor within the Subject Space; (v) the Transferee is not a party of reasonable financial worth and/or financial stability in light of the responsibilities involved under the Lease on the date consent is requested; (vi) the proposed Transfer would cause Landlord to be in violation of another lease or agreement to which Landlord is a party, or would give an occupant of the Building a right to cancel its lease; (vii) the terms of the proposed Transfer will allow the Transferee to exercise a right of renewal, right of expansion, right of first offer, or other similar right held by Tenant (or will allow the Transferee to occupy space leased by Tenant pursuant to any such right); (viii) either the proposed Transferee, or any person or entity which directly or indirectly, controls, is controlled by, or is under common control with, the proposed Transferee, (A) occupies space in the Building at the time of the request for consent, (B) is negotiating with Landlord to lease space in the Building

at such time, or (C) has negotiated with Landlord during the three (3)-month period immediately preceding the Transfer Notice; (ix) the transferee or its proposed use would jeopardize any LEED or similar rating in effect at the Building; or (x) Landlord has not received assurances acceptable to Landlord that all past due amounts owing by Tenant to Landlord, if any, will be paid and all defaults on the part of Tenant, if any, will be cured prior to the effective date of the proposed Transfer.

14.3 <u>Transfer Premium</u>. If Landlord consents to a Transfer, Tenant shall pay to Landlord within thirty (30) days of Tenant's receipt from Transferee, fifty percent (50%) of any "Transfer Premium," as that term is defined in this Section 14.3, received by Tenant from such Transferee. "**Transfer Premium**" shall mean all rent, additional rent or other consideration payable by such Transferee in excess of the Rent and Additional Rent, if any, payable by Tenant under this Lease on a per rentable square foot basis if less than all of the Premises is transferred, after deducting the reasonable expenses incurred by Tenant for (i) any changes, alterations and improvements to the Premises in connection with the Transfer, and (ii) any brokerage commissions in connection with the Transfer (collectively, the "**Subleasing Costs**"). "Transfer Premium" shall also include key money and bonus money paid by Transferee to Tenant in connection with such Transfer. Landlord or its authorized representatives shall have the right at all reasonable times to audit the books, records and papers of Tenant relating to any Transfer, and shall have the right to make copies thereof. If the Transfer Premium respecting any Transfer shall be found understated, Tenant shall, within thirty (30) days after demand, pay the deficiency and Landlord's costs of such audit.

14.4 <u>Landlord's Option as to Subject Space</u>. Notwithstanding anything to the contrary contained in this Article 14, Landlord shall have the option, by giving written notice to Tenant within thirty (30) days after receipt of any Transfer Notice which requests consent to the Transfer of the entire Premises for the entire remaining Term of this Lease, to recapture the Premises from Tenant. Such recapture notice shall cancel and terminate this Lease as of the date stated in the Transfer Notice as the effective date of the proposed Transfer. If Landlord declines, or fails to elect in a timely manner to recapture the Subject Space under this Section 14.4, then, provided Landlord has consented to the proposed Transfer, Tenant shall be entitled to proceed to transfer the Subject Space to the proposed Transferee.

14.5 <u>Effect of Transfer</u>. If Landlord consents to a Transfer, (i) the terms and conditions of this Lease shall in no way be deemed to have been waived or modified, (ii) such consent shall not be deemed consent to any further Transfer by either Tenant or a Transferee, (iii) Tenant shall deliver to Landlord, promptly after execution, an original executed copy of all documentation pertaining to the Transfer in form reasonably acceptable to Landlord, and (iv) no Transfer relating to this Lease or agreement entered into with respect thereto, whether with or without Landlord's consent, shall relieve Tenant or any guarantor of the Lease from liability under this Lease. Landlord may consent to future Transfers of this Lease without notifying Tenant and without obtaining Tenant's consent thereto. In the event of a default by any Transferee of Tenant or any successor to Tenant's interest in this Lease, Landlord may proceed directly against Tenant without the necessity of exhausting remedies against such Transferee or successor.

14.6 <u>Additional Transfers</u>. For purposes of this Lease, the term "Transfer" shall also include a change in the ownership of fifty percent (50%) or more of the ownership interests of

Tenant within a twelve (12)-month period, unless Tenant is a publicly-held company whose stock trades on a nationally-recognized exchange.

14.7 Affiliate Transfers. Notwithstanding anything to the contrary contained in this Article 14, an assignment or subletting of all or a portion of the Premises to an affiliate ("Affiliate") of Tenant (an entity which is controlled by, controls, or is under common control with, Tenant, or that becomes a parent, successor or affiliate of Tenant, or is a successor of Tenant by reason of merger, consolidation, public offering, reorganization, dissolution, or sale of stock, membership or partnership interests or assets) shall not be deemed a Transfer under this Article 14, provided that (i) Tenant notifies Landlord of any such assignment or sublease prior to the effective date thereof and promptly supplies Landlord with any documents or information requested by Landlord regarding such assignment or sublease to such Affiliate (including, in the event of an assignment, evidence of the assignee's assumption of Tenant's obligations under this Lease or, in the event of a sublease, evidence of the sublessee's assumption, in full, of the obligations of Tenant with respect to the portion of the Premises so subleased, other than the payment of rent), (ii) such assignment or sublease is not a subterfuge by Tenant to avoid its obligations under this Lease, (iii) such assignment or sublease does not cause Landlord to be in default under any existing lease at the Building, and (iv) the net worth of such Affiliate is not less than reasonably required to fulfill the terms of this Lease. An assignee of Tenant's entire interest in this Lease pursuant to the immediately preceding sentence may be referred to herein as an "Affiliated Assignee." "Control," as used in this Article 14 shall mean the ownership, directly or indirectly, of greater than fifty-one percent (51%) of the voting securities of, or possession of the right to vote, in the ordinary direction of its affairs, of greater than fifty-one percent (51%) of the voting interest in, an entity. Nothing contained in this Section 14.7 shall be deemed to release Tenant from its obligations under this Lease.

15. SURRENDER OF PREMISES AND REMOVAL OF TENANT'S PROPERTY

No act or thing done by Landlord or any agent or employee of Landlord during the Lease Term shall be deemed to constitute an acceptance by Landlord of a surrender of the Premises unless such intent is specifically acknowledged in a writing signed by Landlord. Upon the expiration of the Lease Term, or upon any earlier termination of this Lease, Tenant shall quit and surrender possession of the Premises to Landlord in as good order and condition as when Tenant took possession and as thereafter improved by Landlord and/or Tenant, reasonable wear and tear and repairs which are specifically made the responsibility of Landlord hereunder excepted. Upon such expiration or termination, Tenant shall remove from the Premises all debris and rubbish, such items of furniture, equipment, and cabling installed by or at the request of Tenant that is not contained in protective conduit or metal raceway, other articles of personal property owned by Tenant and any property Landlord requires Tenant to remove pursuant to Section 8.3. Tenant shall repair at its own expense all damage to the Premises and Building resulting from such removal. In addition, if Tenant fails to remove Tenant's personal property from the Premises after the expiration or earlier termination of this Lease, Landlord may (but shall not be obligated to) deem all or any part of Tenant's personal property to be abandoned, and title to Tenant's personal property (except with respect to any Hazardous Material (defined in Paragraph 29)) shall be deemed to be immediately vested in Landlord with no obligation on the part of Landlord to compensate Tenant for such property,

but Tenant shall remain responsible for the cost of removal, storage and/or disposal of such property, as well as any damage caused by such removal.

16. HOLDING OVER

If Tenant holds over after the expiration of the Lease Term hereof, with or without the consent of Landlord, such tenancy shall be from month-to-month only, and Base Rent shall be payable at a monthly rate equal to the one hundred fifty percent (150%) of the Base Rent applicable during the last rental period of the Lease Term under this Lease. Such month-to-month tenancy shall be subject to every other term, covenant and agreement contained herein. Such holdover shall not constitute a renewal or extension of the Lease Term and Landlord expressly reserves the right to require Tenant to surrender possession of the Premises to Landlord as provided in this Lease upon the expiration or other termination of this Lease. The provisions of this Article 16 shall not be deemed to limit or constitute a waiver of any other rights or remedies of Landlord provided herein or at law. If Tenant fails to surrender the Premises upon the termination or expiration of this Lease, Tenant shall be liable for all damages incurred by Landlord resulting therefrom, including without limitation any consequential damages or damages resulting from Landlord's inability to deliver the Premises to a successor tenant.

17. ESTOPPEL CERTIFICATES

Within ten (10) business days following a request in writing by Landlord, Tenant shall execute and deliver to Landlord an estoppel certificate on Landlord's or its lender's standard form.

18. SUBORDINATION

This Lease is subject and subordinate to all present and future ground or underlying leases of the Real Property and to the lien of any mortgages or trust deeds, now or hereafter in force against the Real Property and the Building, if any, and to all renewals, extensions, modifications, consolidations and replacements thereof, and to all advances made or hereafter to be made upon the security of such mortgages or trust deeds, unless the holders of such mortgages or trust deeds, or the lessors under such ground lease or underlying leases, require in writing that this Lease be superior thereto; provided that a condition to Tenant's obligation to subordinate to any future lender will be that Tenant receives from such future lender a commercially reasonable SNDA (as defined below). Tenant covenants and agrees to attorn, without any deductions or set-offs whatsoever, to lender or holder of any mortgage or deed of trust upon foreclosure, the purchaser upon any foreclosure sale, or to the lessor of a ground or underlying lease upon the termination thereof, as the case may be, if so requested to do so by such purchaser or lessor, and to recognize such lender, purchaser or lessor as the lessor under this Lease. Tenant shall, within ten (10) business days of request by Landlord, execute such further instruments or assurances as Landlord may reasonably deem necessary to evidence or confirm such attornment and/or the subordination or superiority of this Lease to any such mortgages, trust deeds, ground leases or underlying leases. Within thirty (30) days after the full execution and delivery of this Lease, Landlord shall, at no cost to Landlord, attempt to obtain a commercially reasonable subordination, nondisturbance and attornment agreement ("SNDA") from the current lender of the Building; provided that in the event Landlord is unable to obtain the SNDA, this Lease and Tenant's obligations hereunder will not be affected.

19. DEFAULTS; REMEDIES

19.1 <u>Tenant Default</u>. The occurrence of any of the following shall constitute a default of this Lease by Tenant:

19.1.1 Any failure by Tenant to pay any Rent or any other charge required to be paid under this Lease, or any part thereof, within three (3) business days after written notice the same is past due; or

19.1.2 Any failure by Tenant (other than a failure pursuant to Section 19.1.1 or 19.1.4) to observe or perform any other provision, covenant or condition of this Lease to be observed or performed by Tenant where such failure continues for thirty (30) days after written notice thereof from Landlord to Tenant; provided however, that if the nature of such default is such that the same cannot reasonably be cured within a thirty (30)-day period, Tenant shall not be deemed to be in default if it diligently commences such cure within such period and thereafter diligently proceeds to rectify and cure said default as soon as possible; or

19.1.3 The entry of an order for relief with respect to Tenant under any chapter of the Federal Bankruptcy Code, the dissolution or liquidation of Tenant, the insolvency of Tenant or the inability of Tenant to pay its debts when due, or the appointment of a trustee or receiver to take possession of all or substantially all of Tenant's assets or Tenant's interest under this Lease that is not discharged within thirty (30) days; or

19.1.4 The failure of Tenant to execute any documents referenced in Article 17 or 18 within two (2) business days after notice from Landlord that such documents were not delivered on time.

Any notice required under this Section 19.1 shall be in lieu of, and not in addition to, any notice required under California Code of Civil Procedure Section 1161 or any successor law.

19.2 <u>Remedies Upon Default</u>. Upon the occurrence of any event of default by Tenant, Landlord shall have, in addition to any other remedies available to Landlord at law or in equity, the option to pursue any one or more of the following remedies, each and all of which shall be cumulative and nonexclusive, without any notice or demand whatsoever.

19.2.1 Terminate this Lease, in which event Tenant shall immediately surrender the Premises to Landlord, and if Tenant fails to do so, Landlord may, without prejudice to any other remedy which it may have for possession or arrearages in rent, enter upon and take possession of the Premises and expel or remove Tenant and any other person who may be occupying the Premises or any part thereof, without being liable for prosecution or any claim or damages therefor; and Landlord may recover from Tenant the following: (i) the worth at the time of award of any unpaid rent which has been earned at the time of such termination; plus (ii) the worth at the time of award of the amount by which the unpaid rent which would have been reasonably avoided; plus (iii) the worth at the time of award of the amount by which the unpaid rent for the balance of the Lease Term after the time of award exceeds the amount of such rental loss that Tenant

proves could have been reasonably avoided; plus (iv) any other amount necessary to compensate Landlord for all the detriment proximately caused by Tenant's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom. The term "rent" as used in this Section 19.2 shall be deemed to be and to mean all sums of every nature required to be paid by Tenant pursuant to the terms of this Lease, whether to Landlord or to others. As used in clauses (i) and (ii), above, the "worth at the time of award" shall be computed by allowing interest at the rate set forth in Article 25 of this Lease. As used in clause (iii) above, the "worth at the time of award" shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of award plus one percent (1%).

19.2.2 Landlord shall have the remedy described in California Civil Code Section 1951.4. If Landlord does not elect to terminate this Lease on account of any default by Tenant, Landlord may, from time to time, without terminating this Lease, enforce all of its rights and remedies under this Lease, including the right to recover all rent as it becomes due.

19.2.3 For the purposes of this Article 19, Tenant's right to possession shall not be deemed to have been terminated by efforts of Landlord to relet the Premises, by its acts of maintenance or preservation with respect to the Premises, or by appointment of a receiver to protect Landlord's interests hereunder. The foregoing enumeration is not exhaustive, but merely illustrative of acts which may be performed by Landlord without terminating Tenant's right to possession.

19.2.4 Whether or not Landlord elects to terminate this Lease on account of any default by Tenant, as set forth in this Article 19, Landlord shall have the right to terminate any and all subleases, licenses, concessions or other consensual arrangements for possession entered into by Tenant and affecting the Premises or may, in Landlord's sole discretion, succeed to Tenant's interest in such subleases, licenses, concessions or arrangements

19.3 <u>Form of Payment After Default</u>. Following the occurrence of an event of default by Tenant, Landlord shall have the right to require that any or all subsequent amounts paid by Tenant to Landlord hereunder, whether in the cure of the default in question or otherwise, be paid in the form of cash, money order, cashier's or certified check drawn on an institution acceptable to Landlord, or by other means approved by Landlord, notwithstanding any prior practice of accepting payments in any different form.

19.4 <u>Waiver of Default</u>. No waiver by Landlord or Tenant of any violation or breach of any of the terms, provisions and covenants herein contained shall be deemed or construed to constitute a waiver of any other or later violation or breach of the same or any other of the terms, provisions, and covenants herein contained. Forbearance by Landlord in enforcement of one or more of the remedies herein provided upon an event of default shall not be deemed or construed to constitute a waiver of such default. The acceptance of any Rent hereunder by Landlord following the occurrence of any default, whether or not known to Landlord, shall not be deemed a waiver of any such default, except only a default in the payment of the Rent so accepted.

19.5 Landlord Default. Landlord shall not be in default in the performance of any obligation required to be performed by Landlord under this Lease unless Landlord has failed to perform such obligation within thirty (30) days after the receipt of written notice to Landlord (and any mortgagee of whom Tenant has been notified) from Tenant specifying in detail Landlord's failure to perform; provided however, that if the nature of Landlord's obligation is such that more than thirty (30) days are required for its performance, then Landlord shall not be deemed in default if it commences such performance within such thirty (30) day period and thereafter diligently pursues the same to completion. Tenant agrees that, prior to commencing a legal action against Landlord for failure to cure such default as provided in the preceding sentence, any mortgagee which received notice of such default shall have an additional thirty (30) day period). Upon any such uncured default by Landlord and any mortgagee which received notice of such default, Tenant may exercise any of its rights provided in law or at equity; provided, however: (a) Tenant shall have no right to offset or abate rent in the event of any default by Landlord under this Lease; (c) Tenant's rights and remedies hereunder shall be limited to the extent (i) Tenant has expressly waived in this Lease any of such rights or remedies and/or (ii) this Lease otherwise expressly limits Tenant's rights or remedies; and (d) Landlord will not be liable for any consequential damages.

20. COVENANT OF QUIET ENJOYMENT

Landlord covenants that Tenant, on paying the Rent, charges for services and other payments herein reserved and on keeping, observing and performing all the other terms, covenants, conditions, provisions and agreements herein contained on the part of Tenant to be kept, observed and performed, shall, during the Lease Term, peaceably and quietly have, hold and enjoy the Premises subject to the terms, covenants, conditions, provisions and agreements hereof without interference by any persons lawfully claiming by or through Landlord. The foregoing covenant is in lieu of any other covenant express or implied.

21. SECURITY DEPOSIT

Concurrent with Tenant's execution of this Lease, Tenant shall deposit with Landlord a security deposit (the "Security Deposit") in the amount set forth in Section 10 of the Summary. The Security Deposit shall be held by Landlord as security for the faithful performance by Tenant of all the terms, covenants, and conditions of this Lease to be kept and performed by Tenant during the Lease Term. If Tenant defaults with respect to any provisions of this Lease, Landlord may use, apply or retain all or any part of the Security Deposit for the payment of any Rent or any other sum in default, to cure Tenant's default hereunder, or to compensate Landlord for any other direct loss or damage that Landlord may suffer by reason of Tenant's default. If any portion of the Security Deposit is so used or applied, Tenant shall, within five (5) business days after written demand therefor, deposit cash with Landlord in an amount sufficient to restore the Security Deposit to its original amount, and Tenant's failure to do so shall be a default under this Lease. Landlord shall return the Security Deposit (less any portion thereof used, applied or retained by Landlord as permitted herein) to Tenant within sixty (60) days following the expiration of the Lease Term;

provided that Landlord may retain from said Security Deposit an amount reasonably estimated by Landlord to cover potential Operating Expense, Insurance Expense, Utility Expense and Tax Expense reconciliation payments due for the Expense Year in which the Lease Term expires and any amounts so retained and not applied will be returned to Tenant within thirty (30) days after the issuance of a Statement for the applicable Expense Year. Tenant shall not be entitled to any interest on the Security Deposit and Landlord will not be required to segregate the Security Deposit or keep the Security Deposit separate from the general funds of Landlord. Tenant hereby waives the provisions of Section 1950.7 of the California Civil Code and all other provisions of law, now or hereafter in force, which provide that Landlord may claim from a security deposit only those sums reasonably necessary to remedy defaults in the payment of rent, to repair damage caused by Tenant or to clean the Premises.

22. INTENTIONALLY OMITTED

23. SIGNS

23.1 <u>Premises Identification and Building Directory Signage</u>. Tenant shall be entitled, at Tenant's sole cost and expense, to (i) Building-standard identification signage outside of Tenant's Premises on the floor on which Tenant's Premises are located, and (ii) to one (1) line on the Building directory to display Tenant's name and location in the Building. The location, quality, design, style, and size of the above referenced signage shall be consistent with the Landlord's Building standard signage program and will be subject to Landlord's approval, in its sole discretion. Notwithstanding the foregoing, so long as Tenant leases an entire floor of the Building, Tenant may elect to install a sign identifying Tenant in the elevator lobby of such full floor, subject to Landlord's reasonable prior approval of such signage. All signage will be subject to receipt of any and all required governmental permits and approvals, and shall be subject to any covenants, conditions and restrictions affecting the Real Property. Any change in Tenant's signage shall be at Tenant's sole cost and expense.

23.2 <u>Prohibited Signage and Other Items</u>. Any other signs, notices, logos, pictures, names or advertisements which are installed in the Common Areas or on the exterior of the Building or are visible from outside the Premises and that have not been individually approved by Landlord may be removed without notice by Landlord at the sole expense of Tenant.

23.3 Exterior Signage. Subject to this Section 23.3, Tenant shall be entitled to install, at its sole cost and expense, one strip on the Building monument sign on Treena Street and one (1) sign on the exterior of the Building visible from Interstate 15 ("Signage"). Tenant's right to the Building top Signage will be exclusive for so long as Tenant has the right to such Building top Signage pursuant to this Lease; however, Landlord retains the absolute right to permit third parties to install eyebrow signage on the Building. The graphics, materials, size, color, design, lettering, lighting (if any), specifications and exact location of the Signage (collectively, the "Signage Specifications") shall be subject to the prior written approval of Landlord, which approval shall not be unreasonably withheld. In addition, the Signage and all Signage Specifications therefore shall be subject to Tenant's receipt of all required governmental permits and approvals, shall be subject to all applicable governmental laws and ordinances, and all covenants, conditions and restrictions affecting the Real Property. Tenant hereby acknowledges that, notwithstanding

Landlord's approval of the Signage and/or the Signage Specifications therefor, Landlord has made no representations or warranty to Tenant with respect to the probability of obtaining such approvals and permits. In the event Tenant does not receive the necessary permits and approvals for the Signage, Tenant's and Landlord's rights and obligations under the remaining provisions of this Lease shall not be affected. The cost of installation of the Signage, as well as all costs of design and construction of such Signage and all other costs associated with such Signage, including, without limitation, permits, maintenance and repair, shall be the sole responsibility of Tenant. Notwithstanding anything to the contrary contained herein, in the event that at any time during the Term of this Lease (or any Option Term, if applicable), Tenant fails to occupy at least 20,000 rentable square feet in the Building, Tenant's right to the Signage shall thereupon terminate and Tenant shall remove such Signage as provided in this Section 23.3 below. The rights to the Signage shall be personal to the Original Tenant and may not be transferred. Should the Signage require maintenance or repairs as determined in Landlord's reasonable judgment, Landlord shall have the right to provide written notice thereof to Tenant and Tenant shall cause such repairs and/or maintenance to be performed within thirty (30) days after receipt of such notice from Landlord at Tenant's sole cost and expense. Should Tenant fail to perform such maintenance and repairs within the period described in the immediately preceding sentence, Landlord shall have the right to cause such work to be performed and to charge Tenant, as Additional Rent, for the cost of such work. Upon the expiration or earlier termination of this Lease (or the termination of Tenant's Signage right as described above). Tenant shall, at Tenant's sole cost and expense, cause the Signage to be removed from the exterior of the Building and the Building monument sign and shall cause the exterior of the Building and the Building monument sign to be restored to their condition existing prior to the placement of such Signage. If Tenant fails to remove such Signage and to restore the exterior of the Building and the Building monument sign as provided in the immediately preceding sentence within thirty (30) days following the expiration or earlier termination of this Lease (or the termination of Tenant's Signage as provided above), then Landlord may perform such work, and all costs and expenses incurred by Landlord in so performing such work shall be reimbursed by Tenant to Landlord within ten (10) days after Tenant's receipt of invoice therefor. The immediately preceding sentence shall survive the expiration or earlier termination of this Lease. Any signs, notices, logos, pictures, names or advertisements which are installed and that have not been individually approved by Landlord may be removed without notice by Landlord at the sole expense of Tenant. Except as provided in this Section 23.3 above, Tenant may not install any signs on the exterior or roof of the Building or the common areas of the Building or the Real Property. Any signs, window coverings, or blinds (even if the same are located behind the Landlord approved window coverings for the Building), or other items visible from the exterior of the Premises or Building are subject to the prior approval of Landlord, in its sole discretion. In the event Tenant has not installed either of the signs constituting Tenant's Signage on or before the date which is six (6) months after the Lease Commencement Date, Tenant's right to such Signage will terminate and Tenant will not be permitted to install the Tenant's Signage after such date

Should the name of the Original Tenant change, Tenant shall be entitled to modify, at Tenant's sole cost and expense, Tenant's Signage to reflect Tenant's new name, but only if Tenant's new name is not an "Objectionable Name." The term "**Objectionable Name**" shall mean any name that (i) relates to an entity that is of a character or reputation, or is associated with a political orientation or faction that is materially inconsistent with the quality of the Real Property, or which would

otherwise reasonably offend a landlord of a building comparable to the Real Property, taking into consideration the level and visibility of Tenant's Signage, or (ii) conflicts with any covenants in other leases of space in the Real Property.

24. COMPLIANCE WITH LAW

24.1 <u>Compliance With Laws</u>. Landlord represents that, to the best of its knowledge, as of the date of this Lease, Landlord has not received any notice from the applicable legal authority stating that the Building, Premises and/or Common Areas are in violation of any applicable law, statute, ordinance or other governmental rule, regulation or requirement in force. Landlord shall take all steps which Landlord believes are reasonably required to maintain the Common Areas in compliance with the Americans with Disabilities Act throughout the Term of this Lease, subject to inclusion within Operating Expenses to the extent permitted by Article 4 above. Tenant shall not do anything or suffer anything to be done in or about the Premises which will in any way conflict with any law, statute, ordinance or other governmental rule, regulation or requirement now in force or which may hereafter be enacted or promulgated. At its sole cost and expense, Tenant shall promptly comply with all governmental measures governing the Premises, other than the making of structural changes or changes to the Systems and Equipment or Common Areas, such changes will be made by Landlord at its expense, but subject to reimbursement as an Operating Expense to the extent permitted by Article 4; however, if such changes are required due to Tenant's Alterations or the particular nature of Tenant's use of the Premises, Tenant shall, as Additional Rent, reimburse Landlord for the cost thereof within thirty (30) days following receipt of an invoice therefor.

California Certified Access Specialist Inspection. In accordance with California Civil Code section 1938, Landlord 24.2 states that the Premises have not undergone inspection by a Certified Access Specialist (CASp) (defined in California Civil Code § 55.52(a)(3)). Pursuant to Section 1938 of the California Civil Code, Landlord hereby provides the following notification to Tenant: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction related accessibility standards within the premises." If Tenant requests to perform a CASp inspection of the Premises, Tenant shall, at its cost, retain a CASp approved by Landlord (provided that Landlord may designate the CASp, at Landlord's option) to perform the inspection of the Premises at a time agreed upon by the parties. Tenant shall provide Landlord with a copy of any report or certificate issued by the CASp (the "CASp Report") and Tenant shall, at its cost, promptly complete any modifications necessary to correct violations of construction related accessibility standards identified in the CASp Report, which modifications will be completed as part of Tenant's Work or as an Alteration, as applicable, notwithstanding anything to the contrary in this Lease. Tenant agrees to keep the information in the CASp Report confidential except as necessary for the Tenant to complete such modifications.

25. LATE CHARGES

If any installment of Rent or any other sum due from Tenant shall not be received by Landlord or Landlord's designee within five (5) business days after said amount is due, then (i) Tenant shall pay to Landlord a late charge equal to five percent (5%) of the amount due (but in no event shall such charge be in excess of the maximum amount permitted by applicable law) plus any attorneys' fees incurred by Landlord by reason of Tenant's failure to pay Rent and/or other charges when due hereunder, and (ii) such unpaid amounts shall thereafter bear interest until paid at a rate equal to the prime rate established from time to time by Wells Fargo Bank (or if Wells Fargo ceases to exist or to publish such a rate, then the rate published by the largest federally chartered banking institution in California) plus five percent (5%) per annum, provided that in no case shall such rate be higher than the highest rate permitted by applicable law. The late charge and interest shall be deemed Additional Rent and the right to require it shall be in addition to all of Landlord's other rights and remedies hereunder or at law and shall not be construed as liquidated damages or as limiting Landlord's remedies in any manner.

26. LANDLORD'S RIGHT TO CURE DEFAULT

All covenants and agreements to be kept or performed by Tenant under this Lease shall be performed by Tenant at Tenant's sole cost and expense and without any reduction of Rent. If Tenant shall fail to perform any of its obligations under this Lease, within a reasonable time after such performance is required by the terms of this Lease, Landlord may, but shall not be obligated to, after reasonable prior notice to Tenant, make any such payment or perform any such act on Tenant's part without waiving its right based upon any default of Tenant and without releasing Tenant from any obligations hereunder, in which event Tenant shall reimburse Landlord, upon demand, for the sums incurred by Landlord in connection therewith. Tenant's reimbursement obligations under this Article 26 shall survive the expiration or sooner termination of the Lease Term.

27. ENTRY BY LANDLORD

Landlord reserves the right at all reasonable times and upon reasonable notice to the Tenant to enter the Premises to (i) inspect them; (ii) show the Premises to prospective purchasers, mortgagees or tenants, or to the ground or underlying lessors; (iii) post notices of nonresponsibility; or (iv) alter, improve or repair the Premises or the Building if necessary to comply with current Building codes or other applicable laws, or for structural alterations, repairs or improvements to the Building. Notwithstanding anything to the contrary contained in this Article 27, Landlord may enter the Premises at any time to (A) perform services required of Landlord; (B) take possession due to any breach of this Lease in the manner provided herein; (C) perform any covenants of Tenant which Tenant fails to perform; or (D) to address an emergency. Except as specifically provided in this Lease, any such entries shall be without the abatement of Rent, shall not be deemed an unlawful entry, or an actual or constructive eviction, and shall include the right to take such reasonable steps as required to accomplish the stated purposes. Tenant hereby waives any claims for damages or for any injuries or inconvenience to or interference with Tenant's business, lost profits, any loss of occupancy or quiet enjoyment of the Premises, and any other loss occasioned thereby. Landlord will comply with any reasonable security requirements of Tenant during any entry pursuant to this

Article 27, except in the event of an emergency; provided such requirements do not result in an increased cost to Landlord.

28. TENANT PARKING

Subject to compliance with the rules and regulations reasonably prescribed by Landlord, Tenant shall rent parking passes on a monthly basis throughout the Lease Term in the amount set forth in Section 11 of the Summary to park in the Parking Facilities, on a non-exclusive basis, at the prevailing rate charged for parking passes at the location of such passes; provided such passes shall be free throughout the initial Lease Term. Of Tenant's parking passes, Tenant will have the right to designate two (2) unreserved parking passes as reserved exterior spaces ("Exterior Reserved Spaces") at the initial locations shown on Exhibit E attached hereto. Tenant will be permitted to construct a cover over the Exterior Reserved Spaces, subject to Landlord's review and approval of all plans for such cover and the construction of such cover will be performed at Tenant's sole cost and in accordance with Article 8 above. Any cover constructed by Tenant will be removed and the parking spaces restored to their original condition if Landlord so requests upon the expiration or earlier termination of this Lease. Landlord may relocate the Exterior Reserved Spaces at any time on reasonable prior notice to Tenant; provided that Landlord will not relocate such Exterior Reserved Spaces more than three (3) times during the initial Lease Term; provided further that Tenant will pay the cost to remove the existing parking covers and restore the parking area and rebuild parking covers in the relocated location for the first (1st) relocation event only, and in the event Landlord subsequently relocates the Exterior Reserved Spaces, Landlord will pay the cost to remove the existing parking covers and rebuild parking covers in the relocated location. Landlord may, without incurring any liability to Tenant and without any abatement of Rent under this Lease, from time to time, close-off or restrict access to the Parking Facilities, or relocate Tenant's parking passes to other parking structures and/or surface parking areas within a reasonable distance of the Premises, for purposes of permitting or facilitating any such construction, alteration, improvements or repairs with respect to the Parking Facilities or to accommodate or facilitate renovation, alteration, construction or other modification of other improvements or structures located on the Real Property. If the applicable governmental authorities impose a parking tax or other charge on Tenant's use of such parking, such taxes and/or charges shall be paid directly by Tenant or the parking users, or, if directly imposed against Landlord, Tenant shall reimburse Landlord for all such taxes and/or charges.

29. HAZARDOUS MATERIALS

Landlord hereby represents that to Landlord's current actual knowledge, without investigation, there are no Hazardous Materials present in the Premises in violation of applicable laws as of the date hereof. Tenant shall not cause or permit any Hazardous Materials (as defined below) to be brought upon, kept or used in or about the Premises or the Building by Tenant or any Tenant Parties, without the prior written consent of Landlord, which may be withheld in Landlord's sole discretion. If Tenant breaches the obligations stated in the preceding sentence, or if the presence of Hazardous Material on the Premises caused or permitted by Tenant results in contamination of the Premises, or if contamination of the Premises by Hazardous Material otherwise occurs for which Tenant is legally liable to Landlord for damage resulting therefrom, then Tenant shall indemnify, defend and hold Landlord harmless from any and all claims, judgments, damages, penalties, fines,

costs, liabilities or losses (including, without limitation, diminution in value of the Premises or the Real Property, damages for the loss or restriction on use of rentable or usable space or of any amenity of the Real Property, damages arising from any adverse impact on marketing of space on the Real Property, and sums paid in settlement of claims, and reasonable attorney's fees, consultant fees and expert fees) which arise during or after the term of the Lease as a result of such contamination. This indemnification of Landlord by Tenant includes, without limitation, costs incurred in connection with any investigation of site conditions or any cleanup, remedial, removal or restoration work required by any federal, state or local governmental agency or political subdivision because of Hazardous Material present in the soil or ground water on or under the Real Property. Without limiting the foregoing, if the presence of any Hazardous Material at the Real Property is caused by or is permitted by Tenant to remain and thereafter results in any contamination of the Real Property. Tenant shall promptly take all actions at its sole expense as are necessary to return the Real Property to the condition existing prior to the introduction of any such Hazardous Material; provided, that Landlord's approval of such actions shall first be obtained. Landlord's approval shall not be unreasonably withheld so long as such actions would not potentially have any material adverse long-term or short-term effect on the Real Property. Any removal or remedial actions taken by Tenant hereunder shall be done so as to cause the least possible interference with the business of other tenants at the Real Property. Tenant shall give Landlord written notice of any evidence of mold, water leaks or water infiltration into the Premises promptly upon discovery of the same. At its expense. Tenant shall investigate, clean up and remediate any mold in the Premises that results from Tenant's activities in the Premises or failure to notify Landlord of a water leak or other condition known to Tenant leading to the presence of mold. Investigation, clean up and remediation may only be performed after Landlord has approved Tenant's plan for such remediation. All clean up and remediation will be done in compliance with Applicable Laws. The provisions of this Article 29 shall survive the expiration or earlier termination of this Lease.

"Hazardous Materials" means (i) any chemical, compound, material, mixture or substance that is now or hereafter defined or listed in, regulated under, or otherwise classified pursuant to, any Environmental Laws as "hazardous substance," "hazardous material," "hazardous waste," "extremely hazardous waste or substance," "infectious waste," "toxic substance," "toxic pollutant," or any other formulation intended to define, list or classify substances by reason of deleterious properties, including ignitability, corrosiveness, reactivity, carcinogenicity, toxicity or reproductive toxicity; (ii) petroleum, any fraction of petroleum, natural gas, natural gas liquids, liquefied natural gas, synthetic gas usable for fuel (or mixtures of natural gas and such synthetic gas), and ash produced by a resource recovery facility utilizing a municipal solid waste stream, and drilling fluids, produced waters and other wastes associated with the exploration, development or production of crude oil, natural gas or geothermal resources; (iii) asbestos and any asbestos containing material; and (iv) any other material that, because of its quantity, concentration or physical or chemical characteristics, is the subject of regulation under Applicable Law or poses a significant present or potential hazard to human health or safety or to the environment if released into the workplace or the environment, or to Hazardous Materials, including the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended by the Superfund Amendments and Reauthorization Act of 1986, and the Resource Conservation and Recovery Act of 1976, as

amended by the Used Oil Recycling Act of 1980, the Solid Waste Disposal Act Amendments of 1980, and the Hazardous and Solid Waste Amendments of 1984.

30. MISCELLANEOUS PROVISIONS

30.1 <u>Binding Effect</u>. Each of the provisions of this Lease shall extend to and shall, as the case may require, bind or inure to the benefit not only of Landlord and of Tenant, but also of their respective successors or assigns, provided this clause shall not permit any assignment by Tenant contrary to the provisions of Article 14 of this Lease.

30.2 <u>No Air Rights</u>. No rights to any view or to light or air over any property, whether belonging to Landlord or any other person, are granted to Tenant by this Lease.

30.3 <u>Modification of Lease</u>. Should any current or prospective mortgagee or ground lessor for the Building require a modification or modifications of this Lease, which modification or modifications will not cause an increased cost or expense to Tenant or in any other way materially and adversely change the rights and obligations of Tenant hereunder, then and in such event, Tenant agrees that this Lease may be so modified and agrees to execute whatever documents are required therefor and deliver the same to Landlord within ten (10) business days following the request therefor.

30.4 <u>Transfer of Landlord's Interest</u>. In the event Landlord transfers all or any portion of its interest in the Real Property and Building and in this Lease (including, without limitation, the transfer of the Security Deposit to the transferee), Landlord shall automatically be released from all remaining liability under this Lease and Tenant agrees to look solely to such transferee for the performance of Landlord's obligations hereunder after the date of transfer.

30.5 <u>Captions</u>. The captions of Articles and Sections are for convenience only and shall not be deemed to limit, construe, affect or alter the meaning of such Articles and Sections.

30.6 <u>Time of Essence</u>. Time is of the essence of this Lease and each of its provisions.

30.7 <u>Partial Invalidity</u>. If any term, provision or condition contained in this Lease shall, to any extent, be invalid or unenforceable, the remainder of this Lease, or the application of such term, provision or condition to persons or circumstances other than those with respect to which it is invalid or unenforceable, shall not be affected thereby, and each and every other term, provision and condition of this Lease shall be valid and enforceable to the fullest extent possible permitted by law.

30.8 <u>Landlord Exculpation</u>. In consideration of the benefits accruing hereunder to Tenant and notwithstanding anything in this Lease to the contrary, it is expressly understood and agreed that notwithstanding any applicable law to the contrary, the liability of Landlord hereunder (including any successor landlord) and any recourse by Tenant against Landlord shall be limited solely and exclusively to the interest of Landlord in the Building, and neither Landlord, nor any of its constituent partners, members, shareholders, officers, directors or employees shall have any personal liability therefor, and Tenant hereby expressly waives and releases such personal liability on behalf of itself and all persons claiming by, through or under Tenant.

30.9 Entire Agreement. It is understood and acknowledged that there are no oral agreements between the parties hereto affecting this Lease and this Lease supersedes and cancels any and all previous negotiations, arrangements, brochures, agreements and understandings, if any, between the parties hereto or displayed by Landlord to Tenant with respect to the subject matter thereof, and none thereof shall be used to interpret or construe this Lease. This Lease and any side letter or separate agreement executed by Landlord and Tenant in connection with this Lease and dated of even date herewith contain all of the terms, covenants, conditions, warranties and agreements of the parties relating in any manner to the rental, use and occupancy of the Premises. None of the terms, covenants, conditions or provisions of this Lease can be modified, deleted or added to except in writing signed by the parties hereto. Any deletion of language from this Lease prior to its execution by Landlord and Tenant shall not be construed to raise any presumption, canon of construction or implication, including, without limitation, any implication that the parties intended thereby to state the converse of the deleted language. The parties hereto acknowledge and agree that each has participated in the negotiation and drafting of this Lease shall not be resolved by any rule of interpretation providing for interpretation against the party who caused the uncertainty to exist or against the draftsman.

30.10 <u>Force Majeure</u>. Any prevention, delay or stoppage due to strikes, lockouts, labor disputes, acts of God, inability to obtain services, labor, or materials or reasonable substitutes therefor, governmental actions, civil commotions, fire or other casualty, and other causes beyond the reasonable control of the party obligated to perform, except with respect to the obligations imposed with regard to Rent and other charges to be paid by Tenant pursuant to this Lease (collectively, the "Force Majeure"), notwithstanding anything to the contrary contained in this Lease, shall excuse the performance of such party for a period equal to any such prevention, delay or stoppage and, therefore, if this Lease specifies a time period for performance of an obligation of either party, that time period shall be extended by the period of any delay in such party's performance caused by a Force Majeure.

30.11 <u>Notices</u>. Any notice, demand or other communication given under the provisions of this Lease (collectively, "**Notices**") by either party to the other party shall be effective only if in writing and (a) personally served, (b) mailed by United States registered or certified mail, return receipt requested, postage prepaid, or (c) sent by a nationally recognized courier service (*e.g.*, Federal Express) for next-day delivery. Notices shall be directed to the parties at their respective addresses set forth in the Summary. In the event that a different address is furnished by either party to the other party in accordance with the procedures set forth in this Section 30.11, Notices shall thereafter be sent or delivered to the new address. Notices given in the foregoing manner shall be deemed given (a) when actually received or refused by the party to whom sent if delivered by carrier or personally served, (b) if mailed, on the day of actual delivery or refusal as shown by the addressee's registered or certified mail receipt or (c) on the date shown on the courier service receipt as the delivery date (or date of refusal of delivery). For purposes of this Section 30.11, a "business day" is Monday through Friday, excluding holidays observed by the United States Postal Service.

30.12 <u>Joint and Several Liability</u>. If more than one person or entity executes this Lease as Tenant: (a) each of them is and shall be jointly and severally liable for the covenants, conditions,

provisions and agreements of this Lease to be kept, observed and performed by Tenant; and (b) the act or signature of, or notice from or to, any one or more of them with respect to this Lease shall be binding upon each and all of the persons and entities executing this Lease as Tenant with the same force and effect as if each and all of them had so acted or signed, or given or received such notice.

30.13 <u>Attorneys' Fees</u>. If either party commences litigation against the other for the specific performance of this Lease, for damages for the breach hereof or otherwise for enforcement of any remedy hereunder, the parties hereto agree to and hereby do waive any right to a trial by jury and, in the event of any such commencement of litigation, the prevailing party shall be entitled to recover from the other party such costs and reasonable attorneys' fees as may have been incurred, including any and all costs incurred in enforcing, perfecting and executing such judgment.

30.14 <u>Governing Law</u>. This Lease shall be construed and enforced in accordance with the laws of the State of California.

30.15 <u>Brokers</u>. Landlord and Tenant hereby warrant to each other that they have had no dealings with any real estate broker or agent in connection with the negotiation of this Lease, excepting only the real estate brokers or agents specified in Section 12 of the Summary (the "**Brokers**"), and that they know of no other real estate broker or agent who is entitled to a commission in connection with this Lease. Each party agrees to indemnify and defend the other party against and hold the other party harmless for, from and against any and all claims, demands, losses, liabilities, lawsuits, judgments, and costs and expenses (including without limitation reasonable attorneys' fees) with respect to any leasing commission or equivalent compensation alleged to be owing on account of the indemnifying party's dealings with any real estate broker or agent other than the Brokers.

30.16 <u>Paragraph Headings</u>. The headings and titles to the Articles and Sections of this Lease are for convenience only and shall have no effect on the interpretation of any part of this Lease.

30.17 <u>Title and Recording</u>. Landlord's title is and always shall be paramount to the title of Tenant. Nothing herein contained shall empower Tenant to do any act which can, shall or may encumber the title of Landlord. Tenant shall not record this Lease or any memorandum of lease.

30.18 <u>Authority</u>. Tenant covenants, warrants, and represents that each individual executing, attesting, and/or delivering this Lease on behalf of Tenant is authorized to do so on behalf of Tenant; this Lease is binding upon and enforceable against Tenant; and Tenant is duly organized and legally existing in the state of its organization and is qualified to do business in the state in which the Premises are located.

30.19 <u>Relationship</u>. This Lease shall create only the relationship of landlord and tenant between the parties, and not a partnership, joint venture, or any other relationship.

30.20 <u>Survival of Obligations</u>. The expiration of the Lease Term, whether by lapse of time or otherwise, shall not relieve Tenant of any obligations which accrued prior to or which may

continue to accrue after the expiration or earlier termination of this Lease. Those terms or provisions of this Lease which this Lease expressly states shall survive, or which by their context are clearly intended to survive the expiration or termination of this Lease, shall survive the expiration or earlier termination of this Lease.

30.21 <u>Transportation Management</u>. Tenant shall fully comply with all present or future programs intended to manage parking, transportation or traffic in and around the Building and in connection therewith, Tenant shall take responsible action for the transportation planning and management of all employees located at the Premises by working directly with Landlord, any governmental transportation management organization or any other transportation-related committees or entities.

30.22 <u>Waiver of Jury Trial</u>. To the maximum extent permitted by law, Landlord and Tenant hereby waive their respective rights to trial by jury of any cause of action, claim, counter-claim or cross-complaint in any action, proceeding or hearing brought by either Landlord against Tenant, or Tenant against Landlord on any matter whatsoever arising out of, or in any connected with, this Lease, the relationship of Landlord and Tenant, Tenant's use or occupancy of the Premises, or any claim of injury or damage, or the enforcement of any remedy under any law, statute or regulation, emergency or otherwise now or hereinafter in effect.

30.23 <u>Confidentiality</u>. Tenant acknowledges that the content of this Lease and any related documents are confidential information. Except as Tenant may be required to disclose under law or regulation. Tenant shall keep such confidential information strictly confidential and shall not disclose such confidential information to any person or entity other than Tenant's financial, legal, and space planning consultants.

30.24 Landlord Renovations. Tenant acknowledges that Landlord may, but shall not be obligated to (other than as specifically set forth herein), during the Lease Term renovate, improve, alter, or modify (collectively, the "**Renovations**") the Building, Premises, and/or Real Property, including without limitation the Parking Facilities, Common Areas, systems and equipment, roof, and structural portions of the same. Tenant hereby agrees that such Renovations and Landlord's actions in connection with such Renovations shall in no way constitute a constructive eviction of Tenant nor entitle Tenant to any abatement of Rent. Landlord shall have no responsibility or for any reason be liable to Tenant for any direct or indirect injury to or interference with Tenant's business arising from the Renovations, nor shall Tenant be entitled to any compensation or damages from Landlord for loss of the use of the whole or any part of the Premises or of Tenant's personal property or improvements resulting from the Renovations or Landlord's actions in connection with such Renovations, or for any inconvenience or annoyance occasioned by such Renovations or Landlord's actions in connection with such Renovations (other than any rent abatement which may be permitted pursuant to Section 6.10 above).

30.25 <u>Financial Statements</u>. In the event Tenant is not a publicly traded company with publicly available financial statements, upon ten (10) business days prior written request from Landlord (which Landlord may make at any time during the Term but no more often that two (2) times in any calendar year), Tenant shall deliver to Landlord a current financial statement of Tenant and any guarantor of this Lease certified by an officer of Tenant or guarantor, as applicable.

30.26 <u>Excepted Rights</u>. Landlord reserves the absolute right to effect such other tenancies in the Building and/or Real Property as Landlord, in the exercise of its sole business judgment, shall determine to best promote the interests of the Building and/or Real Property. Landlord shall also have the right (but not the obligation) to temporarily close the Building if Landlord reasonably determines that there is an imminent danger of significant damage to the Building or of personal injury to Landlord's employees or the occupants of the Building. The circumstances under which Landlord may temporarily close the Building shall include, without limitations, electrical interruptions, hurricanes, terrorist activities and civil disturbances. A closure of the Building under such circumstances shall not constitute a constructive eviction nor entitle Tenant to an abatement or reduction of rent payable hereunder (other than for an electrical interruption caused by Landlord which will be governed by Section 6.10 above as applicable).

30.27 <u>Counterparts</u>. This Lease may be executed in any number of counterparts, which may be delivered electronically, via facsimile or by other means. Each party may rely upon signatures delivered electronically or via facsimile as if such signatures were originals. Each counterpart of this Lease shall be deemed to be an original, and all such counterparts (including those delivered electronically or via facsimile), when taken together, shall be deemed to constitute one and the same instrument.

30.28 <u>USA Patriot Act Compliance</u>. Tenant represents and warrants that neither Tenant nor any person or entity who directly owns a ten percent (10%) or greater equity interest in Tenant, nor any of its officers, directors or managing members, is, or will be at any time during the Lease Term, a person or entity with whom United States persons or entities are restricted from doing business under regulations of the Office of Foreign Asset Control of the Department of the Treasury (including those named on the Office of Foreign Asset Control's "Specially Designated Nationals and Blocked Persons" list, found at http://www.treas.gov/offices/enforcement/ofac) or under any statute, Executive Order (including Executive Order 13224 signed on September 24, 2001 and entitled "Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism"), or other governmental action; and (b) Tenant's activities do not violate the International Money Laundering Abatement and Financial Anti-Terrorism Act of 2001 or the regulations or orders promulgated thereunder (as amended from time to time).

30.29 <u>Fitness Center Usage</u>. Landlord currently provides a fitness center in the Other Building, which is available for use by tenants of the Real Property. Landlord makes no representation or warranty regarding the fitness center, the type of equipment which will be provided or any other amenities associated with such fitness center, and will have no obligation to continue to provide the fitness center at the Real Property once such fitness center is completed. So long as Landlord provides a fitness center at the Real Property for the use of tenants, Tenant will have the right to allow its employees working in the Premises with access to the fitness center at no additional rental charge. Any such use will be subject to compliance with Landlord's rules and regulations relating to the fitness center, any security or access control system and each employee's signature on Landlord's standard waiver/release form. Neither Tenant nor its employees shall do or permit anything to be done in or about the fitness center which would in any way unreasonably obstruct or interfere with the rights of other tenants or occupants of the Real Property; nor shall Tenant or its employees use or allow the fitness center to be used for any improper, immoral, unlawful or

objectionable purpose; nor shall Tenant or its employees cause, maintain or permit any nuisance or offensive sound, smell or light in, on or about the fitness center. Tenant will be responsible for any damage done to the fitness center and the Building arising out of or in any way pertaining to Tenant's or its employees' use of the fitness center. If Tenant or its employees fail to abide by the Landlord's rules for use or violate the terms of this <u>Section 30.29</u>, Landlord may revoke Tenant's right to use the fitness center, and such revocation shall have no effect on this Lease. Landlord and its agents shall not be responsible for any loss or theft whatsoever of any property or anything placed or stored by Tenant or its employees in the fitness center. Tenant, as a material part of the consideration of this Lease, waives all claims or demands against Landlord for any such loss, damage or injury of Tenant or Tenant's property, and agrees to indemnify and hold all Landlord Indemnified Parties harmless therefrom. Tenant's indemnity obligations as set forth in <u>Section 10.1</u> of this Lease will apply to Tenant's use of the fitness center, and solely for purposes of this sentence, all references to the "Premises" in <u>Section 10.1</u> shall include the fitness center.

[Signature Page Follows]

IN WITNESS WHEREOF, Landlord and Tenant have caused this Lease to be executed the day and date first above written.

"Landlord": TPP 212 SCRIPPS, LLC, a Delaware limited liability company

By: /s/ John M. Mearns

Name: John M. Mearns

Title: Vice President

"Tenant": KRATOS DEFENSE & SECURITY SOLUTIONS, INC., a Delaware corporation *By: /s/ Eric DeMarco Name: Eric DeMarco Title: CEO & President

*By: /s/ Marie Mendoza Name: Marie Mendoza Title: Secretary

*NOTE:

If Tenant is a California corporation, then one of the following alternative requirements must be satisfied:

(A) This Lease must be signed by two (2) officers of such corporation: one being the chairman of the board, the president or a vice president, and the other being the secretary, an assistant secretary, the chief financial officer or an assistant treasurer. If one (1) individual is signing in two (2) of the foregoing capacities, that individual must identify the two (2) capacities.
(B) If the requirements of (A) above are not satisfied, then Tenant shall deliver to Landlord evidence in a form reasonably acceptable to Landlord that the signatory(ies) is (are) authorized to execute this Lease.

If Tenant is a corporation incorporated in a state other than California, then Tenant shall deliver to Landlord evidence in a form reasonably acceptable to Landlord that the signatory(ies) is

(are) authorized to execute this Agreement.

[Exhibits mentioned in this document have been excluded from this filing]

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Eric M. DeMarco, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Kratos Defense & Security Solutions, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2017

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

/s/ ERIC M. DEMARCO

Eric M. DeMarco Chief Executive Officer, President (Principal Executive Officer)

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Deanna H. Lund, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Kratos Defense & Security Solutions, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2017

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

/s/ DEANNA H. LUND

Deanna H. Lund Executive Vice President, Chief Financial Officer (Principal Financial Officer and Acting Principal Accounting Officer)

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

In connection with the accompanying Quarterly Report of Kratos Defense & Security Solutions, Inc. (the "Company") on Form 10-Q for the quarter ended March 26, 2017 (the "Report"), I, Eric M. DeMarco, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 4, 2017

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

/s/ ERIC M. DEMARCO

Eric M. DeMarco Chief Executive Officer, President (Principal Executive Officer)

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

In connection with the accompanying Quarterly Report of Kratos Defense & Security Solutions, Inc. (the "Company") on Form 10-Q for the quarter ended March 26, 2017 (the "Report"), I, Deanna H. Lund, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 4, 2017

KRATOS DEFENSE & SECURITY SOLUTIONS, INC.

/s/ DEANNA H. LUND

Deanna H. Lund Executive Vice President, Chief Financial Officer (Principal Financial Officer and Acting Principal Accounting Officer)